Curtin Medical School

Epigenetics, cholesterol lowering and diabetes: a comparative study of statins and HDAC inhibitors

Stephanie Claire Bridgeman

0000-0002-8399-5458

This thesis is presented for the Degree of Doctor of Philosophy of Curtin University

January 2022

Author's Declaration

To the best of my knowledge and belief this thesis contains no material previously published by any other person except where due acknowledgment has been made.

This thesis contains no material which has been accepted for the award of any other degree or diploma in any university.

The research presented and reported in this thesis was conducted in compliance with the National Health and Medical Research Council Australian code for the care and use of animals for scientific purposes 8th edition (2013). The proposed research study received animal ethics approval from the Curtin University Animal Ethics Committee, Approval Number # AEC_2016_17.



Date: 31/01/2022

Statement of contributors

This project was designed by myself and my supervisor, Associate Professor Cyril Mamotte. All experiments were conducted by myself with the following exceptions. The insulin assays were conducted by Dr Gaewyn Ellison with my assistance. Dr Gaewyn Ellison also extracted RNA for the 16 h treatment qPCR experiments. Wendy Northrop performed the HDAC activity assays on the mouse liver samples. The mouse liver samples were obtained from a previous animal study by our group, designed by Dr Gaewyn Ellison and overseen by Associate Professor Cyril Mamotte. Dr Thiruvarutchelvan Sabapathy assisted in the preparation of the mouse liver samples and prepared the insulin treatments for the insulin signalling experiments. All data analysis and visualisation were performed by myself. The draft thesis was written by myself and refined with suggestions from Associate Professor Cyril Mamotte, Dr Phillip Melton, Professor Philip Newsholme and Dr David Allen. Similarly, Dr Gaewyn Ellison, Associate Professor Cyril Mamotte, Dr Phillip Melton, Professor Philip Newsholme and Wendy Northrop contributed to the editing of the portions of peer-reviewed publications which appear in this thesis.

Acknowledgements

First and foremost, this thesis would not exist without the guidance and support from my supervisor, Associate Professor Cyril Mamotte, who encouraged me to pursue a PhD despite having little laboratory experience, and whose enthusiasm to explore new ideas and attention to detail, though sometimes frustrating when editing papers, has taught me so much about what it means to be a scientist.

Thanks to my co-supervisors Dr Phillip Melton and Professor Philip Newsholme, who, despite similar names came from different areas of expertise and as such were able to provide different perspectives on my ideas and writings and thus helped produce a more rounded thesis.

I owe an enormous debt to the elder members of our little laboratory team, Dr Gae Ellison and Dr Thiru Sabapathy, who did the groundwork in setting up the laboratory and taught me so much about both the basics of laboratory work and specific techniques. Furthermore, they were there to assist me when, pregnant in the midst of the coronavirus panic, I was reluctant to spend too long on campus or to use certain dangerous reagents.

Thanks to Wendy Northrop, who acted as my own personal assistant for a period and was happy to do some of the boring, thankless laboratory tasks, spending hours washing Western blots.

Thanks to Dr Ross Graham and Clinton Kidman for teaching me RT qPCR and the sharing of reagents that made the whole process much faster. Also, thanks to everyone else in the Curtin Health Innovation Research Institute who may have lent me equipment, shared reagents, offered tips or just friendly conversation over my four years in the building.

Of course, there are my parents, who instilled in me a scientific curiosity from the beginning, Dad, whose own PhD gave me the belief that I could do it, and Mum for so much, tracking down journal articles, looking after my baby so I could write this thesis, the many lunches and constant support. Also, thanks to my mother-in-law Lee for the many babysitting sessions while I worked on this thesis.

And, saving the best for last, the amazing Aaron, who throughout the PhD journey, went from my boyfriend to my husband and the father of my child. Thank you for making me feel confident in myself, for the reassurance when things were not going well, for the encouragement when I didn't feel like working and most of all, for making my life better in so many ways. You are the best!

Dedication

This thesis is dedicated to Ava Fiona Bridgeman, my darling daughter, whom I carried with me through the final experiments and draft writing of this thesis and has been a constant source of joy, inspiration and perspective through the endless editing and revising. Ihope this thesis will be an inspiration for you to reach your own aspirations, whatever they may be, and to never let go of your inquisitive and adventurous spirit.

Abstract

Cardiovascular disease (CVD) and type 2 diabetes (T2D) are two major causes of morbidity and mortality in the modern world, and both have been linked to altered cholesterol levels. In particular, dyslipidaemia, typically characterised by elevated levels of low-density lipoprotein (LDL) cholesterol and triglyceride and/or decreased levels of high-density lipoprotein (HDL) cholesterol, is a major risk factor for both conditions. HMG-CoA reductase (HMGCR) inhibitors, a class of LDL cholesterol (LDL-C) lowering drugs commonly known as statins, are widely used to reduce CVD risk by reducing cellular cholesterol biosynthesis. However, statins have been associated with an increased risk of developing T2D. Epigenetic modifications have also been associated with both CVD and T2D and there have been suggestions that stating have epigenetic effects such as inhibition of histone deacetylases (HDACs), enzymes that remove the epigenetic mark of histone acetylation. Established HDAC inhibitors, particularly the short-chain fatty acid butyrate, have been investigated in animal models of metabolic disease and reportedly not only lower serum cholesterol but also protect against the development and progression of diabetes in these animals. The project described in this thesis therefore set out to compare the effects of statins and established HDAC inhibitors on epigenetic modifications, cholesterol metabolism and T2D.

Studies were performed using a cell culture model, primarily using HepG2 hepatoma cells and BRIN-BD11 insulin secreting cells. THP-1 macrophages, due to the importance of macrophages in cholesterogenic atherosclerosis and CVD, and MDA-MB-231 breast carcinoma cells, as most previous studies reporting epigenetic effects of statins were conducted in the context of cancer, were used in some experiments. The livers of mice treated with statins from a previous animal study were also utilised. In most experiments, cells were treated with statins or HDAC inhibitors for 24 h in lipoprotein-deficient serum (LPDS). The activity of epigenetic modifying enzymes, the cellular cholesterol and triglyceride content and cellular cholesterol uptake were assessed using commercially available kits. Insulin secretion was quantified by sandwich enzyme-linked immunosorbent assay (ELISA). Global levels of modified histones and of proteins involved in cholesterol and glucose metabolism were assessed by immunoblotting. Gene expression of proteins involved in cholesterol and

glucose metabolism was assessed by quantitative real-time PCR (qPCR) following RNA extraction and reverse transcription. Histone modification in proximity to specific genomic regions was determined by chromatin immunoprecipitation-qPCR (ChIP-qPCR).

Both statins and HDAC inhibitors lowered cellular cholesterol in HepG2 and BRIN-BD11 cells. Lowering cholesterol as a result of HMGCR inhibition by statins resulted in activation of the SREBP-2 signalling pathway, leading to increased expression of HMGCR and the LDL receptor (LDLR) and subsequently increased cellular uptake of cholesterol. Conversely, HDAC inhibition reduced SREBP-2 signalling, providing a mechanism for cellular cholesterol lowering, and did not increase cholesterol uptake, calling into question whether HDAC inhibition could lower serum cholesterol in the same manner as statins. Both butyrate and atorvastatin altered levels of proteins involved in reverse cholesterol transport, or HDL cholesterol metabolism, but this was dependant on the cell type and the amount of lipoprotein in the media, making it difficult to draw conclusions about effects on reverse cholesterol transport *in vivo*.

Previous studies by the laboratory group of A/Prof Mamotte indicated that both increasing and decreasing cellular cholesterol, including by statins, impairs insulin secretion. Similarly, high dose 24 h HDAC inhibition impaired both chronic and acutely stimulated insulin secretion and decreased cellular insulin content. However, low dose (1 mM) butyrate for 24 h and acute high dose (5 mM) butyrate enhanced stimulated insulin secretion. Increased expression of hexokinase I, increasing glucose sensing at low concentrations, and of *TXNIP*, increasing oxidative stress, may account for HDAC inhibition induced impairment. Conversely, the statin induced impairment is most likely a direct result of cholesterol lowering, as all diabetes related genes altered by statin treatment are regulated by transcription factors whose activity is linked to cellular cholesterol levels. Butyrate also interfered with insulin signalling, decreasing AKT phosphorylation and increasing IRS-1 phosphorylation, possibly by increasing acetylation of these proteins.

In contrast to the literature, statins did not inhibit the activity of epigenetic modifying enzymes, namely HDACs, DNA methyltransferases (DNMTs) or histone acetyltransferases (HATs). HDAC activity was investigated in numerous cell types, in live cells and cell extracts and in the livers of statin treated mice. Atorvastatin also failed to alter global histone acetylation in multiple cell types. In all cell-based experiments, the established HDAC inhibitors butyrate and trichostatin A (TSA) significantly inhibited HDAC activity and increased global histone acetylation. Despite having no effect on HDAC activity, atorvastatin tended to increase histone acetylation at SREBP-2 target genes, suggesting a mechanism by which statins may indirectly impact epigenetic modifications. The effect of butyrate on histone modifications in proximity to key genes implicated in cholesterol metabolism and T2D was dependent on the statistical methodology used to normalise ChIP-qPCR signals. Increased histone acetylation in proximity to the glucose transporter *SLC2A2* (GLUT2) was consistent across the different methods, suggesting that butyrate increases *SLC2A2* expression through HDAC inhibition.

Statins had much more limited effects on gene expression than HDAC inhibitors and all the noted effects of statins in this study are on-target effects of HMGCR inhibition and subsequent cellular cholesterol lowering. Conversely, HDAC inhibitors have more widespread effects on gene expression, altering the expression of many genes involved in cholesterol and glucose metabolism, in ways that would be expected to be beneficial and harmful for CVD and T2D.

Acknowledgements

This research was supported by an Australian Government Research Training Program (RTP) Scholarship, and funding by the School of Biomedical Sciences and the Curtin Medical School.

Copyright notice

Aspects of the research included in this thesis have been previously published. Where I did not hold permission to reuse the material as dictated by the journal's conditions, I obtained permission from the publishers. All aspects included in this thesis were my own work except where previously acknowledged in the Statement of Contributors. I have obtained permission from the co-authors of the articles. Portions of the following articles were used in this thesis:

<u>Allen SC</u>, Mamotte CDS. Pleiotropic and adverse effects of statins: do epigenetics play a role? *J Pharmacol Exp Ther* 2017; 362(2): 319-326. doi: 10.1124/jpet.117.242081. (Review)

Bridgeman SC, Ellison GC, Melton PE, Newsholme P, Mamotte CDS. Epigenetic effects of metformin: From molecular mechanisms to clinical implications. *Diabetes Obes Metab* 2018; 20: 1553-1562. doi: 10.1111/dom.13262. (Review)

Bridgeman S, Northrop W, Ellison G, Sabapathy T, Melton PE, Newsholme P, Mamotte CDS. Statins do not directly inhibit the activity of major epigenetic modifying enzymes. *Cancers* 2019; 11(4): 516. doi: 10.3390/cancers11040516.

<u>Bridgeman SC</u>, Northrop W, Melton PE, Ellison GC, Newsholme P, Mamotte CDS. Butyrate, generated by gut microbiota, and its therapeutic role in metabolic syndrome. *Pharmacol Res* 2020; 160: 105174. doi: 10.1016/j.phrs.2020.105174. (Review)

Bridgeman S, Ellison G, Newsholme P, Mamotte, C. The HDAC inhibitor butyrate impairs β cell function and activates the disallowed gene hexokinase I. *Int J Mol Sci* 2021; 22(24): 13330. doi: 10.3390/ijms222413330.

Contents

Author's I	Declaration	i
Statement	of contributors	ii
Acknowle	dgements	iii
Dedication	1	v
Abstract		vi
Acknowle	dgements	ix
Copyright	notice	x
List of Fig	ures	xvi
List of Tal	ples	xix
List of Ab	bre viations	xx
Chapter 1.	Introduction	25
1.1. Hy	potheses	25
1.2. Ob	jectives	25
1.3. Sig	nificance	
1.4. The	esis overview	
Chapter 2.	Literature Review	
2.1. Intr	roduction	
2.2. Cho	blesterol	
2.2.1.	Cholesterol metabolism	
2.2.2.	Cholesterol and disease risk	33
2.3. Glu	cose and diabetes	
2.3.1.	Glucose metabolism	
2.3.2.	Diabetes	37
2.3.3.	Cholesterol and diabetes	
2.4. Epi	genetics	39
2.4.1.	DNA methylation	41
2.4.2.	Histone modifications	41
2.4.3.	DNA methylation in cholesterol metabolism and T2D	43
2.4.4.	Histone acetylation in cholesterol metabolism and T2D	44
2.5. Sta	tins	45
2.5.1.	Statins and cholesterol	45
2.5.2.	Statins and diabetes	45

2.5	.3.	Statins and epigenetics	47
2.6.	Buty	yrate	49
2.6	.1.	Butyrate and cholesterol	51
2.6	.2.	Butyrate and diabetes	52
2.7.	Con	clusion	53
Chapter	3.	Methods and Materials	55
3.1.	Crea	ation of stock solutions	55
3.1	.1.	Creation of lipoprotein-deficient serum and high-lipoprotein serum	55
3.2.	Sam	ple preparation	55
3.2	.1.	Cell culture and treatments	55
3.2	.2.	Animal Study	56
3.2	.3.	Nuclear extraction	57
3.2	.4.	Whole cell protein extraction	57
3.3.	Viał	bility assays	57
3.4.	Lipi	d quantification	58
3.4	.1.	Lipid extraction	58
3.4	.2.	Cellular cholesterol content	58
3.4	.3.	Cellular triglyceride content	58
3.4	.4.	Cholesterol uptake and export assays	59
3.5.	Imn	nunoblotting	59
3.6.	Gen	e expression	61
3.6	.1.	RNA extraction	61
3.6	.2.	Reverse transcription	61
3.6	.3.	qPCR	62
3.7.	Insu	lin secretion	64
3.7	.1.	Insulin secretion assay	64
3.7	.2.	Cellular insulin content	65
3.7	.3.	Insulin quantification	65
3.8.	Insu	lin signalling	65
3.9.	Epig	genetic modifying enzyme activity assays	65
3.9	.1.	HDAC activity	65
3.9	.2.	HAT activity assays	66
3.9	.3.	DNMT activity assays	66
3.10.	С	hromatin immunoprecipitation	67
3.1	0.1.	ChIP qPCR	68
3.11.	S	tatistical analysis	69

Chapter 4	4. Cholesterol metabolism
4.1.	Abstract
4.2.	Objectives
4.3.	Introduction71
4.4.	Results
4.4.1 conte	1. Effects of statins and sodium butyrate on cell viability and protein ent72
4.4.2 statii	2. Sodium butyrate reduces cellular cholesterol to a similar extent as ns
4.4.3 conte	3. Neither sodium butyrate nor statins alter cellular triglyceride ent76
4.4.4 chole	4. Statins increase cholesterol uptake; sodium butyrate affects neither esterol uptake nor export
4.4.5 signa	5. Sodium butyrate and statins have opposite effects on SREBP-2 alling
4.4.6 reve	5. The effect of sodium butyrate and statins on proteins involved in rse cholesterol transport depends on cell type and media lipoproteins 82
4.4.7 gene	7. Sodium butyrate alters the expression of multiple cholesterol related es, atorvastatin has more targeted effects on gene expression
4.5.	Discussion
4.5.1	1. Mechanisms of cholesterol lowering by statins and butyrate
4.5.2 prote	2. Effects of atorvastatin and butyrate on reverse cholesterol transport eins
4.5.3 gene	3. Effects of butyrate and atorvastatin on expression of cholesterol
4.6.	Conclusion
Chapter 5	5. Glucose metabolism
5.1.	Abstract
5.2.	Objectives
5.3.	Introduction
5.4.	Results
5.4.1 conte	1. HDAC inhibition impairs insulin secretion and reduces insulin ent
5.4.2	2. Butyrate acts as an acute β cell secretagogue
5.4.3	3. Butyrate interferes with insulin signalling in hepatic cells
5.4.4 in H	4. Butyrate increases protein levels of the disallowed gene hexokinase I epG2 and BRIN-BD11 cells
544	5. Butvrate significantly alters expression of numerous T2D candidate

5.4.5. Butyrate significantly alters expression of numerous T2D candidate genes, atorvastatin has more limited effect on gene expression103

5.5. Discussion105
5.5.1. Effects of butyrate and statins on the expression of genes involved in diabetes and glucose metabolism
5.5.2. Effects of HDAC inhibition on insulin secretion107
5.5.3. Cholesterol lowering by statins as a mechanism for impaired insulin secretion
5.5.4. Effects of butyrate and statins on insulin signalling114
5.6. Conclusion115
Chapter 6. Epigenetics
6.1. Abstract117
6.2. Objectives117
6.3. Introduction118
6.4. Results118
6.4.1. Statins do not inhibit HDAC activity118
6.4.2. Statins do not inhibit HAT activity123
6.4.3. Sodium butyrate, but not statins, increase global histone acetylation124
6.4.4. Statins do not directly inhibit DNMT activity
6.4.5. Statins alter histone acetylation in specific genomic regions using ChIP-qPCR
6.4.6. Butyrate increases ChIP signals and H4 acetylation at numerous genomic regions
6.5. Discussion133
6.5.1. Sodium butyrate and curcumin inhibit epigenetic modifying enzymes
6.5.2. Statins do not directly inhibit epigenetic modifying enzymes134
6.5.3. Statins may impact epigenetic modifications in specific genomic regions
6.5.4. Results of ChIP-qPCR experiments of butyrate treated cells depend on analysis and normalisation methods
6.6. Conclusion140
Chapter 7. Concluding remarks
7.1. Future directions and perspectives144
7.2. Limitations146
7.3. Clinical relevance
7.4. Conclusion148
Chapter 8. References
Appendix A: Supplementary figures

Appendix B: qPCR optimisation	201
Appendix C: Data tables	227
Appendix D: Copyright Permissions	

List of Figures

Figure 2-1	Lipoprotein metabolism
Figure 2-2	Glucose metabolism
Figure 2-3	Epigenetic modifications 40
Figure 2-4	Postulated mechanisms by which statins may impact histone
acetylation	
Figure 2-5	Butyrate's mechanisms of action
Figure 4-1	Influence of statins and butyrate on the viability of HepG2 cells 73
Figure 4-2	Influence of butyrate and atorvastatin on protein content in $\ensuremath{HepG2}$
cells	
Figure 4-3	Influence of statins and butyrate on cellular cholesterol content in
HepG2 cells,	BRIN-BD11 cells and THP-1 cells75
Figure 4-4	Influence of butyrate and atorvastatin on cellular triglyceride content
in HepG2 cell	ls76
Figure 4-5	Influence of butyrate and atorvastatin on cholesterol uptake and export
in HepG2 cell	ls from HepG2 cells
Figure 4-6	Influence of butyrate and atorvastatin on apoprotein secretion from
HepG2 cells.	
Figure 4-7	Influence of butyrate and atorvastatin on SREBP-2 pathway in HepG2
and THP-1 ce	lls
Figure 4-8	Influence of HDAC inhibition on the SREBP-2 pathway at 8 h and 16 h
in HepG2 cell	ls
Figure 4-9	Influence of butyrate and atorvastatin on reverse cholesterol transport
proteins in He	pG2 cells, BRIN-BD11 cells and THP-1 cells
Figure 4-10	Influence of HDAC inhibition on ABCA1 at 8 h and 16 h in $\mbox{HepG2}$
cells	
Figure 4-11	Influence of butyrate and atorvastatin on cholesterol-related gene
expression in	HepG2 cells and BRIN-BD11 cells
Figure 4-12	Influence of butyrate on protein levels of upregulated genes in HepG2
cells	
Figure 5-1	Influence of HDAC inhibition on insulin secretion
Figure 5-2	Influence of acute sodium butyrate treatment on insulin secretion 98

Figure 5-4	Influence of butyrate and atorvastatin on glucose metabolism proteins
in HepG2 cell	s101
Figure 5-5	Influence of butyrate and atorvastatin on hexokinases in BRIN-BD11
cells	
Figure 5-6	Influence of HDAC inhibition on hexokinase I in HepG2 cells at 8 h
and 16 h	
Figure 5-7	Influence of butyrate and atorvastatin on diabetes-related gene
expression	
Figure 5-8	Hexokinase and glucokinase activity109
Figure 5-9	Summary of effects of chronic butyrate treatment on insulin
secretion	
Figure 6-1	Influence of statins and butyrate on HDAC activity in HepG2
cells	
Figure 6-2	Influence of atorvastatin on HDAC activity in HepG2 cells120
Figure 6-3	Influence of statins and butyrate on HDAC activity in MDA-MB-231
cells and BRI	N-BD11 cells121
Figure 6-4	Influence of statins and HDAC inhibitors on HDAC activity using the
Fluor De Lys	HDAC activity kit122
Figure 6-5	Influence of statin and diet on HDAC activity in mice livers123
Figure 6-6	Influence of atorvastatin on HAT activity123
Figure 6-7	Influence of butyrate, atorvastatin and curcumin on global histone
modifications	in HepG2 cells125
Figure 6-8	Influence of butyrate and atorvastatin on global histone modifications
in THP-1 cell	s126
Figure 6-9	Influence of atorvastatin on DNMT activity in HepG2 cell
extracts	
Figure 6-10	Influence of atorvastatin on histone acetylation in proximity to
specific genor	nic regions129
Figure 6-11	Influence of butyrate on histone signals in ChIP-qPCR130
Figure 6-12	Influence of butyrate on histone signals in ChIP-qPCR relative to total
Н3	
Figure 6-13	Influence of butyrate on histone acetylation in proximity to SLC2A2
and HMGCR.	
Figure 6-14	Postulated mechanism by which statins may impact histone
acetylation	

Figure 7-1	Postulate	mechanisms	by	which	statins	and	butyrate	impact	cellular
cholesterol an	d insulin s	ecretion							149

List of Tables

Table 2.1	A selection of key genes involved in cholesterol metabolism	
Table 2.2	A selection of key genes involved in glucose metabolism	
Table 2.3	Common histone modifications	
Table 3.1	Antibodies used in immunoblotting 60	
Table 3.2	Human primers for qPCR experiments in HepG2 cells	
Table 3.3	Rat primers for qPCR experiments in BRIN-BD11 cells 64	
Table 3.4	Antibodies used for ChIP	
Table 3.5	Primers used for ChIP qPCR 69	
Table 4.1	Influence of butyrate and statins on ABCA1 protein levels in different	
cell types		
Table 4.2	Influence of butyrate and atorvastatin on a selection of genes involved	
in cholester	rol metabolism	
Table 5.1	Influence of butyrate and atorvastatin on a selection of genes involved	
in glucose metabolism and T2D105		

List of Abbreviations

ABCA1	ATP-binding cassette transporter sub-family A member 1
ABCG1	ATP-binding cassette transporter sub-family G member 1
ADP	Adenosine diphosphate
AKT	Protein kinase B
ANOVA	Analysis of variance
APO	Apolipoprotein
ATAC-seq	Assay of Transposase Accessible Chromatin sequencing
ATP	Adenosine triphosphate
BCA	Bicinchoninic acid
BCP	1-Bromo-3-chloropropane
bp	Base pairs
BSA	Bovine serum albumin
cAMP	Cyclic adenosine monophosphate
CBP	CREB binding protein
cDNA	Complementary DNA
ChIP	Chromatin immunoprecipitation
СоА	Coenzyme A
CREB	cAMP response element binding protein
CVD	Cardiovascular disease
CYP7A1	Cholesterol 7 alpha-hydroxylase/Cytochrome P450 family 7 subfamily A member 1
DMEM	Dulbecco's modified Eagle's medium
DMSO	Dimethyl sulfoxide

DNA	Deoxyribonucleic acid
DNMT	DNA methyltransferase
DTT	Dithiothreitol
ELISA	Enzyme-linked immunosorbent assay
EWAS	Epigenome wide association study
EZH2	Enhancer of zeste homolog 2
FBS	Foetal bovine serum
FFAR	Free fatty acid G-coupled receptor
FOXO1	Forkhead box class O1
G6PC/G6Pase	Glucose 6-phosphatase
GLP-1	Glucagon-like peptide 1
GLUT2/SLC2A2	Glucose transporter 2
GLUT4/SLC2A4	Glucose transporter 4
GSIS	Glucose-stimulated insulin secretion
НАТ	Histone acetyltransferase
HDAC	Histone deacetylase
HDL	High-density lipoprotein
HDL-C	HDL cholesterol
HEPES	4-(2-hydroxyethyl)-1-piperazineethanesulfonic acid
HFD	High fat diet
HLPS	High lipoprotein serum
HMG-CoA	3-hydroxy-3-methyl-glutaryl-coenzyme A
HMGCR	HMG-CoA reductase
НМТ	Histone methyltransferase

HNF4α	Hepatocyte nuclear factor 4α
HRP	Horseradish peroxidase
IC50	Half maximal inhibitory concentration
IDL	Intermediate-density lipoprotein
INS	Insulin
IR	Insulin receptor
IRS-1	Insulin receptor substrate 1
К	Lysine
KRBB	Krebs-Ringer Bicarbonate buffer
LDL	Low-density lipoprotein
LDL-C	LDL cholesterol
LDLR	Low-density lipoprotein receptor
LDPS	Lipoprotein-deficient serum
LXR/NR1H3	Liver receptor X
MetS	Metabolic syndrome
mRNA	Messenger RNA
NAD	Nicotinamide adenine dinucleotide
NAFLD	Non-alcoholic fatty liver disease
ND	Normal diet
PBS	Phosphate-buffered saline
PDX1	Pancreatic and duodenal homeobox 1
РЕРСК	Phosphoenolpyruvate carboxykinase
PGC1a/PPARGC1A	PPARγ coactivator 1 alpha
PIC	Protease inhibitor cocktail

PPARy/PPARG	Peroxisome proliferator-activated receptor gamma
РҮҮ	Peptide YY
qPCR	Quantitative real-time polymerase chain reaction
R	Arginine
RCT	Reverse cholesterol transport
RIPA	Radioimmunoprecipitation assay (lysis buffer)
RNA	Ribonucleic acid
RPMI	Roswell Park Memorial Institute (media)
RT	Room temperature
SAM	S-adenosylmethionine
SB	Sodium butyrate
SCFA	Short-chain fatty acid
SE	Standard error
SEM	Standard error of the mean
SIRT	Sirtuin
SRB1/SCARB1	Scavenger receptor class B type 1
SREBP/SREBF	Sterol regulatory element-binding protein/factor
T2D	Type 2 diabetes mellitus
TSBT	Tris-buffered saline with Tween 20
TG	Triglyceride
tRNA	Transfer RNA
TSA	Trichostatin A
TSS	Transcription start site
TXNIP	Thioredoxin-interacting protein

VLDL

Chapter 1. Introduction

This thesis explores the possible links between epigenetics, cholesterol lowering and type 2 diabetes (T2D). The rationale for this is that the widely used cholesterol lowering statin class of drugs have been shown to be associated with a small but significant increase in T2D and have been reported to have epigenetic effects, in particular inhibition of histone deacetylases (HDACs), but whether epigenetic effects could contribute to their diabetogenic effects has not been explored in detail. HDAC inhibitors, most notably the short-chain fatty acid (SCFA) butyrate, have also been shown to improve lipid profiles in animal models of metabolic disease, however, in contrast to statins, they have also been reported to be protective against diabetes in these animals. The work described in this thesis aimed to compare and contrast the effects of statins and established HDAC inhibitors, primarily butyrate, on cholesterol metabolism, glucose metabolism and epigenetic modifications.

1.1. Hypotheses

The objectives of this thesis are grounded in two hypotheses

- 1. Epigenetic modifications contribute to the diabetogenic effects of statins
- 2. Butyrate lowers cholesterol and protects against diabetes as a result inhibiting HDAC activity in cells such as hepatocytes and the insulin secreting β cells

1.2. Objectives

- 1. Investigate how statins and HDAC inhibitors affect cholesterol metabolism
- 2. Investigate how these treatments affect insulin secretion and signalling.
- Investigate how these treatments influence epigenetic modifying enzymes and alter genome-wide epigenetics.

4a. Identify genes involved in cholesterol metabolism and implicated in T2D whose expression is altered by these treatments.

4b. Investigate if epigenetic modifications contribute to the altered expression of these target genes.

1.3. Significance

Cardiovascular disease (CVD) and T2D are two of the biggest causes of morbidity and mortality in the modern world and both are associated with altered cholesterol metabolism. As a result, the low-density lipoprotein (LDL) cholesterol lowering drugs known as statins are amongst the most prescribed medications in many advanced economies. However, statins are associated with a small but significant increased risk of developing T2D which, due to the widespread use of statins, may affect millions of users worldwide. Understanding the mechanisms by which statins alter diabetes risk may lead to identification of those most at risk and the development of mitigating strategies in high-risk individuals. In contrast to statins, epigenetic modifying agents, most notably HDAC inhibitors, have been reported to not only lower cholesterol but also protect against diabetes in animal models of metabolic disorders. If these results are translatable to humans and HDAC inhibitors are able to either replace or complement statin use, this could potentially benefit many millions of people at risk of CVD and T2D. This thesis makes a substantial contribution to the body of knowledge of how statins impact epigenetic modifications and of how HDAC inhibitors impact cholesterol and glucose metabolism.

1.4. Thesis overview

Chapter 2 provides the background to the problems explored in the subsequent chapters. It introduces the three major areas of exploration described in this thesis, epigenetics, cholesterol and T2D, and discusses the current understanding of links between the three. The effects of statins and the HDAC inhibitor butyrate on epigenetics, cholesterol and glucose metabolism are discussed, highlighting the gaps and inconsistencies in the current literature.

Chapter 3 details the methods and materials used throughout the thesis. A cell-culture based model was used, using cell types important to diabetes and cholesterol, most notably hepatic cells due to their importance in both cholesterol and glucose metabolism, pancreatic β cells due to the central role of insulin secretion in diabetes, and macrophages due to their role in the formation of cholesterol plaques in CVD. Despite the limitations of using immortalised cell lines and in extrapolating the results of *in vitro* studies to *in vivo* effects, this approach was chosen for several reasons. Firstly, many studies have explored the *in vivo* effects of statins and HDAC inhibitors

on cholesterol and glucose metabolism, in the case of statins the effect on humans *in vivo* is well-documented while with HDAC inhibitors most of the evidence comes from animal models. What these studies lack is exploration into the specific mechanisms governing these effects. The difficulty in exploring exacting mechanisms *in vivo* is that effects in one organ may have downstream effects on multiple tissues. For example, HDAC inhibitors such as butyrate have significant impact on intestinal cells, which has downstream effects on multiple cell types including hepatocytes and β cells. The work described in this thesis thus aimed to cover gaps in the literature by examining the possible mechanisms behind the observed *in vivo* effects without the compounding factors of whole-body effects.

While multiple statins and HDAC inhibitors were used in various experiments, atorvastatin and butyrate were chosen as the exemplar statin and HDAC inhibitor respectively for more detailed study. Curcumin was also used in preliminary experiments due to its reported direct inhibitory effects on multiple epigenetic modifying enzymes,¹ however experimentation revealed it was unsuitable for further study due to its lack of effect on cells *in vitro*, most likely due to its instability in cell culture media.² Atorvastatin was chosen as it was the most commonly prescribed statin in numerous advanced economies, including the United States,³ England,⁴ and Australia,⁵ and, being lipophilic may be more representative of other statins such as simvastatin, fluvastatin, lovastatin and pitavastatin, while the two hydrophilic statins, rosuvastatin and pravastatin, vary greatly in potency and reported diabetes risk. Butyrate was chosen as it has been widely explored in animal studies, has been used in multiple human clinical trials for gastrointestinal disorders where it has shown a favourable safety and tolerance profile and has been used in small-scale clinical trials for diabetes and metabolic syndrome.^{6,7}

Chapter 4 compares and contrasts the effects of statins and HDAC inhibitors on cellular cholesterol metabolism (Objective 1). In particular, this section of the thesis represents the most comprehensive study of the HDAC inhibitor butyrate on cholesterol metabolism. This includes looking at cellular cholesterol content in a range of cell types and conditions and the uptake and export of cholesterol from cells. Effects of treatments on proteins involved in cholesterol synthesis, uptake and reverse cholesterol transport are examined in a range of cell types and conditions, thus exploring the mechanisms by which statins and HDAC inhibitors lower cellular cholesterol and impact cholesterol transport. Key genes involved in cholesterol

metabolism whose expression is altered by statins or butyrate are identified (Objective 4a).

Chapter 5 describes the effects of statins and HDAC inhibitors on insulin secretion in pancreatic cells and insulin signalling in hepatic cells (Objective 2). Key proteins involved in glucose sensing are examined and key genes that have been implicated in T2D whose expression is altered by statins or butyrate are identified (Objective 4a), in order to examine the potential mechanisms by which statins and HDAC inhibitors impact insulin secretion and T2D.

Chapter 6 explores the effects of statins, HDAC inhibitors and curcumin on epigenetic modifications (Objective 3). In particular, extensive experimentation was conducted into effects on HDAC activity using multiple cell types and experimental protocols; this is the most comprehensive study on the effects of statins on HDAC activity. It also contains the most comprehensive study on the effects of statins on DNA methyltransferase (DNMT) activity as well as an examination of the effects of statins on histone acetyltransferase (HAT) activity, which has not been reported in the literature. This chapter also examines the effects of atorvastatin, sodium butyrate and curcumin on global histone modifications. Finally, the mechanism by which statins may indirectly impact epigenetic modifications are explored, as are the effects of atorvastatin and butyrate on histone acetylation in proximity to key genes altered by these treatments (Objective 4b).

Chapter 7 summarises the significant findings and implications of this thesis, addresses the limitations of the methodologies used and discusses ways in which future research can address these limitations and fill in remaining gaps in the literature. It also discussed the potential of using epigenetic modification in therapies for metabolic disorders.

Chapter 2. Literature Review

2.1. Introduction

CVD is the leading cause of mortality worldwide according to the World Health Organisation,⁸ while T2D is also a major cause of morbidity and mortality in the modern world. The International Diabetes Foundation estimates that 463 million adults worldwide were living with diabetes,⁹ and 4.2 million adults died of the disease in 2019.¹⁰ Dyslipidaemia, typically characterised by elevated levels of LDL cholesterol (LDL-C) and triglycerides (TG) and/or decreased levels of high-density lipoprotein cholesterol (HDL-C), is a major risk factor for both CVD and T2D.¹¹ As over five million Australians have dyslipidaemia,¹² the burden of CVD and T2D may be significantly lessened by improving the population's cholesterol levels. As a result, HMG-CoA reductase (HMGCR) inhibitors, commonly known as statins, are widely used to lower LDL-C. Despite the proven ability of statins to reduce morbidity and mortality from CVD by approximately 25%,¹³ meta-analyses of clinical trials have found statin use is associated with a small but significant increase in the risk of developing T2D.¹⁴

There has been much research linking epigenetics and developmental programming to dyslipidaemia,^{15,16} insulin resistance and T2D,¹⁷⁻²⁰ and CVD.²¹⁻²³ As a result, there is interest in the use of epigenetic modifiers, in particular HDAC inhibitors, to prevent and treat these disorders.^{24,25} Butyrate, a HDAC inhibitor produced by the gut microbiome, has shown promising results in pre-clinical experiments and has reached clinical trials for lipid lowering and T2D,^{6,26,27} although there is still significant uncertainty over how HDAC inhibitors influence cholesterol and glucose metabolism.

There is also growing interest in how existing medications influence epigenetic processes. For example, statins have been reported to affect numerous epigenetic modifying enzymes and the subsequent epigenetic modifications have been linked to protection from CVD and cancer.²⁸ There has been little research into whether epigenetic modifications contribute to the adverse effects of statins, including T2D, or if statins alter epigenetic mechanisms regulating cholesterol metabolism genes. This chapter outlines cholesterol metabolism, glucose metabolism and epigenetics, and the current literature regarding statins and butyrate and their effects on epigenetic modifications, cholesterol metabolism and T2D.

2.2. Cholesterol

2.2.1. Cholesterol metabolism

Cholesterol is a lipid sterol (steroid precursor) that is essential to animal cells due to its vital role in plasma membranes. In addition, cholesterol is a precursor to steroid hormones, bile acids and vitamin D.²⁹ Excess cholesterol, however, is a contributory factor to some of the most common chronic diseases in the modern world, most notably CVD, but also non-alcoholic fatty liver disease (NAFLD)³⁰ and T2D.³¹ Cholesterol is both synthesised by cells and sourced from the diet, and transported around the body in lipoproteins. Lipoproteins provide a means of transporting water insoluble lipids through the bloodstream by forming complexes containing amphipathic proteins known as apolipoproteins.²⁹

Absorption of dietary cholesterol by enterocytes in the small intestine is regulated by Niemann-Pick-C1-like-1 protein (NPC1L1).³² The cholesterol, together with TG, phospholipids and apolipoproteins, is packaged into chylomicrons, apo-B48 containing lipoproteins, for release into circulation. Chylomicrons deliver TG to peripheral cells such as adipose and muscle cells, leaving behind cholesterol-enriched chylomicron remnants, which are taken up by the liver.³³ Enterocytes also secrete cholesterol back into the intestinal lumen through ATP-binding cassette transporter sub-family G member 5 (ABCG5) and member G 8 (ABCG8).

In addition to dietary sources, cholesterol is synthesised by all nucleated cells. Cholesterol synthesis is a complicated process comprising of more than 30 chemical reactions.³² To summarise, three molecules of acetyl-CoA are reversibly condensed to form 3-hydroxy-3-methylglutaryl-CoA coenzyme A (HMG-CoA). HMG-CoA is converted to mevalonate by HMGCR, the rate-limiting enzyme in cholesterol biosynthesis and the target of statins. Mevalonate is then converted to isopentenyl pyrophosphate, the precursor not only to cholesterol, but also many other sterol and non-sterol isoprenoids including vitamin K, heme and ubiquinone.³⁴ In the cholesterol pathway, isopentenyl pyrophosphate is converted to squalene, then lanosterol, and then cholesterol.

The liver packages cholesterol from multiple sources together with TG and apolipoproteins into very low-density lipoprotein (VLDL) which, similarly to chylomicrons, delivers TG to peripheral cells and leaves behind intermediate-density lipoprotein (IDL).³⁵ IDL is either taken up by hepatic cells or further metabolised to form cholesterol-rich LDL, the primary source of plasma cholesterol in humans. Cholesterol is taken up by cells according to requirements through the LDL receptor (LDLR), which recognises the apolipoprotein apo-B100 in LDL.³⁶

Cholesterol is removed from peripheral cells in a process known as reverse cholesterol transport (RCT).³² In this process, ATP-binding cassette transporter sub-family A member 1 (ABCA1) mediates the efflux of cholesterol to apolipoprotein apo-A1 to form HDL. Mature HDL particles also take up cellular cholesterol through ATP-binding cassette transporter sub-family G member 1 (ABCG1) and scavenger receptor class B type 1 (SRB1). SRB1 can also take up cholesterol from HDL, particularly in the liver, where it can be recycled, converted to bile acids or excreted through bile as free cholesterol.³³

The balance between cholesterol uptake, synthesis and efflux is regulated by transcription factors, including sterol regulatory element-binding proteins (SREBPs) and liver receptor X (LXR), that increase or decrease expression of cholesterol metabolism genes in response to altered cellular cholesterol levels. There are two SREBP genes encoding three proteins: SREBP-1a, SREBP-1c and SREBP-2. SREBP-2 activates cholesterol synthesis and uptake genes, while SREBP-1c regulates fatty acid synthesis and SREBP-1a is involved in both cholesterol and fatty acid metabolism.³⁷ In its inactive state, SREPB-2 is retained in the endoplasmic reticulum (ER) membrane, where it associates with SREBP cleavage activating protein (SCAP). When ER membrane sterol levels drops, the SCAP-SREBP-2 complex is released from the ER membrane and transported to the Golgi apparatus. Here, SREBP-2 is cleaved by two resident proteases, site-1 (S1P) and site-2 (S2P), releasing the active domain and allowing it to translocate to the nucleus, where it activates transcription of LDLR and genes in the cholesterol biosynthesis pathway, including HMGCR, to increase cholesterol uptake and synthesis, and thus increase cellular cholesterol.³² Conversely, LXR is activated by oxysterols, which reflect cellular cholesterol levels, to promote cholesterol efflux and clearance. LXR target genes include ATP-binding cassette transporters ABCA1, ABCG1, ABCG5 and ABCG8 in addition to CYP7A1, which is the rate-limiting step in the conversion of cholesterol to bile acid, and genes involved in fatty acid synthesis and TG metabolism.³⁸ When oxysterols are present LXR activates these target genes, however, in the absence of oxysterols, LXR instead acts to repress target genes.³⁹

Cholesterol transport pathways are summarised in Figure 2-1 and important proteins involved in cholesterol metabolism are summarised in Table 2.1.





Dietary cholesterol (C) is taken up by enterocytes and packaged into chylomicrons together with other constituents including TG, phospholipids and apolipoproteins. The TG component is hydrolysed by lipoprotein lipase into monoacylglycerol and free fatty acids which can be taken up by peripheral tissues, particularly fat and muscle tissue, and the cholesterol-rich chylomicron remnant is delivered to the liver. The liver packages cholesterol and TG into very low-density lipoproteins (VLDL). TG from VLDL is hydrolysed and taken up by peripheral cells, resulting in low-density lipoproteins (LDL) which deliver cholesterol to cells. Cholesterol is exported from peripheral cells to the liver in high density lipoproteins (HDL). Cholesterol is eliminated from the body through bile.

Table 2.1	A selection	of key genes	involved	in cholesterol	metabolism
-----------	-------------	--------------	----------	----------------	------------

Gene	Protein	Function
NPC1L1	Niemann-Pick-C1-like-1	Absorption of dietary cholesterol in
	protein	intestine.
HMGCR	HMG-CoA reductase	Rate-limiting enzyme in cholesterol
		biosynthesis.

Gene	Protein	Function
LDLR	LDL receptor	Binds LDL and allows cellular uptake of
		cholesterol.
АРОВ	Apolipoprotein B-100	Component of VLDL and LDL, mediates
		binding of LDL to cell surface LDLR.
	Apolipoprotein B-48	Component of chylomicrons
APOA1	Apolipoprotein A1	Component of HDL, reverse cholesterol
		transport from peripheral cells to liver.
ABCA1	ATP binding cassette	Transport of cholesterol from cells to
	subfamily A member 1	apo-A1.
ABCG1	ATP binding cassette	Transport of cholesterol, including
	subfamily G member 1	within cells and export to HDL.
SCARB1	Scavenger receptor class	HDL receptor, bidirectional of transfer of
	B member 1 (SRB1)	cholesterol between cells and HDL.
CYP7A1	Cytochrome P450 family	Rate-limiting enzyme in bile acid
	7 subfamily A member 1	synthesis, allows for cholesterol
		elimination.
SREBF2	Sterol regulatory element-	Master transcription factor for proteins
	binding protein 2	involved in cholesterol synthesis and
	(SREBP-2)	cellular cholesterol uptake.
SREBF1	Sterol regulatory element-	Transcription factor for fatty acid
	binding protein 1	synthesis (SREBP-1c) and/or cholesterol
	(SREBP-1a or c)	metabolism (SREBP-1a).
NR1H3	Liver X receptor (LXR)	Transcription factor for cholesterol
		transport and efflux.

2.2.2. Cholesterol and disease risk

Although all animals require cholesterol to function, excess levels of cholesterol and impaired cholesterol metabolism are linked to a number of common diseases. In fact, cholesterol was first discovered in 1769 as a component of gallstones,⁴⁰ and it is most notorious today for its link to CVD. In particular, high levels of LDL-C and low levels of HDL-C are strongly associated with mortality from coronary heart disease;²⁹ verified by the efficacy of statins in reducing cardiovascular events in proportion to

LDL-C lowering.⁴¹ Although various models have been proposed, the initiation of atherosclerosis is classically associated with the accumulation of LDL in the subendothelial space of arteries.⁴² Activation of endothelial cells may cause oxidation of LDL and the recruitment of monocytes, which then differentiate into proinflammatory macrophages. Uptake of LDL, especially oxidised LDL, and reduced cholesterol efflux to HDL by the macrophages may result in their transformation into cholesterol laden foam cells.⁴³ The macrophages may promote the infiltration and proliferation of smooth muscle cells, which can contribute to the formation of atherosclerotic plaques by producing a fibrous cap that surrounds a core of cholesterol laden foam cells, extracellular cholesterol and other debris from dead cells.⁴² Rupture of atherosclerotic plaques can lead to thrombus formation and blockage of vessels causing myocardial infarction or stroke.

Cholesterol has also been implicated in numerous other diseases, including NAFLD, where the accumulation of free cholesterol is thought to be a major contributor to liver damage,³⁰ and neurodegenerative disorders such as Alzheimer's disease, Huntington's disease and Parkinson's disease.⁴⁴ Cholesterol is also linked to T2D, as discussed in detail in section 2.3.3.

2.3. Glucose and diabetes

Glucose is the primary energy source of mammalian cells and thus is vital for health, although elevated concentrations have detrimental effects on numerous organs. Thus, there are numerous pathways and hormones that act to maintain plasma glucose levels within a certain range.

2.3.1. Glucose metabolism

Following a meal, glucose is taken up by enterocytes primarily through the sodiumdependent glucose co-transporter SGLT1 and exported to the bloodstream by the glucose transporter GLUT2, which has a low affinity for glucose and thus is activated when glucose levels are high.⁴⁵ Circulating glucose is then taken up by pancreatic β cells and liver hepatocytes through GLUT2. The rise in glucose levels is recognised by these cells due to the glucose sensing protein glucokinase, also known as hexokinase IV, which also has a low affinity for glucose and thus phosphorylates glucose only when glucose levels are high.⁴⁶ In other cells types, glucose phosphorylation is carried out by hexokinases I, II and III, which respond to lower levels of glucose and are usually silenced in β cells and hepatocytes to prevent insulin secretion and glycogenesis, respectively, when glucose levels are low.⁴⁷ β cells respond to increased glucose levels by secreting the hormone insulin in a process known as glucose-stimulated insulin secretion (GSIS). Insulin binds to insulin receptors in peripheral cells and initiates a series of phosphorylation events that results in skeletal muscle and adipocytes increasing their uptake of glucose through the insulin-responsive glucose transporter GLUT4. These cells subsequently use glucose for cellular energy production through glycolysis, the breakdown of glucose to produce pyruvate and ATP.⁴⁸ In the liver, insulin and glucose promote the production of glycogen, a storage form of glucose, in a process known as glycogenesis.⁴⁹ This process is enhanced by the fact that insulin also promotes the expression of glucokinase in the liver.⁵⁰

In fasting conditions, the liver plays a crucial role in generating glucose for peripheral tissues through the breakdown of glycogen, known as glycogenolysis, and the *de novo* production of glucose, known as gluconeogenesis. These processes are promoted by the hormone glucagon, released by pancreatic α cells in low glucose and insulin conditions,⁵¹ and inhibited by insulin in high glucose conditions.^{52,53} Glycogenolysis is the primary glucose source during short-term fasting, such as overnight, and can provide fuel for approximately 30 hours, following which gluconeogenesis is required to provide energy for cells.⁵⁴ Gluconeogenesis produces glucose from non-carbohydrate substrates including amino acids (from proteins), glycerol (from lipids) and lactate (from metabolic processes) through enzymes including glucose 6-phosphatase (G6Pase) and phosphoenolpyruvate carboxykinase (PEPCK).

Many other proteins play important roles in glucose metabolism including hormones, such as amylin from β cells and glucagon-like peptide-1 (GLP-1) from intestinal cells, and transcription factors such as pancreatic and duodenal homeobox 1 (PDX1), forkhead box class O1 (FOXO1), hepatocyte nuclear factor 4 α (HNF4 α) and peroxisome proliferator-activated receptor gamma coactivator 1 alpha (PGC1 α).^{53,54} Glucose metabolism is summarised in Figure 2-2 and the key proteins involved summarised in Table 2.2.


Figure 2-2 Glucose metabolism

For proper functioning of cells, it is important plasma glucose levels are maintained within a limited range (4.0-6.0 mM). Pancreatic islet cells play an important part in maintaining this balance. In conditions of fasting when plasma glucose levels are low, pancreatic α cells release glucagon, which promotes the liver to break down glycogen to release glucose, while also stimulating the liver to synthesise *de novo* glucose (gluconeogenesis), resulting in the release of glucose into the bloodstream for peripheral cells to utilise. When glucose levels rise post feeding, pancreatic β cells release insulin, which promotes glucose uptake in peripheral cells and in the liver promotes glucose storage in the form of glycogen, resulting in a decrease of circulating glucose.

 Table 2.2
 A selection of key genes involved in glucose metabolism

Gene	Protein	Function			
GK	Glucokinase	Phosphorylates glucose in β cells and			
		hepatocytes in high glucose conditions.			
SLC2A2	Glucose transporter	Transports glucose into β cells and hepatocytes			
	type 2 (GLUT2)	in high glucose conditions.			

Gene	Protein	Function		
INS	Insulin	Released by β cells in response to high glucose,		
		promotes glucose uptake and utilisation.		
SLC2A4	Glucose transporter	Transports glucose into adipocytes and		
	type 4 (GLUT4)	myocytes in response to insulin.		
GCG	Glucagon	Released by α cells in response to low insulin		
		and glucose, promotes glycogenolysis and		
		gluconeogenesis.		
РЕРСК	Phosphoenolpyruvate	Converts oxaloacetate into		
	carboxykinase	phosphoenolpyruvate in first committed step of		
		gluconeogenesis.		
G6PC	Glucose	Hydrolyses glucose 6-phosphate in final step		
	6-phosphatase	of gluconeogenesis.		
	(G6Pase)			
PDX1	Pancreatic and	Transcription factor important in β cell		
	duodenal homeobox 1	development and function.		
FOX01	Forkhead box protein	Transcription factor important in		
	01	gluconeogenesis, glycogenolysis, cell cycle		
		regulation.		
HNF4A	Hepatocyte nuclear	Transcription factor important in hepatocyte		
	factor 4a	and β cell development and function.		
PPARGC1A	Peroxisome	Transcription co-activator whose targets		
	proliferator-activated	include LXR, HNF4, FOXO1 and		
	receptor gamma	gluconeogenic genes.		
	coactivator 1 alpha			
	(PGC1a)			

2.3.2. Diabetes

In diabetes, failure to properly regulate glucose levels results in chronically elevated blood glucose levels, known as hyperglycaemia. Type 1 diabetes mellitus (T1D), or insulin dependent diabetes, occurs when the insulin secreting β cells are destroyed in an auto-immune reaction and usually occurs in childhood. T2D usually occurs in adulthood when pancreatic β cells are unable to cope with the increased demand for insulin due to the insulin resistance of peripheral cells, resulting in chronic

hyperglycaemia. Insulin resistance is the failure of cells to respond properly to insulin, resulting in reduced glucose uptake, particularly in skeletal cells, and failure to suppress gluconeogenesis in the liver, further increasing blood glucose.⁵⁵ Insulin resistance has been shown to occur due to a reduction in the number of insulin receptors on the cell surface, resulting in reduced insulin binding, and/or the impairment of downstream phosphorylation events.⁵⁶

As a result of insulin resistance, β cells typically initially increase insulin secretion to compensate for insulin resistance, this may be associated with an increase in β cell mass. Progression from insulin resistance to T2D arises when β cell dysfunction occurs, typically due to the accumulative exposure to high levels of glucose and fatty acids, known as glucolipotoxicity, and the action of proinflammatory cytokines associated with insulin resistance and obesity, leading to reduced insulin secretion and increased β cell death.⁵⁷

The result is chronic hyperglycaemia, which damages blood vessels, leading to macrovascular (CVD) and microvascular (renal failure, blindness and neuropathy) complications.⁵⁸ Development of T2D depends on a number of risk factors including genetic disposition and lifestyle factors such as diet and lack of exercise.⁵⁹ Cholesterol and epigenetic modifications have also been linked to the development of T2D, as discussed in sections 2.3.3 and 2.4 respectively.

2.3.3. Cholesterol and diabetes

Dyslipidaemia is strongly linked to T2D; men with dyslipidaemia are at twice the risk of developing T2D, while for women the risk is tripled.¹¹ In particular, diabetic dyslipidaemia is classically associated with high levels of TG and small dense LDL particles, and low levels of HDL, thought to be a result of hepatic insulin resistance leading to uncontrolled secretion of apo-B containing lipoproteins and impaired clearance of lipids.⁶⁰

Treating β cells with LDL^{61,62} or cholesterol⁶³ has been found to reduce insulin secretion *in vitro*, effects that were prevented by HDL or cholesterol depletion, respectively. Reducing islet levels of several genes involved in cholesterol metabolism, namely LXR, ABCA1,^{31,64} or ABCG1,⁶⁴ or increasing levels of SREBP-2,⁶⁵ all of which would increase cholesterol accumulation in β cells, has been

found to impair insulin secretion in animal models. Furthermore, several *ABCA1* variants have been linked with an increased risk of T2D in humans.^{31,64}

In addition to impaired insulin secretion, dyslipidaemia may impair insulin signalling in peripheral tissues. Llanos et al.⁶⁶ reported increased cholesterol levels in muscle from mice fed a high fat diet (HFD). Depletion of cholesterol from these isolated muscle fibres with cyclodextrin resulted in increased glucose uptake, both in the presence and absence of insulin. Similarly, HDL and apo-A1 have been reported to increase glucose uptake in muscle cells and adipocytes, and, in a human trial, infusion of HDL acutely decreased blood glucose in diabetics, suggesting increased insulin sensitivity.^{67,68}

Reduced cholesterol levels, however, have also been associated with T2D. In particular, the use of statins to lower LDL-C has been found to increase the risk of developing T2D, as later discussed in section 2.5.2. Furthermore, people with familial hypercholesterolaemia due to mutations in the *LDLR* gene or *APOB* gene have half the risk of developing T2D compared to unaffected relatives when adjusted for other risk factors,⁶⁹ and people with mutations in the *HMGCR* gene that result in lower plasma LDL-C levels have an increased risk of T2D.⁷⁰ Previous studies by our research group found that either increased or decreased cellular cholesterol impairs insulin binding in hepatic and muscle cells,⁷¹ and insulin secretion from pancreatic cells.⁷²

2.4. Epigenetics

Epigenetics is a rapidly growing field in medical research. Epigenetic modifications, changes to DNA structure that alter gene expression without altering the base nucleotide code, are crucial in the development of organisms and the differentiation and function of specific cell types. For example, epigenetic modifications permit the expression of the insulin gene in pancreatic β cells, while silencing it in other cells.⁷³ Epigenetic modifications have also been associated with numerous disorders, including T2D, CVD and cancer.^{17,74} Epigenetic processes can be altered by environmental factors including diet, exercise and exposure to toxins, with the foetal environment playing a particularly important role in influencing epigenetic modifications that impact metabolism in adult life. Both maternal nutrient deficiency and maternal obesity have been linked to epigenetic changes and the development of T2D in offspring.⁷⁵ There is growing evidence that pharmaceuticals also alter

epigenetic processes, which may contribute to the beneficial and deleterious effects of widely-used medications.⁷⁶ This includes the widely used anti-diabetic drug metformin, as reviewed in Bridgeman et al.⁷⁷ Epigenetic modifications fall under two main categories: histone modifications and DNA methylation as summarised in

Figure 2-3. RNA mediated interference, such as microRNAs, are often included as a third category, but will not be discussed here.



Figure 2-3 Epigenetic modifications

Chromatin consists of DNA wound around histone cores consisting of histone proteins H2A, H2B, H3 and H4. Compaction of chromatin reduces access of transcription factors and other proteins to DNA and is associated with transcriptionally silent genes. More relaxed, 'active' chromatin, mediated by modifications such as a cetylation (Ac) and methylation (Me) of residues on histone

tails protruding from the nucleosome, allows easier access to DNA and is associated with active transcription. Methylation of cytosines in CpG dinucleotides represses mRNA transcription by blocking transcription factors (TF) and/or interacting with histones. Originally published in Allen and Mamotte.²⁸

2.4.1. DNA methylation

DNA methylation when methyl transferred occurs groups are from S-adenosylmethionine (SAM) to nucleotides by DNMTs. In vertebrates, it is thought that methylation only occurs to cytosines in CpG dinucleotides.⁷⁸ CpG dinucleotides are rare throughout the genome, accounting for only 1% of cytosines, but occur at a high frequency in the majority (approximately 70%) of gene promoters.⁷⁹ These regions of CpG clusters are termed CpG islands. Most cytosines in CpG islands are unmethylated, with hypermethylation of CpG islands generally thought to result in transcriptional silencing due to decreased transcription factor binding and increased methyl-CpG binding protein binding which may initiate histone modifications that promote chromatin condensation.78

Four DNMTs with DNA methyltransferase activity have been identified, DNMT1, DNMT3a, DNMT3b and DNMT3c, while DNMT3L lacks methyltransferase activity and instead acts as a cofactor to *de novo* methyltransferases and DNMT2 methylates tRNA.⁸⁰ DNMT3a and DNMT3b are the *de novo* methyltransferases; they establish methylation in previously unmethylated cytosines, whereas DNMT1 maintains a state of methylation, for example in daughter cells following replication.⁸¹ The newly described DNMT3c has been reported to silence retrotransposons in the germ cells of male mice.⁸²

2.4.2. Histone modifications

DNA is wrapped around histone cores composed of two each of histone proteins H2A, H2B, H3 and H4; this structure is termed a nucleosome and comprises the fundamental unit of chromatin. DNA can be loosely packed and amenable to transcription as in euchromatin, or highly condensed and silenced as in heterochromatin. The state of chromatin condensation is determined by post-translational modifications to histone amino-terminal tails, most notably acetylation, phosphorylation and methylation, with the most common modifications summarised in Table 2.3.

Acetylation of histone lysine residues, particularly of H3 and H4, neutralises their positive charge and subsequently promotes an open chromatin structure.⁸³ Histone hyperacetylation is thus considered a signature of active transcription. Histone acetylation is highly dynamic and regulated by two opposing families of enzymes, HATs and HDACs. HATs add acetyl groups to lysine residues using acetyl-CoA as a cofactor. Of the two classes of HATs, Type-A HATs, including the CBP/p300 family, modify multiple sites in histone tails and act as transcriptional co-activators, whereas Type-B HATs, including HAT1, are cytoplasmic and acetylate newly formed histone proteins but not those already complexed to DNA.⁸³ HDACs are divided into four classes; class I HDACs, including HDAC 1, 2, 3 and 8, are nucleic whereas class II HDACs, including HDAC 4, 5, 6, 7, 9 and 10, move between the nucleus and cytoplasm.²⁴ Class III HDACs, also known as sirtuins, include SIRTs 1 to 7 and are associated with longevity and decreased disorders of aging, including T2D.⁸⁴ Unlike class I, II and IV HDACs, which are zinc dependent, sirtuins rely on NAD+ for their deacetylating activity. HDAC11 is the only class IV HDAC, of which little is known.

Similar to acetylation, histone phosphorylation neutralises the positive charge of the histone.⁸³ Serine, threonine and tyrosine residues serve as the phosphorylation and dephosphorylation sites for protein kinases and phosphatases respectively, in a process considered important for transcription, and therefore gene expression, as well as DNA repair, mitosis and apoptosis.⁸⁵

Unlike acetylation and phosphorylation, methylation of histone lysine or arginine residues does not alter the net charge of the histone and can have varying influences on transcription depending on the specific residue methylated (denoted by the histone protein and the lysine [K] or arginine [R] that is methylated) and the degree of methylation (mono-, di- or trimethylation). Histone methylation involves the transfer of methyl groups from SAM to histone residues, catalysed by histone methyltransferase (HMT) enzymes. Methylation of histone lysines H3K4, H3K36, and H3K79 is generally associated with active transcription, whereas methylation of H3K9, H3K27, and H4K20 is more commonly found on transcriptionally silent genes.⁸⁶ Most HMTs are specific to a certain histone residue, for example enhancer of zeste homolog 2 (EZH2) trimethylates H3K27.⁸⁷ Histone methylation is not as dynamic as acetylation and was thought to be a stable event until the discovery of the first histone demethylase was reported in 2004.⁸⁸

Numerous other post-translational modifications to histones have been discovered, including ubiquitination, sumoylation, O-GlcNAcylation, and ADP-ribosylation, although it is less clear how these modifications impact gene expression and chromatin configuration.

Modification	Major histone	Associated effects	
	residues		
Acetylation	H3K9, H3K27,	Activation of gene expression	
	H4K5, H4K8		
Phosphorylation	H3S10, H3S28	DNA replication, chromatin condensation	
		during mitosis and apoptosis	
Methylation	H3K4, H3K36,	Activation of gene expression	
	H3K79		
Methylation	H3K9, H3K27,	Repression of gene expression	
	H4K20		

Table 2.3Common histone modifications

2.4.3. DNA methylation in cholesterol metabolism and T2D

There is an increasing body of evidence that epigenetic modifications are associated with dyslipidaemia and T2D. The majority of evidence in humans comes from epigenome wide DNA methylation association studies (EWAS) looking for associations between individual CpGs from peripheral blood cells and lipid levels or T2D. Multiple studies have identified that lower levels of HDL are associated with increased methylation of ABCG1, in particular the CpG cg06500161,⁸⁹⁻⁹² which has also been associated with increased TG levels and decreased expression of ABCG1.92 Notably, the ABCG1 CpG cg06500161 is also commonly reported to be hypermethylated in peripheral blood cells of people with T2D,93-98 and methylation of ABCG1 has been found to be a predictor of new onset T2D.94,99 Another differentially methylated gene commonly reported in EWAS of T2D is thioredoxininteracting protein (TXNIP), which is hypomethylated in the blood cells of people with diabetes.93,96,98,100,101 TXNIP is associated with oxidative stress as it inhibits the antioxidant protein thioredoxin and has been found to induce β cell apoptosis.¹⁰² Methylation of TXNIP was also found to be inversely correlated with triglyceride levels, but not LDL, HDL or total cholesterol levels.⁹¹

One study found increased methylation of *SREBF2*, the gene encoding SREBP-2, was associated with total cholesterol levels,⁹¹ and *ABCA1* methylation was reported to inversely correlate with HDL levels in a study of familial hypercholesterolemia patients.¹⁰³ In a study of coronary heart disease patients, methylation of the glucokinase gene was found to correlate with LDL levels.¹⁰⁴ Increased glucokinase methylation has also been reported in the blood cells of men with diabetes.¹⁰⁵ The correlation of glucokinase methylation with LDL and of *ABCG1* methylation with diabetes thus highlights the relatedness of cholesterol and diabetes and shows that this relatedness is reflected in epigenetic modifications. Notable differentially methylated regions in the pancreatic islets of diabetics include the insulin gene, insulin receptor substrate 1 (*IRS1*), peroxisome proliferator-activated receptor gamma (*PPARG*) and PPAR γ coactivator 1-alpha (*PPARGC1A*), the latter two of which are important regulatory factors in glucose and lipid metabolism.¹⁰⁶

2.4.4. Histone acetylation in cholesterol metabolism and T2D

There is also evidence that histone acetylation impacts cholesterol metabolism and diabetes risk. HDAC3 has been found to negatively regulate cholesterol synthesis in vitro and in knockout mice,16 while HDAC3 specific inhibition or knockout reportedly protected pancreatic β cells from cytokine-induced apoptosis and increased insulin secretion.¹⁰⁷⁻¹⁰⁹ The HDAC inhibitor trichostatin A (TSA) downregulated a number of genes involved in cholesterol metabolism, including SREBF2 and its target genes HMGCR and LDLR, in the hepatic HepG2 cell line.²⁵ TSA also downregulated SREBP-2 target genes in neuronal cells, while increasing ABCA1 mRNA, with a resultant decrease in cellular cholesterol levels.¹¹⁰ On the other hand, the HAT proteins p300 and CBP act as coactivators in SREBP-2 mediated activation of cholesterol genes and increase histone H3 acetylation in the promoters of LDLR and HMGCR in sterol depleted cells.¹¹¹ Furthermore, SREBPs themselves can be acetylated by p300 and CBP, increasing their stability,¹¹² and conversely can be deacetylated by the HDAC SIRT1, promoting their degradation.¹¹³ p300 has also been shown to regulate gluconeogenesis by increasing *FOXO1* expression,¹¹⁴ and FOXO1 subsequently recruits p300 and CBP to gluconeogenic genes PEPCK and G6Pase, promoting their expression.¹¹⁵ Somewhat counterintuitively, HDAC inhibition has been found to decrease expression of PEPCK and G6PC.¹¹⁶ In animal studies, HDAC inhibition with TSA¹¹⁷ or sodium butyrate¹¹⁸ has been reported to reduce serum

cholesterol and glucose in mice fed a HFD, however evidence is lacking in human trials.

2.5. Statins

HMGCR inhibitors, a class of cholesterol lowering drugs commonly known as statins, are the most commonly prescribed drugs in numerous advanced economies, with approximately one quarter of Americans over the age of 40 being prescribed statins.¹¹⁹ The efficacy of statins in preventing morbidity and mortality from CVD is well established; a meta-analysis of 62 placebo-controlled trials reported reductions in major coronary events by 31% and overall mortality by 13% with statin treatment.¹²⁰

2.5.1. Statins and cholesterol

Statins are competitive inhibitors of HMGCR, the rate-limiting enzyme in the mevalonate pathway of cholesterol biosynthesis. By reducing the rate of cholesterol biosynthesis, particularly in hepatocytes, statins lower cellular cholesterol and thus promote the activation of SREBP-2, as discussed in section 2.2.1. SREBP-2 activates expression of LDLR which results in the uptake of plasma LDL.¹²¹ The continued inhibition of HMGCR by statins ensures that cholesterol biosynthesis does not increase despite the increased expression of HMGCR and other cholesterol synthesis genes as a result of SREBP-2 activation. Numerous clinical trials have demonstrated significant reductions of LDL-C with statin treatment, with atorvastatin and rosuvastatin possessing the most potent cholesterol lowering ability and pravastatin, which is associated with a more tolerable safety profile,¹²² lowering cholesterol to a lesser degree.¹²³ Meta-analysis has shown that the degree of cholesterol lowering by statins is directly correlated with the reduced risk of major cardiovascular events.¹²⁴ Statins also have been found to increase levels of HDL-C, albeit to a lesser extent. A meta-analysis by Barter et al.¹²⁵ reported that different statins increase HDL-C to different extents independently of LDL-C reductions, with rosuvastatin and simvastatin increasing HDL-C to a greater degree than atorvastatin.

2.5.2. Statins and diabetes

Statins have been associated with small but statistically significant increased risk of T2D; a meta-analysis of 13 clinical trials reported a 9% increase in risk of T2D with statin use, although results were variable and some studies reported a possible

protective effect, especially for pravastatin.¹⁴ In vitro, statins have been found to alter glucose uptake, utilisation and generation. In β cells, studies both in our group⁷² and elsewhere,¹²⁶⁻¹²⁹ have shown that statins impair GSIS. Additionally, atorvastatin and pravastatin reportedly decrease expression of GLUT2 in β cells.¹²⁹ Similarly, in myocytes and adipocytes, statins reportedly decrease expression of GLUT4,¹²⁹⁻¹³¹ the insulin-regulated glucose transporter, as well as other proteins involved in insulin signalling including the insulin receptor¹³² and IRS-1.¹³³ Furthermore, lovastatin treatment was reported to reduce glucose uptake by adipocytes, skeletal myocytes and hepatocytes,¹³⁴ and atorvastatin decreased insulin-stimulated glucose uptake in adipocytes.¹³⁵ Finally, in hepatocytes, statins have been shown to decrease expression of glucokinase,¹³⁶ which phosphorylates glucose in the first step of glucose utilisation for both glycogen synthesis and glycolysis, and to increase expression of the gluconeogenic proteins G6Pase and PEPCK.137 All these mechanisms may contribute to hyperglycaemia. Much like in clinical trials, variable results have been reported; Chen et al.¹³⁸ reported that atorvastatin treatment protected β cells from cholesterolinduced apoptosis in vitro and animal studies have reported improved insulin sensitivity with rosuvastatin^{139,140} and lovastatin.¹⁴¹

Despite this research, the exacting mechanism by which statins may increase diabetes risk is not known, nor is why the risk seems to vary between difference statins. The possible protective effect of pravastatin has been attributed to its hydrophilic nature, whereas atorvastatin, simvastatin, fluvastatin and lovastatin are lipophilic and thus can enter numerous cell types readily, whereas hydrophilic statin are more hepatoselective.¹⁴² However, rosuvastatin is also hydrophilic but is associated with a significant increase in diabetes risk, and meta-analysis has shown there is no difference in risk between hydrophilic and lipophilic statins.¹⁴ The difference in risk may instead be directly related to the potency of LDL-C lowering, however, another meta-analysis by Navarese et al.¹⁴³ found that although diabetes risk was increased with higher statin doses, this was not influenced by the degree of LDL-C reduction. Numerous potential exacting mechanisms have been proposed, as reviewed by Paseban et al.¹⁴⁴ These include alterations in membrane lipids impacting Ca²⁺ channels in β cells and GLUT4 translocation in peripheral cells, and reduced production of other products of the mevalonate pathway, such as dolichol, which is involved in membrane receptor processing, and coenzyme Q10, which is involved in mitochondrial ATP generation.

2.5.3. Statins and epigenetics

There are several studies on the influences of statins on epigenetic modifications in the context of different pathologies, most notably in cancer. Statins may affect gene expression by modifying histone acetylation, reportedly by inhibiting HDACs. Studies of various cancer cell lines,^{145,146} a macrophage line,¹⁴⁷ an epithelial line,¹⁴⁸ and a rabbit model of atherosclerosis¹⁴⁹ reported that statin treatment resulted in hyperacetylation of histones H3 and H4 and/or inhibition of HDAC activity or expression. Statin use has also been reported to increase the recruitment of the HAT p300 to specific loci.¹⁴⁷ Cooney¹⁵⁰ proposed that with less acetyl-CoA used in cholesterol biosynthesis, increased amounts are available to act as acetate donors to acetyltransferases, however this mechanism has not been investigated in the reported studies. Potential mechanisms by which statins may alter histone acetylation are summarised in Figure 2-4.



Figure 2-4 Postulated mechanisms by which statins may impact histone acetylation

Without statin treatment (top) acetyl-CoA is used in the mevalonate pathway in the synthesis of cholesterol and other biomolecules. Statins, HMG-CoA reductase inhibitors, block the reduction of HMG-CoA into mevalonate, potentially providing additional acetyl-CoA for histone acetyltransferases (HAT) to add acetyl groups (Ac) to histone tails (bottom). Statins may also increase the recruitment of HATs to target loci. Histone deacetylases remove acetyl groups from histone tails (top); statins may prevent this through HDAC inhibition (bottom). Originally published in Allen and Mamotte.²⁸

Statins may also impact histone methylation. Ishikawa et al.¹⁵¹ reported that statin treatment resulted in downregulation of mRNA and protein expression of EZH2, a HMT that methylates H3K27 and thus suppresses gene expression, in colorectal cancer cells, although this study did not investigate subsequent epigenetic modifications.

Statin treatment may increase expression of certain genes through reduced DNA methylation. Studies of statin treated cancer cell lines by Karlic et al.¹⁴⁶ and Kodach et al.¹⁵² reported downregulated DNMT mRNA expression and protein activity, respectively. Statin treatment was also reported to reduce the methylation of promoters of certain genes and subsequently increase gene expression in cancer cells¹⁵² and T cells.¹⁵³

Studies on the epigenetic effects of statins in the context of cholesterol lowering and T2D are limited, despite being the primary goal of statin treatment and a significant adverse effect respectively. Recently, several studies have been published showing effects on DNA methylation in human statin users. In an EWAS of 8313 participants from five cohorts, differentially methylated CpGs in DHCR24, SC4MOL, and ABCG1 were found in the blood cells of statin users, with increased methylation at one CpG, cg06500161, in *ABCG1* independently predicting incident T2D.¹⁵⁴ Methylation at this site was also associated with reduced ABCG1 gene expression and with increased fasting glucose and insulin. Similarly, in a separate cohort of 2741 participants, and validated in an additional cohort of 2020 participants, statin use was associated with increased methylation of ABCG1.155 This methylation was also associated with T2D, fasting glucose and insulin. In a published abstract of a cohort of 896 statin users and 943 non-users by Yao et al.,¹⁵⁶ statin use was associated with differentially methylated sites in DHCR24, SC4MOL, ABCG1 and NACA. Importantly, the CpG at which statins increase methylation of ABCG1, cg06500161, is the same site associated with T2D, discussed in section 2.4.3. Both DHCR24 and SC4MOL are involved in cholesterol synthesis, while NACA binds to newly synthesised peptides that lack a signal peptide motif and does not appear to have been previously linked to cholesterol or diabetes. ClinicalTrials.gov shows a clinical trial (NCT02817230) investigating the effect of statin treatment on H3K4 trimethylation in the gene promoter regions of proinflammatory cytokines in hypercholesterolaemic patients, although no results or updates have been posted.

2.6. Butyrate

Butyrate is a SCFA produced by the bacterial fermentation of dietary fibre, in particular resistant starch, and is also present in some foods such as butter.¹⁵⁷ Butyrate is a known HDAC inhibitor with a reported IC50 ranging from 90 μ M in HT-29 nuclear extracts to 1.13 mM in live HeLa cells.¹⁵⁸ Many of butyrate's reported effects

have been attributed to HDAC inhibition and subsequent increased histone acetylation and gene expression. Conversely, it has also been reported that butyrate may decrease acetylation and subsequent gene expression in some sites, particularly with prolonged exposure.¹⁵⁹ By altering histone acetylation, butyrate alters the expression of a significant number of genes; microarray analysis of nearly 20 000 genes in butyrate treated colonic epithelial cells found more than 10% of genes had a two-fold or greater change in expression compared to untreated controls, with 4% upregulated and 6% downregulated.¹⁶⁰ Butyrate may also influence metabolism through the binding and activation of free fatty acid G-coupled receptors, FFAR2 and FFAR3, particularly in the intestine. Binding of SCFAs to these receptors in intestinal enteroendocrine cells is thought to promote the release of GLP-1 and peptide YY (PYY),¹⁶¹ the former promoting pancreatic insulin secretion and inhibiting glucagon secretion, and the latter reducing appetite and slowing gastric emptying.¹⁶²

Butyrate's mechanisms of action are summarised in Figure 2-5.



Figure 2-5 Butyrate's mechanisms of action

Butyrate activates free fatty acid receptors (FFAR), which in enteroendocrine cells may lead to the release of gut hormones such as GLP-1 and PYY. Butyrate also inhibits HDAC activity, increasing histone a cetylation (Ac) and altering gene expression. Originally published in Bridgeman et al.¹⁶³

In animal models of metabolic diseases, butyrate supplementation reportedly exerts numerous benefits, including lowering of serum triglyceride, cholesterol and glucose levels, reducing weight gain, improving glucose tolerance and protecting against cardiac dysfunction.^{164,165} However, the few small-scale human trials conducted have lacked significant results; a randomised placebo-controlled clinical trial investigated the effect of six daily 100 mg sodium butyrate tablets for 45 days in type 2 diabetics.^{6,7}

The study found butyrate did not significantly alter fasting blood glucose, fasting insulin, cholesterol or triglyceride levels, although it did increase GLP-1 levels and reduce diastolic blood pressure.⁶ The lack of efficacy for diabetes in this study could be due to the small population (15 people per group). In a pilot study, four weeks of 4 g daily oral sodium butyrate improved peripheral and hepatic insulin sensitivity in lean males but not in males with metabolic syndrome (MetS).¹⁶⁶ This trial was not placebo-controlled and used small study groups (nine lean, ten MetS). Well-designed, large-scale clinical trials are needed to determine if there is any effect of butyrate on cholesterol and diabetes in humans.

2.6.1. Butyrate and cholesterol

Multiple studies have reported that butyrate lowers serum cholesterol in animals fed a HFD.118,165,167,168 Reduced serum cholesterol could be due to alterations in cholesterol biosynthesis, cholesterol import/export from cells or dietary cholesterol uptake. In vitro studies suggest that butyrate may influence all three processes. Marcil et al.¹⁶⁹ reported that butyrate reduced cholesterol synthesis in Caco-2 enterocytes, associated with a reduction in HMGCR activity. Reductions in HMGCR mRNA expression have also been reported in Caco-2 cells, in addition to reduced HMG-CoA synthetase 2 expression.¹⁷⁰ Butyrate also decreased expression of genes in downstream steps of cholesterol synthesis, namely in the conversion of mevalonate to squalene (isopentenyl diphosphate isomerase, dimethylallyl/geranyl transtransferase and farnesyl-diphosphatase franesyltransferase) and in the conversion of squalene to cholesterol (lanosterol 5-desaturase and methylsterol monooxygenase). These studies did not investigate if the altered expression was due to HDAC inhibition. Conversely, in hamsters on a high cholesterol diet, although butyrate decreased plasma total cholesterol, it increased hepatic HMGCR gene expression, in addition to SREBF2 and LDLR, possibly as a compensatory measure in response to lowered cholesterol as is seen with statin treatment.¹⁶⁸ In this study, similar effects were found with acetate and propionate, suggesting this is an effect common to SCFAs and not a result of HDAC inhibition.

There is conflicting evidence for how butyrate influences cholesterol transport. Kaptein et al.¹⁷¹ found increased secretion of cholesterol esters from HepG2 cells with 2 mM sodium butyrate. Conversely, Marcil et al.¹⁶⁹ found reduced secretion of cholesterol esters with 20 mM sodium butyrate in Caco-2 cells. Differences between hepatic cells and intestinal cells and differences in butyrate concentration may contribute to these disparate effects. Marcil et al.¹⁷² also reported reduced synthesis of apolipoproteins apoB-48 and apoA-I in butyrate-treated cells, whereas other studies found butyrate increased apolipoprotein expression and secretion. Nazih et al.¹⁷³ found butyrate increased apoA-IV mRNA expression and protein secretion in Caco-2 cells, and increases in apoA-1 and apoB-100 secretion have been reported in butyrate treated HepG2 cells.¹⁷⁴ Other proteins involved in cholesterol transport that may be upregulated by butyrate in HepG2 cells include phospholipid transfer protein, involved in the transfer of phospholipids and cholesterol amongst lipoproteins and lipids,¹⁷⁵ and cholesteryl-ester-transfer protein, which transfers lipids between VLDL or LDL and HDL.¹⁷⁶ Lecithin:cholesterol acyltransferase, which promotes the transfer of cholesterol from cell membranes to HDL, may be downregulated.¹⁷⁷ It was not investigated if changes in cholesterol transport were due to HDAC inhibition.

Finally, butyrate may reduce dietary cholesterol uptake by intestinal cells. In a study of Caco-2 cells, butyrate significantly reduced cholesterol uptake and reduced expression of *NPC1L1*.¹⁷⁸ The gene expression of *ABCG5* and *ABCG8*, involved in transporting cholesterol from intestinal cells into the intestinal lumen and thus away from the circulation, was increased. These effects were not seen in cells treated with acetate or propionate, indicating this is not a common effect of SCFAs.

2.6.2. Butyrate and diabetes

The ability of butyrate to induce insulin secretion in farmed animals has been known for decades, long before its ability to inhibit HDACs was discovered. In 1967, Manns et al.¹⁷⁹ reported that infusion of butyrate directly into a pancreatic artery of sheep resulted in a marked increase in plasma insulin within 3 min. Investigations into the effect of butyrate on insulin secretion *in vitro* have reported variable results, possibly as a result of the use of cell lines with defective GSIS. In pancreatic islets from mice fed an HFD with or without butyrate supplementation, butyrate increased GSIS by reducing basal (2.8 mM glucose) and increasing glucose-stimulated (16.7 mM) insulin secretion.¹⁸⁰ Conversely, in a study of BRIN-BD11 cells, three days of butyrate treatment reduced cell growth, insulin content and insulin secretion in both the presence and absence of glucose.¹⁸¹ In RIN-m5F-2A cells, which are usually unresponsive to glucose, sodium butyrate increased insulin secretion, intracellular content and mRNA expression approximately three-fold in the presence and absence

of glucose and other secretagogues (glyceraldehyde and potassium), although GSIS remained impaired.¹⁸² This was associated with increased activity of both hexokinase and glucokinase, indicating increased responsiveness to both low and high glucose concentrations, and increased GLUT2 expression. A similar study in RINm5F cells also reported increased insulin secretion with impaired GSIS, associated with increased hexokinase, but not glucokinase, gene expression and protein activity.¹⁸³

Butyrate has been reported to improve insulin signalling, with multiple studies finding that butyrate increases phosphorylation of IRS-1 and/or AKT in animal models of diabetes.^{118,165,167,184} Chriett et al.¹⁸⁵ showed that butyrate increased histone acetylation in proximity to the IRS-1 promoter, increasing its expression, in L6 myocytes exposed to palmitate in a model of insulin resistance, suggesting a direct mechanism by which butyrate can increase insulin sensitivity through HDAC inhibition. This increase in insulin signalling should lead to increased glucose uptake, and in porcine adipocytes an increase in glucose uptake has been reported with sodium butyrate treatment.¹⁸⁶ This was associated with an increase in expression of the glucose transporter GLUT4, which has also been reported in rat adipose tissue.¹¹⁸ Similarly, butyrate has also been reported to increase GLUT2 expression in liver cells.¹¹⁸

Increased insulin signalling in the liver may also ameliorate hyperglycaemia through inhibition of gluconeogenesis. There is also evidence that butyrate inhibits gluconeogenesis through additional mechanisms. It has been shown that HDACs can deacetylate the gluconeogenic transcription factor FOXO1, leading to an alteration of its subcellular localisation and DNA binding.¹⁸⁷ HDAC inhibition may in this way decrease the expression of gluconeogenic enzymes such as PEPCK and G6Pase. In the livers of diabetic rats, butyrate treatment was shown to reduce nuclear FOXO1.¹⁶⁷ Furthermore, the same study found that sodium butyrate reduced the expression of glucagon, a hormone that promotes gluconeogenesis and glycogenolysis, in rat islets.

2.7. Conclusion

There is a growing body of evidence that epigenetic modifications are associated with the development and pathology of dyslipidaemia, insulin resistance and T2D. Furthermore, agents with reported epigenetic effects, namely butyrate and statins, have been shown to lower cholesterol levels and impact insulin sensitivity and the risk of developing T2D. However, there are significant gaps in the literature concerning the underlying biological mechanisms. Few studies have linked specific epigenetic modifications to the observed phenotypic changes. In the case of statins, studies proposing epigenetic effects have been conducted primarily in the context of cancer and not in the context of cholesterol lowering or diabetes. For butyrate, *in vitro* studies have reported inconsistent results, while for *in vivo* animal studies, it is not known if the observed effects are a result of direct epigenetic changes in the involved tissues or indirect as a result of FFAR activation in intestinal cells and the subsequent release of hormones such as GLP-1. Further research is therefore needed, examining epigenetic changes in the vicinity of key genes involved in cholesterol and glucose metabolism in response to these agents.

Chapter 3. Methods and Materials

3.1. Creation of stock solutions

All statins were sourced from Cayman Chemical. Atorvastatin (calcium salt), rosuvastatin (calcium salt), simvastatin (sodium salt) and pravastatin (sodium salt), Trichostatin A (TSA) (Cayman Chemical) and curcumin and RG-108 (abcam) were dissolved in dimethyl sulfoxide (DMSO) (Sigma-Aldrich) to a concentration of 10 mM and stored at -20°C or at -80°C for long-term use. Sodium butyrate (Selleck Chemicals) and sodium acetate (Toronto Research Chemicals) were dissolved in ultra-pure distilled water to a concentration of 50 mM and stored at -20°C.

3.1.1. Creation of lipoprotein-deficient serum and highlipoprotein serum

The density of foetal bovine serum (Serana) was adjusted to 1.21 g/ml with sodium bromide. The solution was transferred to 11.5 ml polyallomer ultracrimp tubes and centrifuged for 20 h at 70000 g in a Sorvall WX ultracentrifuge. The base of the tube was pierced with a butterfly needle and 1 ml aliquots collected. Cholesterol and total protein concentrations of the aliquots were measured using the AmplexTM Red Cholesterol Assay Kit (Life Technologies) and the Pierce BCA Protein Assay Kit (ThermoFisher) respectively, to determine which aliquots were deficient in lipoproteins (lipoprotein-deficient serum; LPDS) and those enriched in lipoproteins (high-lipoprotein serum; HLPS). Salt was removed from the serum by centrifugation with ZebaTM desalting spin columns 7K MWCO (ThermoFisher) and serum was sterilised with a Millex 0.22 µm syringe filter prior to storage at -20°C.

3.2. Sample preparation

3.2.1. Cell culture and treatments

HepG2 human hepatocellular carcinoma cells were obtained from ATCC via Dr Ross Graham (Curtin University). MDA-MB-231 human breast carcinoma cells were obtained from ATCC via Professor Arunasalam Dharmarajan (Curtin University). BRIN-BD11 rat insulinoma cells were obtained from Professor Peter Flatt via Professor Philip Newsholme (Curtin University). THP-1 human leukemic monocytes were obtained from Dr Hilary Warren (Canberra Hospital) via Professor Deirdre Coombe (Curtin University).

HepG2 human hepatocellular carcinoma cells were maintained in Dulbecco's Modified Eagle's medium: Ham's F-12 Nutrient Mixture (DMEM:F12 media) (Sigma-Aldrich) supplemented with 10% foetal bovine serum (FBS). BRIN-BD11 rat insulin-secreting cells, MDA-MB-231 human breast cancer cells and THP-1 human monocytes were maintained in Roswell Park Memorial Institute (RPMI) media (Sigma-Aldrich) supplemented with 10% FBS. All cells were maintained in 25-cm² or 75-cm² tissue culture flasks at 37°C in a humidified incubator equilibrated with 5% CO₂. All cells tested negative for mycoplasma contamination. Cells were seeded in 6-well plates, 24-well plates, 96-well plates, 25-cm² vented tissue culture flasks or 75-cm² vented tissue culture flasks and allowed to recover overnight prior to treatment. Prior to treatment, THP-1 cells were differentiated into macrophages by the addition of 50 nM phorbol 12-myristate 13-acetate (PMA) (Sigma-Aldrich) to the culture media. THP-1 cells were differentiated for 48 h, then allowed to recover for 24 h before treatment. All cells were treated in media supplemented with 10% LPDS unless otherwise indicated. Control cells were treated with media only or media containing DMSO as a vehicle control as indicated.

3.2.2. Animal Study

Liver samples were utilised from a previous animal study conducted by our group. In that study, eight-week-old male C57B1/6J mice from the Animal Resource Centre, Murdoch, Western Australia were delivered to the animal facility, Curtin University. Following acclimatisation, mice were randomly assigned to either normal diet (ND) (14% of energy from lipids, Meat Free Rat and Mouse Maintenance Diet, Specialty Feeds, Glen Forrest, WA, Australia) or high-fat diet (HFD) (59% of energy from lipids, 36% Fat Modification of AIN93G, Specialty Feeds). From week 4, mice were further divided into treatment groups to receive either 10 mg/kg/day of atorvastatin or vehicle (water) by gastric gavage for a further 12 weeks. Following completion of the treatment period, mice were starved for 6 h and anesthetised in an isoflurane chamber before being euthanised by cervical dislocation. Liver samples were removed from the carcass and collected in prechilled microcentrifuge tubes and snap-frozen either in dry ice or liquid nitrogen. Animal experiments were approved by

Curtin University's Animal Ethics Committee (AEC_2016_17, approval date 14/4/2016).

3.2.3. Nuclear extraction

Nuclear extraction was conducted using the abcam Nuclear Extraction Kit (ab113474) according to the manufacturer's instructions. For experiments on cultured cells, confluent cells were scraped and incubated on ice for 10 min in pre-extraction buffer containing dithiothreitol (DTT) and protease inhibitor cocktail (PIC), then centrifuged at 12 000 rpm for 1 min. The cytoplasm containing supernatant was removed, and the pellets incubated on ice for 15 min in nuclear extraction buffer containing DTT and PIC, then sonicated three times for 10 s in a Biorupter sonicating water bath. The extract was then centrifuged at 14 000 rpm for 10 min at 4°C and the supernatant removed for protein quantification using the Pierce Coomassie Protein Assay Kit (ThermoFisher). For experiments on murine liver from the animal study, samples were placed in a glass homogeniser containing pre-extraction buffer with DTT and PIC and homogenised manually. Homogenised samples were transferred to microcentrifuge tubes and incubated on ice for 15 min then centrifuged at 10000 rpm for 10 min at 4°C. The supernatant was removed, and nuclear extraction continued as for cultured cells.

3.2.4. Whole cell protein extraction

Confluent cells in were scraped then lysed on ice for 20 min radioimmunoprecipitation assay (RIPA) lysis buffer (Sigma-Aldrich) containing protease and phosphatase inhibitors (Sigma-Aldrich). The lysate was sonicated 10 times for 10 s then centrifuged at 14000 rpm for 10 min and the protein rich supernatant was removed for protein quantification by the bicinchoninic acid (BCA) method using the Pierce BCA Protein Assay Kit (ThermoFisher).

3.3. Viability assays

Viability was determined using alamarBlue[®] (ThermoFisher) as a measure of cellular metabolism. Upon entering living cells, resazurin is reduced to resorufin, a compound that is highly fluorescent. Following 22 h treatment in 96 well plates, 10 μ L of alamarBlue[®] was added to the treatment media and cells were incubated for a further 2 h. Fluorescence was then measured using an EnSpire Multimode Plate Reader

(PerkinElmer) with an excitation wavelength of 540 nm and an emission wavelength of 590 nm.

3.4. Lipid quantification

3.4.1. Lipid extraction

After 24 h treatment in 96 well plates, media was removed, and cells were washed twice with PBS. Lipids were extracted using 150 μ L of a solution containing 3:2 hexane:isopropanol. Following 15 s incubation and mixing by pipetting, lipids were transferred to a 96 V well plate and allowed to dry in a fume hood at room temperature (RT) for at least 4 h or overnight.

3.4.2. Cellular cholesterol content

Cholesterol content was assessed using the Amplex Red Cholesterol Assay Kit (Life Technologies). Samples were dissolved in a 1:1 reaction buffer:isopropanol mixture and incubated with catalase for 15 min at 37°C to eliminate endogenous peroxidases which would otherwise interfere in this assay.¹⁸⁸ Amplex red working solution was added and following 30 min incubation at 37°C, fluorescence was read with an excitation wavelength of 540 nm and an emission wavelength of 590 nm. Cholesterol content of samples was determined using a standard curve based on a serial dilution of the cholesterol standard provided with the kit.

3.4.3. Cellular triglyceride content

TG content was assessed using the High Sensitivity Triglyceride Fluorometric Assay Kit (Sigma-Aldrich) as per the manufacturer's instructions. Samples were dissolved in 1:1 reaction buffer:isopropanol and incubated with lipase for 20 min at 37°C to hydrolyse TGs into glycerol and fatty acids. Samples were then incubated with the reaction master-mix containing the developer, probe and enzyme mix that reacts with the glycerol present. Following 30 min incubation at 37°C, fluorescence was measured with an excitation wavelength of 535 nm and an emission wavelength of 587 nm. TG content of samples was determined using a standard curve based on a serial dilution of the TG standard provided with the kit.

3.4.4. Cholesterol uptake and export assays

Cholesterol uptake and export was determined using the Cholesterol Uptake Cell-Based Assay Kit (Cayman Chemical). For cholesterol uptake assays, cells in 96-well plates were incubated in media containing 10% LPDS for 24 h prior to treatment containing 20 ug/ml of a fluorescently-tagged cholesterol (NBD cholesterol). Following 24 h treatment, cells were washed in PBS, assay buffer was added and fluorescence was measured with an excitation wavelength of 485 nm and an emission wavelength of 535 nm. The cholesterol transport inhibitor U-18666A was used as a positive control. For cholesterol export assays, cells in 96-well plates were incubated with media containing 10% LPDS and 20 ug/ml NBD cholesterol for 48 h prior to treatment in media containing 10% FBS. Following 24 h treatment, the media containing exported NBD cholesterol was transferred to fresh wells and fluorescence measured as for uptake assays.

3.5. Immunoblotting

Equal amounts of protein (20-30 µg depending on sample concentration and amount of protein of interest in sample) were denatured at 98°C for 10 min in solution containing Bolt Sample Reducing Agent (ThermoFisher) and SDS Sample Loading Buffer (Sigma Aldrich), then fractionated on BoltTM 4-12% Bis-Tris Plus Gels (ThermoFisher). Proteins were transferred onto nitrocellulose membranes using the iBlot Gel Transfer Device (Invitrogen) and blocked with 3% bovine serum albumin (BSA) (Bovogen) in Tris-buffered saline with 0.1% Tween 20 (TBST) for at least 1 h. Membranes were incubated overnight with primary antibodies (Table 3.1) diluted in blocking buffer. Membranes were incubated with the secondary antibodies conjugated to horseradish peroxidase (HRP) (Table 3.1) in blocking buffer for at least 1 h then washed three times with TSBT. Immunodetection was performed using the Amersham ECL Prime Western Blotting Detection Reagent (GE Healthcare) and the Chemi-Doc[™] Gel Imaging System (Bio-Rad). Band density was measured with Image Lab software (Bio-Rad) and normalised to the density of the housekeeping protein GAPDH and total protein as determined by Coomassie Blue staining in separate analyses.

Target	Туре	Product #	Company
GAPDH	Mouse	ab8245	abcam
	monoclonal		
H3K9ac	Mouse	sc56616	Santa Cruz
	monoclonal		Biotechnology
H4ac (S1, K5, K8	Mouse	sc377520	Santa Cruz
and K12)	monoclonal		Biotechnology
H3K9me3	Mouse	sc130356	Santa Cruz
	monoclonal		Biotechnology
HDAC3	Mouse	sc376957	Santa Cruz
	monoclonal		Biotechnology
ABCA1	Mouse	ab66217	abcam
	monoclonal		
ABCG1	Rabbit	ab52617	abcam
	monoclonal		
Caveolin-1	Rabbit polyclonal	ab18199	abcam
HMGCR	Rabbit	ab174830	abcam
	monoclonal		
LDLR	Rabbit polyclonal	ab30532	abcam
SRB1	Rabbit	ab52629	abcam
	monoclonal		
SREBP-2	Rabbit polyclonal	ab28482	abcam
APO-A1	Mouse	sc376818	Santa Cruz
	monoclonal		Biotechnology
АРО-В	Mouse	sc-13538	Santa Cruz
	monoclonal		Biotechnology
Glucokinase	GlucokinaseRabbit polyclonalab37796abcam		abcam
GLUT2	Rabbit polyclonal	ab54460	abcam
Hexokinase I	Rabbit	2024	Cell Signalling
	monoclonal		
РЕРСК	Mouse	sc166778	Santa Cruz
	monoclonal		Biotechnology

 Table 3.1
 Antibodies used in immunoblotting

Target	Туре	Product #	Company		
АКТ	Rabbit	ab81283	abcam		
(phospho-Ser473)	monoclonal				
ΙRβ	Rabbit polyclonal	ab60946	abcam		
(phospho-Tyr1361)					
IRS-1	Rabbit polyclonal	sc17196	Santa Cruz		
(phospho-Tyr632)			Biotechnology		
Secondary antibodies					
Goat Anti-Mouse	IgG H&L (HRP)	ab6789	abcam		
Goat Anti-Rabbit	IgG H&L (HRP)	ab6721	abcam		

3.6. Gene expression

3.6.1. RNA extraction

RNA extraction was performed using the TRI reagent protocol as described by Rio et al.¹⁸⁹ using 1-bromo-3-chloropropane (BCP) in place of chloroform for phase separation.¹⁹⁰ Treated cells in 6-well plates were washed twice in PBS and 300 μ L TRI reagent (Sigma-Aldrich) was added directly to cells. Following mixing, the solution was transferred to 1.5 ml tubes, BCP was added and tubes were vortexed and incubated at RT for 10 min. Tubes were centrifuged at 14000 rpm for 15 min at 4°C and the upper aqueous phase carefully transferred to fresh low adherence 1.5 ml tubes. An equal volume of isopropanol was then added, tubes were mixed by inversion and incubated at RT for 10 min. The tubes were then centrifuged at 14000 rpm for 15 min at 4°C, the supernatant removed, and the RNA pellet was washed twice with 75% ethanol and allowed to air dry. Pellets were dissolved in nuclease free water and allowed to equilibrate for 1 h at RT. RNA was quantified using the NanoDrop 1000 spectrophotometer (ThermoFisher), with purity considered sufficient if A_{260}/A_{280} measured between 1.8-2.0. Samples were stored at -80°C.

3.6.2. Reverse transcription

Reverse transcription was performed using the SensiFASTTM cDNA Synthesis Kit (Bioline) according to the manufacturer's instructions. 1 μ g of RNA was used per reaction in a 20 μ l reaction mix containing reverse transcriptase and a blend of anchored oligo dT and random hexamer primers. Reverse transcription was

performed in a Veriti 96 Well Thermal Cycler (Applied Biosystems) with the following program: 25°C for 10 min (primer annealing), 42°C for 15 min (reverse transcription), 48°C for 15 min (reverse transcription of complex RNA), 85°C for 5 min (reverse transcriptase inactivation), 4°C hold (cooling). cDNA was diluted in 20 μ l ultrapure water and stored at 4°C.

3.6.3. qPCR

Quantitative real-time PCR (qPCR) was performed using the DNA intercalating fluorescent dye, SYBR-Green.¹⁹¹ Predesigned KiCqStart® SYBR® Green Primers (Sigma-Aldrich, sequences in Table 3.2, Table 3.3) were dissolved in ultrapure water to a concentration of 100 μ M and stored at -80°C. Working primer mixes of 5 μ M forward and reverse primers in ultrapure water were stored at 4°C. SensiFAST SYBR Lo-ROX Mix (Bioline) was used with 200 nM primer mix unless otherwise stated, $2 \mu l cDNA$ sample template and ultrapure water in a 10 μl reaction mix. qPCR was performed in a CFX Connect Real-Time System (Bio-Rad) with the following program: 95°C for 2 min (activation) followed by 40 cycles of 95°C for 15 sec (denaturation), 57°C-62°C for 15 sec (annealing) and 72°C for 1 min (extension). Melt curve analysis was then performed starting at 72°C for 30 sec, then increasing 0.5° C every 5 sec to a maximum of 95°C. The annealing temperature for each primer (Table 3.2, Table 3.3) was determined using a serial dilution of a mixed cDNA sample template and a temperature gradient, with a standard curve of Ct values plotted and efficiency calculated using $E = -1+10^{(-1/slope)}$. If efficiency was poor (<90%), the R² value of the standard curve was poor (<0.99) or multiple peaks were seen by melt curve analysis, additional optimisation was performed using different primer concentrations. Human APOB and PPARG and rat Actb and Rpl13a used 100 nM primers due to the presence of multiple peaks seen on melt curve analysis at 200 nM. Initial gene expression qPCR experiments included serial dilution to determine efficiency of primers, no template controls, no reverse transcriptase controls and three technical replicates of each sample, with two biological replicates of each treatment. Confirmatory experiments used two technical replicates and lacked serial dilutions to reduce costs. Gene expression was normalised to that of reference genes chosen by reviewing the literature for genes shown to be suitable for use in statin or HDAC treated HepG2 cells.¹⁹²⁻¹⁹⁴ Human GAPDH, RPL13A and YWHAZ primers were used for HepG2 cells and rat Actb and Rpl13a primers were used for BRIN-BD11 cells.

Normalised gene expression $(\Delta\Delta C_q)$ was calculated by Bio-rad CFX Manager 3.1 software.

Gene	Forward Primer	Reverse Primer	Annealing
			Temperature
GAPDH	ACAGTTGCCATGTAGACC	TTGAGCACGGGTACTTTA	57-60°
YWHAZ	AACTTGACATTGTGGACATC	AAAACTATTTGTGGGACAGC	57-63°
RPL13A	GTCTGAAGCCTACAAGAAAG	TGTCAATTTTCTTCTCCACG	61-63°
LDLR	GAGGACAAAGTATTTTGGACA	GTAGGTTTTCAGCCAACAAG	57°
	G		
PPARG	AAAGAAGCCAACACTAAACC	TGGTCATTTCGTTAAAGGC	63°
SREBF2	CAGCAGGTCAATCATAAACTG	GGACATTCTGATTAAAGTCCT	61.4°
		С	
NR1H3	CATGACCGACTGATGTTC	CAAACACTTGCTCTGAGTG	61.4°
APOA1	AGGAGTACACTAAGAAGCTC	AAACGTTTATTCTGAGCACC	62°
APOB	CTTACATCCTGAACATCAAGA	AGTTTCCATACACGGTATCC	63°
	G		
GK	GTTCTTCTGAGATCTATGGC	AAATACACACTTATGGCCTG	60°
HMGCR	ACTTCGTGTTCATGACTTTC	GACATAATCATCTTGACCCTC	59°
PEPCK1	ATTCTGGGTATAACCAACCC	GTTGATGGCCCTTAAATGAC	59°
SLC2A2	AGAAGATTAGACTTGGACTCT	GTGACCTTATCTTCTGTCATTG	59°
	С		
SREBF1	AATCTGGGTTTTGTGTCTTC	AAAAGTTGTGTACCTTGTGG	61.4°
CAV1	CAGGGACATCTCTACACC	TCAAAGTCAATCTTGACCAC	61.4°
CYP7A1	AAATCTACCCAGACCCTTTG	TTCCAGGACATATTGTAGCTC	58°
FOX01	GTCAAGACAACGACACATAG	AAACTAAAAGGGAGTTGGTG	59°
INSR	GATCCAATCTCAGTGTCTAAC	CCTTTGAGGCAATAATCCAG	62°
ABCG1	GAGGTGAACCCTTTTCTTTG	CCTTTCTCAACCCCTTTAATC	60°
CEBPA	AGCCTTGTTTGTACTGTATG	AAAATGGTGGTTTAGCAGAG	59°
CPT1A	ACGGGGATTATAAGTCAAGG	CACAGCAAGTGAAAATCAAC	58°
HNF4A	AGTACATCCCAGCTTTCTG	AATGTAGTCATTGCCTAGGAG	61.4°
MVK	CATGTTGTCAGAAGTCCTAC	CTCAAGTTCAAGGATACAGC	59°
ABCA1	GTGTTTCTGGATGAACCC	TTCCATTGACCATGATTGC	62°
HK1	AGGTATGAGAAGATGATCAG	GAGAAACTTGGTCTCAAAGAT	61.4°
	TG	G	
SCARB1	ACAAAAGCAACATCACCTTC	TGGGCTTATTCTCCATCATC	62°

Table 3.2Human primers for qPCR experiments in HepG2 cells

Gene	Forward Primer	Reverse Primer	Annealing
			Temperature
Actb	AAGACCTCTATGCCAACAC	TGATCTTCATGGTGCTAGG	57-61°
Rpl13a	GCACAAGACCAAAAGAGG	CGCTTTTTCTTGTCATAGGG	57-61°
Hk1	AAACTCTGGGAAACAAAGG	AAACTTGGTCTCAAAGATGC	57°
Hnf4a	TGTGTGAGTCTATGAAGGAG	ATGTAGTCATTGCCTAGGAG	61°
Pdx1	AAAGGTTACAAACTTGAGCG	AAACAGCTCCCTTTATTCTC	59°
Srebf1	AAACCTGAAGTGGTAGAAAC	TTATCCTCAAAGGCTGGG	59°
Ins1	AACGTGGTTTCTTCTACAC	TCTCCAGTTGGTAGAGGG	61°
Abcg1	GTTATGTTCTTTGATGAGCCC	CCTTGACTTAGGACATAAAG	57°
		С	
Abca1	CATCTGAAAAACAGGTTTGG	GGGAGAGAATGCTGAATAT	57°
		C	
Txnip	CGTCAATACTCCTGACTTAATG	AAATGTCATCACCTTCACAG	61°
Ldlr	CAGTGAAGATATTGACGAG	TCACTTACGTACCTCATGG	61°

 Table 3.3
 Rat primers for qPCR experiments in BRIN-BD11 cells

3.7. Insulin secretion

3.7.1. Insulin secretion assay

Following 24 h treatment in 96-well plates, media was transferred to a fresh plate and stored at -80°C as 'chronic' insulin secretion samples, as opposed to stimulated insulin secretion following the addition of secretagogues such as glucose. Cells were washed in PBS and incubated for 40 min at 37°C in Krebs Ringer Bicarbonate Buffer (KRBB, 115 mM NaCl, 4.7 mM KCl, 2.5 mM CaCl₂, 1.2 mM KH₂PO₄, 1.2 mM MgSO₄.7H₂O, 24 mM NaHCO₃, 0.1% HEPES (v/v), 0.1% BSA (w/v), pH 7.4) in order to allow the cells to metabolise any residual glucose in the media. A subset of previously untreated cells had 5 mM sodium butyrate in the KRBB for a 40 min treatment. Subsequently, cells were washed in PBS and incubated for 20 min at 37°C in stimulation media (10 mM alanine and 16.7 mM glucose in KRBB), a combination that robustly and reproducibly promotes insulin secretion.¹⁹⁵ A subset of previously untreated cells included 5 mM sodium butyrate in the stimulation media and a further subset used only KRBB \pm 5 mM sodium butyrate to determine if butyrate can act as a secretagogue. Media was then transferred to a fresh plate at stored at -80°C for later insulin measurement. RIPA lysis buffer (Sigma-Aldrich) was added to cells for protein quantification using the Pierce BCA Protein Assay Kit (ThermoFisher).

3.7.2. Cellular insulin content

Following treatment as for insulin secretion, cells were washed with PBS and incubated with acid ethanol (1.5% HCl in 70% ethanol) overnight at 4°C. This solution was transferred to a fresh plate at stored at -80°C for later insulin quantification. RIPA lysis buffer was added to cells for protein quantification as for insulin secretion.

3.7.3. Insulin quantification

Insulin quantification assays were performed by Dr Gaewyn Ellison. Insulin was assayed by sandwich enzyme-linked immunosorbent assay (ELISA) using an ultrasensitive Rat Insulin ELISA kit (Mercodia) as per manufacturer's instructions. Briefly, samples and an enzyme-conjugated detection antibody were added to wells pre-coated with anti-insulin antibodies and incubated for 2 h at RT. Following washing with wash buffer, 3,3',5,5'-tetramethyl-benzidine (TMB) was added and colour was allowed to develop for 15 min. The reaction was stopped with 0.5 M H₂SO₄ and absorbance was measured with an absorbance wavelength of 450 nm.

3.8. Insulin signalling

206 μ M human insulin stocks were prepared by Dr Thiruvarutchelvan Sabapathy. 12 mg recombinant human insulin powder (Sigma-Aldrich) was dissolved in 450 μ l 0.02 M HCl. The solution was made up to 10 ml by the addition of 50 mM HEPES (4-(2-hydroxyethyl)-1-piperazineethanesulfonic acid) containing 0.1% BSA and the solution was sterilised with a Millex 0.22 um syringe filter prior to storage at -20°C. Following 24 h treatment in T25 flasks, 100 nM insulin was added directly to the treatment media. Cells were incubated for 7 min at 37°C prior to protein extraction as in section 3.2.4 and immunoblotting as in section 3.5.

3.9. Epigenetic modifying enzyme activity assays

3.9.1. HDAC activity

The *In Situ* Histone Deacetylase (HDAC) Activity Fluorometric Assay Kit (Sigma-Aldrich) was used to measure HDAC activity according to the manufacturer's instructions. The HDAC substrate was added to wells containing live cells directly to the treatment media after 23 h treatment with statins or HDAC inhibitors. For the nuclear extract experiments, the HDAC substrate was added directly with the treatments and nuclear extracts in phosphate buffered saline (PBS). After 1 h incubation, the HDAC developer was added, the plate incubated for a further 30 min and fluorescence read with an excitation wavelength of 368 nm and an emission wavelength of 442 nm.

Confirmatory experiments were carried out with the Fluor De Lys® HDAC fluorometric activity assay kit (Enzo Life Sciences) according to the manufacturer's instructions. HeLa nuclear extract provided with the kit and HepG2 nuclear extract prepared as above were incubated with treatments and substrate for 30 min. Following incubation with developer for 10 min, fluorescence was measured with an excitation wavelength of 360 nm and an emission wavelength of 460 nm.

3.9.2. HAT activity assays

HAT activity was measured using the EpiQuik[™] HAT Activity/Inhibition Assay Kit (Epigentek) according to the manufacturer's instructions. The HAT substrate was captured in strip wells for 45 min. Following washing, strip wells were incubated with acetyl-CoA, HepG2 nuclear extracts and treatments for 90 min at 37°C. Following washing, wells were incubated sequentially with the capture antibody and detection antibody, followed by the addition of the developer solution. Once the solution in the control wells changed to a medium blue, the enzymatic reaction was stopped with the stop solution and the absorbance measured at 450 nm.

3.9.3. DNMT activity assays

DNMT activity was measured with the abcam DNMT Activity Assay Kit according to the manufacturer's instructions, with the following modification: whole cell extracts were used as opposed to nuclear extracts as experiments with nuclear extracts were not successful. For direct DNMT inhibition experiments, untreated cell extracts were incubated with treatments and the Adomet methyl donor at 37° C for 2 h in strip wells coated with DNMT substrate. For treated cells, 20 µg of total protein was incubated with Adomet. Following incubation and washing, wells were incubated sequentially with the capture antibody, detection antibody and enhancer solution, followed by the addition of the developer solution. Once the solution in the control wells changed to a medium blue, the enzymatic reaction was stopped with the stop solution and the absorbance measured at 450 nm with a reference wavelength of 655 nm.

3.10. Chromatin immunoprecipitation

Histone modifications at specific genomic regions was determined by chromatin immunoprecipitation (ChIP) using the abcam ChIP Kit. Cross-linking with formaldehyde was used in pilot experiments (1% formaldehyde for 10 min at RT followed by addition of glycine to quench the formaldehyde) and was it was found that signals from immunoprecipitated samples were significantly enhanced when this step was skipped (Supplementary Figure S1). Cross-linking is not necessary in ChIP for histones modifications as histones are tightly linked to DNA.¹⁹⁶ Following treatment, cells were trypsinised and equal numbers were lysed in a series of buffers from the ChIP Kit according to the manufacturer's instructions. Chromatin was sheared by sonication in a buffer containing protease inhibitors for three times for 10 s following time course experiments that showed longer sonication times resulted in fragment sizes smaller than 200 bp. Chromatin was then incubated overnight at 4°C on a rotating rack with ChIP grade antibodies against H3 (positive control), H3K9ac, H4K8ac and H3K9me3 (Table 3.4), or frozen as input DNA. Bound chromatin was precipitated with the provided protein A sepharose beads for 1 h at 4° C on a rotating rack and beads were pelleted with centrifugation. Immunoprecipitated chromatin and input chromatin was then purified with the provided DNA purifying slurry and incubated at 98°C for 10 min. Proteinase K was then added and samples were incubated at 55°C for 30 min then 98°C for 10 min. The slurry was pelleted with centrifugation and the supernatant was used immediately for qPCR or frozen at -80°C.

Target	Туре	Product #	Company
H3	Rabbit polyclonal	ab1791	abcam
H3K9ac	Rabbit polyclonal	ab10812	abcam
H4K8ac	Rabbit polyclonal	ab15823	abcam
H3K9me3	Rabbit polyclonal	ab8898	abcam

Table 3.4Antibodies used for ChIP

3.10.1. ChIP qPCR

ChIP primers were designed using the protocol from Bridges Lab (http://bridgeslab.sph.umich.edu).¹⁹⁷ Genomic regions of interest were determined using data from the Encyclopedia of DNA Elements (ENCODE) project.¹⁹⁸ H3K9ac and SREBP-2 ChIP-seq data from HepG2 and other cell types from ENCODE were visualised in the UCSC Genome Browser, on the human Feb. 2009 (GRCh37/hg19) assembly (http://genome.ucsc.edu).¹⁹⁹ For SREBP-2 target genes, H3K9ac peaks located adjacent to SREBP-2 peaks were selected. For other genes, H3K9ac peaks with varying peak strength in HepG2 cells compared to other cell types were selected, typically upstream of the transcription start site (TSS). For GAPDH, used as a positive control, a region with strong H3K9ac peaks in multiple cell types was selected. DNA of the regions of interest was entered into Primer-BLAST (www.ncbi.nlm.nih.gov/tools/primer-blast) and human primers with a PCR product size of 70-150 bp were obtained. Primers were ordered through Sigma-Aldrich fully deprotected and desalted in de-ionised water at a concentration of 100 µM. ChIP primers are listed in Table 3.5.

qPCR was conducted as for gene expression (see section 3.6.3). Relative quantity (ΔC_q) was calculated by Bio-rad CFX Manager 3.1 software and percentage of input was calculated in Microsoft Excel.

Table 3.5	Primers	used	for	ChIP	qPCR
-----------	---------	------	-----	------	------

Genomic	Forward Primer	Reverse Primer	Annealing
region			Temperature
GAPDH	CCACATCGCTCAGACACCAT	CATACGACTGCAAAGAC	63°C
exon2		CCG	
HMGCR	TTCCCTCACGCTCCGATTCT	CTCAGTGCCTGACCCGTG	65°C
~500bp>TSS			
LDLR	CACGGGTTAAAAAGCCGAT	GGCCCACGTCATTTACAG	65°C
~100bp>TSS	GT	CA	
PPARG	ATCCGGGTCAACCTGACTAC	TCCTCCACAGCCCCTAAG	63°C
~1000bp>TSS		AT	
ABCA1	GGCACCAGTGGAATTTGCTT	CGTCCTGAGGGAGATTCA	65°C
~400bp>TSS		GC	
ABCG1	AGCCACACCCACCTGTTTTG	ACAGTGGGGAAGTAAGG	63°C
~600bp>TSS		CAC	
HK1	AGGTCAAAAAGGTGAGCCC	CTGCAGTCCAACTCGATG	63°C
intron 1	С	СТ	
SLC2A2	AGACCCTACAGGGCACAGA	CCTTGCTCCCTTATACGT	63°C
intron 1	Т	GGT	

>TSS- Upstream of transcription start site

3.11. Statistical analysis

At least three independent replicates were conducted of each experiment. Statistical significance was determined by analysis of variance (ANOVA) or Student's *t* test as indicated, with results considered significant if p<0.05. Analyses were conducted using GraphPad Prism software.

Chapter 4. Cholesterol metabolism

4.1. Abstract

Statins are widely used to reduce CVD risk by lowering plasma LDL-C, although they are associated with side effects, including increased risk of T2D. In animal studies, HDAC inhibitors such as butyrate have also been reported to reduce plasma cholesterol, while conferring protection from diabetes, but these studies lack investigation into the exacting mechanisms. This chapter compares the effects of statins and HDAC inhibitors on cholesterol metabolism in multiple cell lines, including hepatic cells due to the importance of the liver in cholesterol metabolism, insulin secreting cells due to the importance of cholesterol in insulin secretion and macrophages due to their role in cholesterogenic atherosclerosis. Cells were treated with statins and HDAC inhibitors and lipids were extracted and quantified. Cholesterol uptake was quantified using fluorescently tagged cholesterol. Expression of genes implicated in cholesterol synthesis and transport was assessed by immunoblotting and qPCR. Statins and the HDAC inhibitors sodium butyrate and TSA all reduced cellular cholesterol in hepatic and insulin secreting cells, while no treatment altered cholesterol in the macrophage cell line. Statins increased cholesterol uptake while butyrate had no effect. This is likely because statins increased SREBP-2 activity as a result of lowered cholesterol, increasing the expression of the LDLR receptor and genes involved in cholesterol synthesis. Conversely, butyrate inhibited SREBP-2 activity in a time-dependant manner. The effects of statins and butyrate on genes involved in reverse cholesterol transport, namely ABCA1 and SRB1, varied with cell type and levels of lipoprotein in the culture media. All genes with statin altered expression are controlled by transcription factors whose activity is influenced by cellular cholesterol levels, while butyrate altered expression of a significant proportion of genes involved in cholesterol synthesis and transport and thus may have unpredictable effects on lipid metabolism.

4.2. Objectives

1. Investigate how statins and HDAC inhibitors affect cholesterol metabolism.

4a. Identify genes involved in cholesterol metabolism or implicated in T2D whose expression is altered by these treatments.

4.3. Introduction

CVD is the leading cause of mortality worldwide.⁸ Dyslipidaemia, usually characterised by high LDL-C, high TG and/or low HDL-C levels, is a major risk factor for CVD. As a result, HMGCR inhibitors, commonly known as statins, are widely used to lower LDL-C; it is estimated that there are over 30 million statins users in the United States.²⁰⁰ By inhibiting HMGCR, the rate-limiting enzyme in the mevalonate pathway of cholesterol biosynthesis, statins decrease cellular cholesterol and subsequently increase the uptake of LDL-C from the bloodstream into cells, primarily hepatocytes. Despite the proven ability of statins to cut the morbidity and mortality from CVD by ~25%,¹³ there has been some controversy in recent years over the widespread use of statins, in particular in people at low risk of CVD.²⁰¹ Statin use has been associated with adverse effects, including myopathy, rhabdomyolysis, T2D and liver damage.¹³ Asymptomatic, transient elevations in aminotransferases are the most common liver complaint with statin use, although statins has also been associated with severe drug-induced liver injury. In particular, simvastatin and atorvastatin have each been implicated in over 60 published case reports of druginduced liver injury and have both been associated with fatal liver injury.²⁰² As a result, there is interest in exploring alternative lipid-lowering agents with lower toxicity and diabetogenic risk.

Butyrate is a SCFA produced by the bacterial fermentation of dietary fibre and is an established HDAC inhibitor. HDACs regulate a variety of metabolic pathways and deregulation of HDACs has been associated with metabolic disorders including T2D and CVD.²⁰³ In animal models of metabolic diseases, supplementation with sodium butyrate reportedly exerts numerous benefits, including reduced serum TG and cholesterol.^{164,165} *In vitro*, butyrate has been found to lower cholesterol synthesis in the Caco-2 colon cancer cell line, associated with a decrease in HMGCR activity.¹⁶⁹ The effect of butyrate on cellular cholesterol levels in liver cells, the major site of cholesterol biosynthesis; macrophages, where cholesterol accumulation contributes to the development of foam cells and atherosclerotic plaques; or insulin-secreting beta cells, where altered cholesterol may impact insulin secretion and thus diabetes risk,⁶³ has not been explored.

This chapter therefore explores the effects of HDAC inhibition and statins on various cell types, including liver cells, due to the importance of the liver in cholesterol
metabolism, and macrophages, due to their key role in cholesterogenic atherosclerosis. Effects in insulin secreting cells were also examined due to our laboratory group's interest in the effects of altered cellular cholesterol on insulin secretion. In these cells, effects on cellular cholesterol content, the SREBP-2 signalling pathway and cholesterol import/export were explored. This chapter also includes preliminary work to establish dosage, ensuring the doses used are not toxic to cells, do not alter total protein content and are sufficient to lower cellular cholesterol.

4.4. Results

4.4.1. Effects of statins and sodium butyrate on cell viability and protein content

Statins decreased HepG2 viability, as determined by the reduction of resazurin using the Alamar Blue® assay, in a dose-dependent manner (Figure 4-1A). This was most apparent for the lipophilic statins, atorvastatin and simvastatin, reducing viability to 13% (\pm 5% SEM) and 3% (\pm 5% SEM) respectively at 500 µM. The hydrophilic pravastatin and rosuvastatin were less toxic, with viability remaining at 70% (\pm 6% SEM) and 83% (\pm 17% SEM) respectively at 500 µM. All statins except pravastatin significantly reduced viability at 50 µM, while pravastatin only reduced viability at 500 µM. Simvastatin reduced viability at all tested concentrations, however no other statin affected viability at 10 µM, a concentration commonly employed in statin studies, as such this concentration was used for subsequent experiments.

Conversely, sodium butyrate did not significantly reduce cell viability at any of the tested concentrations, up to 20 mM (Figure 4-1B). Additionally, neither 10 μ M atorvastatin nor 1 mM or 5 mM sodium butyrate, concentrations used in subsequent experiments, altered the total protein content of HepG2 cells (Figure 4-2).



Figure 4-1 Influence of statins and butyrate on the viability of HepG2 cells

The effect of 24 h treatment with A) statins and B) sodium butyrate on HepG2 viability, measured using the AlamarBlue® assay as described in section 3.3. Results represent the mean from at least three independent experiments. Analysed by ANOVA. See Appendix 3 for full data tables.



Figure 4-2 Influence of butyrate and atorvastatin on protein content in HepG2 cells

Cell protein content was measured using the BCA assay following RIPA protein extraction as described in section 3.2.4. Results represent the mean from three independent experiments, with error bars representing SEMs. Analysed by ANOVA.

4.4.2. Sodium butyrate reduces cellular cholesterol to a similar extent as statins

In HepG2 cells, 5 mM sodium butyrate decreased cellular cholesterol content to a similar degree as treatment with 10 μ M statins (Figure 4-3A). After 24 h treatment in lipoprotein-deficient serum (LPDS), sodium butyrate treated cells showed a 31% (± 15% SEM) decrease in cholesterol content, comparable to the cholesterol lowering

effect of the hydrophilic statins rosuvastatin (44% \pm 15% SEM) and pravastatin (31% \pm 14% SEM), as well as atorvastatin (52% \pm 4% SEM) and simvastatin (53% \pm 8% SEM). Similar results were seen with BRIN-BD11 cells, although results were more variable and statistical significance differences were more marginal (Figure 4-3B). However, in THP-1 macrophages, which had significantly lower cholesterol content overall, neither statins nor butyrate affected cholesterol content after 24 h (Figure 4-3C). In a separate series of experiments, HepG2 cells were treated with both atorvastatin and sodium butyrate in order to determine if there were any additive or synergistic effects on cholesterol lowering. After 24 h treatment in LPDS, cells treated with both 5 mM sodium butyrate and 10 μ M atorvastatin had comparable cholesterol content to cells treated with atorvastatin alone (*p*=0.78) (Figure 4-3D).

Similar results were observed with cells treated in media containing the high lipoprotein fraction of FBS (HLPS), rather than the lipoprotein-deficient FBS (LPDS) supplemented media used in the other experiments, although the magnitude of cholesterol lowering was lower than in LPDS, most likely due to increased uptake of cholesterol from the media. In HLPS, cellular cholesterol was lowered by 28% (\pm 6% SEM) by atorvastatin, 20% (\pm 3% SEM) by sodium butyrate and by 33% (\pm 15% SEM) by both atorvastatin and sodium butyrate (Figure 4-3E). There was no significance difference in cholesterol levels between cells treated by both atorvastatin and sodium butyrate and by atorvastatin alone (*p*=0.5).

In order to determine if the cholesterol lowering is a result of HDAC inhibition or FFAR activation, HepG2 cells were treated with TSA, a potent HDAC inhibitor, and sodium acetate, a SCFA and FFAR agonist but with no known effect on HDAC activity.¹⁵⁸ After 24 h treatment in cells treated with TSA, cellular cholesterol content was decreased by 20% (\pm 9% SEM), while cholesterol content in sodium acetate treated cells was unaffected (Figure 4-3F). Furthermore, there was no difference in cholesterol content between cells treated with a combination of TSA and sodium acetate compared to those treated with TSA alone (*p*=0.5), suggesting HDAC inhibition as the mechanism by which sodium butyrate lowers cellular cholesterol.



Figure 4-3 Influence of statins and butyrate on cellular cholesterol content in HepG2 cells, BRIN-BD11 cells and THP-1 cells

Cellular cholesterol content as measured by the Amplex Red Cholesterol Assay Kit (Life Technologies) as described in section 3.4; in A) HepG2, B) BRIN-BD11 and C) THP-1 cells treated with 10 μ M statins or 5mM sodium butyrate (SB) for 24 h in media supplemented with lipoprotein-deficient serum (LPDS); D) in HepG2 cells treated with a combination of atorvastatin and sodium butyrate for 24 h in media containing LPDS; E) in HepG2 cells treated with statins or sodium butyrate for 24 h in media supplemented with high lipoprotein serum (HLPS); F) in HepG2 cells treated with sodium acetate (SA) or trichostatin A (TSA) for 24 h in media containing LPDS. Results represent the mean from at least three independent experiments, with error bars representing SEMs. * p<0.05, **p<0.01, ***p<0.001, ****p<0.0001, ns not significant. p values are in context of comparison to relevant controls, analysed by ANOVA.

4.4.3. Neither sodium butyrate nor statins alter cellular triglyceride content

In contrast to the effects on cellular cholesterol, neither sodium butyrate (Figure 4-4A), nor atorvastatin (Figure 4-4B) altered cellular TG content following 24 h treatment.



Figure 4-4 Influence of butyrate and atorvastatin on cellular triglyceride content in HepG2 cells

Cellular triglyceride content, as measured by the High Sensitivity Triglyceride Fluorometric Assay Kit (Sigma-Aldrich) as described in section 3.4.3, of HepG2 cells treated for 24 h with A) 1 mM or 5 mM sodium butyrate (SB); B) 10 μ M atorvastatin. Results represent the mean combined from at least three independent experiments, with error bars representing SEMs. Analysed by ANOVA for butyrate and Student's *t* test for a torvastatin.

4.4.4. Statins increase cholesterol uptake; sodium butyrate affects neither cholesterol uptake nor export

The primary mechanism by which statins lower circulating LDL-C is thought to be increased LDLR mediated LDL-C uptake by hepatocytes to compensate for the

reduced cellular cholesterol biosynthesis due to HMGCR inhibition.²⁰⁴ Fluorescently tagged cholesterol in the form of NBD-cholesterol was used to examine the effects of statins and sodium butyrate on cholesterol uptake in HepG2 cells (Figure 4-5A). As expected, 24 h treatment with atorvastatin increased the uptake of NBD-cholesterol, as did the included positive control treatment U-18666A, which inhibits cholesterol synthesis and intracellular cholesterol trafficking.²⁰⁵ Sodium butyrate had no effect on NBD-cholesterol uptake.

NBD-cholesterol was also used to examine cholesterol efflux. In these experiments, HepG2 cells were incubated with NBD-cholesterol for 48 h prior to treatment and fluorescence of the media was measured to detect exported NBD-cholesterol. Neither atorvastatin nor sodium butyrate affected the levels of NBD-cholesterol in media after 24 h treatment, although there was no positive control treatment in these experiments (Figure 4-5B). Furthermore, neither atorvastatin nor sodium butyrate affected secretion of apo-A1 or apo-B, presumably as component of lipoproteins (e.g., HDL and LDL respectively), in the media of treated cells after 24 h treatment (Figure 4-6).



Figure 4-5 Influence of butyrate and atorvastatin on cholesterol uptake and export in HepG2 cells from HepG2 cells

A) Uptake of fluorescently tagged NBD-cholesterol as described in section 3.4.4, by HepG2 cells treated with a torvastatin or sodium butyrate after 24 h; B) export of fluorescently tagged NBD-cholesterol from HepG2 cells treated with 10 μ M a torvastatin or 5 mM sodium butyrate after 24 h. Results represent the mean combined from at least three independent experiments, with error bars representing SEMs. * p<0.05, ***p<0.001, analysed by ANOVA.



Figure 4-6 Influence of butyrate and atorvastatin on apoprotein secretion from HepG2 cells

Immunoblots were completed using the method described in section 3.5, of A) a po-A1 and B) apo-B in the media of HepG2 cells treated with 5 mM sodium butyrate and 10 μ M atorvastatin for 24 h. The graphs represent the mean combined density readings from three independent experiments, with error bars representing SEMs. Analysed by ANOVA.

4.4.5. Sodium butyrate and statins have opposite effects on SREBP-2 signalling

SREBP-2 acts to increase intracellular cholesterol content by promoting the transcription of *LDLR* and cholesterol biosynthesis genes and is thought to be responsible for the increased LDL-C uptake seen in statin use.¹²¹ When cellular cholesterol levels are low, SREBP-2 is cleaved, releasing the active subunit for entry into the nucleus and subsequent activation of gene transcription. In HepG2 cells, sodium butyrate lowered protein levels of the SREBP-2 target LDLR, while atorvastatin increased LDLR (Figure 4-7A). Neither treatment significantly altered protein levels of another SREBP-2 target, HMGCR, or of the cleaved SREBP-2 protein itself after 24 h treatment (Figure 4-7B, C). Similarly, in BRIN-BD11 insulin-secreting cells, sodium butyrate lowered LDLR levels while atorvastatin tended to increase them (p=0.06), and there was no significant effect on HMGCR levels (Supplementary Figure S2). In THP-1 macrophages, 5 mM sodium butyrate significantly lowered levels of cleaved SREBP-2 (Figure 4-7F), while also tending to

lower levels of SREBP-2 targets LDLR (Figure 4-7D) and HMGCR (Figure 4-7E) (significant using Student's *t* test but not ANOVA). Atorvastatin tended to increase SREBP-2 and SREBP-2 targets but this only reached statistical significance for LDLR.





Immunoblots were completed using the method described in section 3.5 for A) LDLR¹, B) HMGCR and C) active SREBP-2² in HepG2 cells and of D) LDLR, E) HMGCR and F) active SREBP-2 in THP-1 cells following treatment with 1 mM (SB1) or 5 mM (SB5 or SB) sodium butyrate or 10 μ M atorvastatin for 24h. The graphs represent the mean combined density readings as normalised to GAPDH from three independent experiments, with error bars representing SEMs. * p<0.05 using ANOVA, ^ p<0.05 with two-tailed *t* test but not ANOVA.

1. LDLR contains multiple glycosylation sites which gives the appearance of multiple bands between 100-160 kDa (206)

2. SREBP-2 appears as two high molecular weight bands around 120 kDa representing the inactive form, and an additional band or bands around 68 kDa representing the cleaved form (207)

As sodium butyrate lowered active SREBP-2 without lowering cellular cholesterol in THP-1 macrophages, we theorised that butyrate may reduce SREBP-2 signalling at an earlier time point in HepG2 cells, lowering cholesterol and resulting in a compensatory increase of SREBP-2 signalling back to baseline levels. We therefore investigated SREBP-2 and HMGCR protein levels in HepG2 cells after 8 h and 16 h treatments. These experiments also included 16 h TSA treatment to determine if any effects seen are a likely result of HDAC inhibition. At 16 h, sodium butyrate and TSA significantly lowered HMGCR protein levels (

Figure 4-8A). Sodium butyrate (p=0.08 at 5 mM) and TSA also decreased active SREBP-2 levels at 16 h (

Figure 4-8B). qPCR was conducted on cells treated with 5 mM sodium butyrate for 16 h and showed a trend towards reduced *HMGCR* mRNA (p=0.0511) and no effect on *SREBF2* mRNA levels. This indicates that sodium butyrate likely alters SREBP-2 protein activity without altering its gene expression.



Figure 4-8 Influence of HDAC inhibition on the SREBP-2 pathway at 8 h and 16 h in HepG2 cells

Immunoblots were completed using the method described in section 3.5 for A) HMGCR and B) active SREBP-2 in HepG2 cells treated with 1 mM (SB1) and 5 mM (SB5) sodium butyrate for 8 h and 16 h or 2.5 μ M TSA for 16 h compared to the untreated control, normalised to GAPDH. Reverse-transcription qPCR was conducted using the method described in section 3.6 for C) *HMGCR* and D) *SREBF2* in HepG2 cells treated with 5 mM sodium butyrate for 16 h compared to the untreated control, normalised to reference genes *GAPDH*, *YWHAZ* and *RPL13A*. Results represent the mean combined from at least three independent experiments, with error bars representing SEMs. * p<0.05, ** p<0.01. Analysed by ANOVA (butyrate immunoblots) and Student's *t* test (TSA immunoblots and butyrate qPCR).

4.4.6. The effect of sodium butyrate and statins on proteins involved in reverse cholesterol transport depends on cell type and media lipoproteins

Reverse cholesterol transport involves the efflux of cholesterol from peripheral cells to the liver via HDL particles. ABCA1 mediates the transfer of cholesterol to apolipoprotein apo-A1 to form HDL, while SRB1 can bind HDL and allows for bidirectional transfer of cholesterol between cells and HDL.²⁰⁸ In HepG2 cells treated in LPDS supplemented media, sodium butyrate decreased protein levels of both ABCA1 and SRB1, while atorvastatin also decreased ABCA1 (Figure 4-9A, D). When HepG2 cells were treated in HLPS supplemented media, i.e., using the high lipoprotein fraction of FBS, atorvastatin increased ABCA1 and tended to decrease SRB1 (p=0.07), while sodium butyrate decreased ABCA1 and SRB1 as in LPDS (Figure 4-9B, E). In BRIN-BD11 cells in LPDS supplemented media, sodium butyrate increased, and atorvastatin tended to decrease ABCA1 levels (significant by Student's t test but not ANOVA) (Figure 4-9C). In THP-1 macrophages in LPDS supplemented media, sodium butyrate increased levels of SRB1 (Figure 4-9F), while ABCA1 was decreased by both sodium butyrate and atorvastatin (Supplementary Figure S3). The decrease in ABCA1 protein levels by sodium butyrate in HepG2 cells in LPDS supplemented media was evident from 8 h and was also observed with TSA, suggesting HDAC inhibition as the likely mechanism (Figure 4-10). The effect of sodium butyrate and statins on ABCA1 in different cell types and media lipoprotein conditions is summarised in Table 4.1.

Table 4.1	Influence	of butyrate	and statins	on ABCA	1 protein	levels	in
different cell t	ypes						

Treatment	HepG2 cells	HepG2 cells	THP-1 cells	BRIN-BD11
	in LPDS	in HLPS	in LPDS	cells in LPDS
Sodium	↓****	↓**	↓**	^ *
butyrate				
Atorvastatin	↓**	^*	↓**	\downarrow (<i>p</i> =0.1)

** *p*<0.01, *** *p*<0.001, **** *p*<0.0001.





Immunoblots were completed using the method described in section 3.5 for ABCA1 in HepG2 cells treated in A) LPDS supplemented media and B) HLPS supplemented media, and in C) BRIN-BD11 cells treated in LPDS supplemented media, and of SRB1 in HepG2 cells treated in D) LPDS supplemented media and E) HLPS supplemented media, and in F) BRIN-BD11 cells treated in LPDS supplemented media. Cells were treated with 1 mM (SB1) or 5 mM (SB5) sodium butyrate, or 10 μ M atorvastatin for 24 h. The graphs represent the mean density readings as normalised to GAPDH from three independent experiments, with error bars representing SEMs. * *p*<0.05, ** *p*<0.01, *** *p*<0.001, **** *p*<0.0001 compared to control. Analysed by ANOVA



Figure 4-10 Influence of HDAC inhibition on ABCA1 at 8 h and 16 h in HepG2 cells

Immunoblots were completed using the method described in section 3.5 for ABCA1 in HepG2 cells treated with 1 mM (SB1) and 5 mM (SB5) sodium butyrate for 8 h and 16 h or TSA for 16 h. The graph represents the mean density readings as normalised to GAPDH from three independent experiments, with error bars representing SEMs. ** p<0.01, *** p<0.001, **** p<0.001. Analysed by ANOVA (butyrate) and Student's *t* test (TSA)

4.4.7. Sodium butyrate alters the expression of multiple cholesterol related genes, atorvastatin has more targeted effects on gene expression

qPCR was employed to examine the effects of sodium butyrate and atorvastatin on gene expression at the mRNA level. Sodium butyrate had a significant effect on mRNA levels of numerous cholesterol related genes. In particular, *ABCG1* was upregulated more than a hundred-fold in HepG2 cells, as was *CAV1*, encoding caveolin-1, the key protein in caveolae lipid rafts (Figure 4-11A). It should be noted that this robust upregulation did not result in a significant increase in protein levels of ABCG1 or caveolin-1 (Figure 4-12). *SREBF1*, *SCARB1*, encoding SRB1, and *APOA1* were also upregulated while *ABCA1* and *CYP7A1*, encoding the rate-limiting enzyme in bile acid production were downregulated. A limited number of cholesterol related genes were also examined in BRIN-BD11 cells. Sodium butyrate upregulated *ABCA1*, *ABCG1* and *SREBF1* and downregulated *LDLR* in BRIN-BD11 cells (Figure 4-11B).

Atorvastatin had more limited effects on gene expression. In HepG2 cells, atorvastatin significantly upregulated *HMGCR* expression and downregulated *SREBF1* (Figure 4-11C)



Figure 4-11 Influence of butyrate and atorvastatin on cholesterol-related gene expression in HepG2 cells and BRIN-BD11 cells

Reverse-transcription qPCR was conducted using the method described in section 3.6 for A) HepG2 and B) BRIN-BD11 cells treated 5 mM sodium butyrate and for C) HepG2 and D) BRIN-BD11 cell treated with 10 μ M atorvastatin compared to the untreated control, normalised to reference genes *GAPDH*, *YWHAZ* and *RPL13A* for HepG2 cells and reference genes *Actb* and *Rpl13a* for BRIN-BD11 cells. Atorvastatin data is shown relative to the untreated control but statistical significance is compared to the DMSO vehicle control. Results represent the mean combined from at least three independent experiments, with error bars representing SEMs. * p<0.05**p<0.01, ***p<0.001, ***p<0.0001. Analysed by ANOVA.



Figure 4-12 Influence of butyrate on protein levels of upregulated genes in HepG2 cells

Immunoblots were completed using the method described in section 3.5 for A) ABCG1 and B) caveolin-1 in HepG2 cells treated with 1 mM or 5 mM sodium butyrate or $10 \,\mu$ M atorvastatin for 24 h. The graphs represent the mean density readings as normalised to GAPDH from three independent experiments, with error bars representing SEMs. Analysed by ANOVA.

4.5. Discussion

4.5.1. Mechanisms of cholesterol lowering by statins and butyrate

Statins are competitive inhibitors of HMGCR, the rate-limiting step in cholesterol biosynthesis. By inhibiting cholesterol biosynthesis and thus reducing cellular cholesterol, particularly in the liver, statins cause cells to increase cholesterol uptake from the bloodstream, lowering plasma LDL-C. This is mediated by SREBP-2, which is activated when cellular cholesterol levels drop, resulting in increased expression of LDLR.¹²¹ Our experiments confirmed increased LDLR expression and cholesterol uptake with atorvastatin treatment in hepatic cells. Conversely, our results indicate that butyrate may inhibit SREBP-2 activity, resulting in reduced LDLR protein levels. Although cleaved (active) SREBP-2 or HMGCR protein levels were not altered after 24 h butyrate treatment in hepatic cells, both proteins were downregulated in macrophages despite cholesterol not being reduced by butyrate in these cells after

24 h. The macrophages had significantly lower cholesterol content than the HepG2 cells and the BRIN-BD11 cells; the lack of cholesterol lowering by the treatments may be because these macrophages do not synthesise as much cholesterol. Consequently, SREBP-2 and HMGCR protein levels were examined at earlier time points in butyrate treated hepatic cells and HMGCR was decreased after 16 h treatment, with a trend towards decreased levels of cleaved SREBP-2. TSA, a potent HDAC inhibitor, also decreased HMGCR and cleaved SREBP-2 protein levels at 16 h and cholesterol content at 24 h, while sodium acetate, a SCFA with no HDAC inhibitory activity, failed to impact cholesterol levels. Thus, it is likely that butyrate decreases cholesterol biosynthesis through HDAC inhibition and a decrease in SREBP-2 activity, lowering cellular cholesterol content and thereby causing a compensatory increase in SREBP-2 signalling back to baseline levels. While downregulation of HMGCR has been reported in butyrate treated intestinal cells, with studies finding reduced mRNA levels¹⁷⁰ and protein activity¹⁶⁹ in Caco-2 enterocytes, this time dependent decrease is a novel finding, as is the discovery that this is mediated via HDAC inhibition.

The qPCR results suggest that butyrate lowered HMGCR but not SREBF2 mRNA after 16 h treatment, implying that butyrate alters SREBP-2 activity but not the expression of its gene. Similar results were seen in a study of TSA treated neuronal cells by Nunes et al.¹¹⁰ which found that HDAC inhibition reduced SREBP-2 target gene expression and active SREBP-2 protein levels without significantly altering SREBF2 gene expression. The SREBP-2 protein itself can be acetylated and this acetylation alters its localisation and activity.¹¹² In particular, the class III HDAC SIRT1 deacetylates SREBP-2, and SIRT1 inhibition increases levels of the active nuclear SREBP-2 protein and expression of SREBP-2 target genes.²⁰⁹ Sodium butyrate does not inhibit class III HDACs,²¹⁰ and it has not been investigated what effect class I and II HDACs, inhibited by butyrate, have on the acetylation of SREBPs or whether inhibition of class I and II HDACs causes a compensatory increase in class III HDAC activity. Another possibility is that butyrate decreases SREBF2 mRNA at an earlier timepoint than investigated in this study. Chittur et al.²⁵ reported that TSA downregulated SREBF2 mRNA expression in HepG2 cells, with maximal repression occurring at 9 h treatment and levels normalising at 48 h.

While a decrease in SREBP-2 activity would explain the lowered cellular cholesterol, this calls into question whether butyrate could reduce plasma cholesterol and thus

CVD risk, especially as there was no effect of butyrate on hepatic cholesterol uptake. Although numerous rodent studies have reported reduced total plasma cholesterol with butyrate supplementation,^{118,165,167,211} several studies found no effect on LDL-C levels.^{167,212} In human clinical trials, one placebo-controlled study (15 individuals per treatment group) found 45 days of butyrate supplementation did not significantly alter plasma cholesterol,⁶ while a study of nine healthy and ten metabolic syndrome participants reported that butyrate increased LDL-C levels in individuals with metabolic syndrome, although this study lacked a placebo group.¹⁶⁶ As HDL, as opposed to LDL, is the primary plasma cholesterol carrier in rodents, ²¹³ reduced HDL due to the noted downregulation of ABCA1 could account for the reduced plasma cholesterol in rodent studies and biological differences could account for the lack of effect in the human study. More large-scale trials are clearly needed to determine if butyrate supplementation alters plasma LDL-C in humans. The lowering of hepatic cholesterol seen with butyrate may be beneficial in and of itself, for example in NAFLD, where the accumulation of free cholesterol is thought to be a contributor to liver damage.³⁰ Indeed, animal models have shown sodium butyrate to be protective against diet-induced liver damage in mice.214,215

4.5.2. Effects of atorvastatin and butyrate on reverse cholesterol transport proteins

The effect of sodium butyrate and atorvastatin on proteins involved in HDL metabolism, namely ABCA1 and SRB1, varied according to cell type and the presence or absence of lipoproteins in serum supplements. ABCA1 mediates the efflux of cholesterol and phospholipids to apolipoprotein apo-A1 to form HDL particles.²¹⁶ ABCA1 expression is regulated by LXR, a nuclear receptor activated by oxysterols, cholesterol metabolites that occur in proportion to total cholesterol.³⁹ In the absence of ligands, LXR acts to repress target genes. Thus, ABCA1 is expressed proportionally to cellular cholesterol, providing a mechanism by which the cholesterol lowering agents atorvastatin and sodium butyrate decreased ABCA1 levels in liver cells in low lipoprotein conditions. Paradoxically, in high-lipoprotein supplemented media, atorvastatin increased ABCA1 levels in hepatic cells, despite lowering cellular cholesterol in these conditions to a similar if not greater extent than sodium butyrate. Similar results have been reported in THP-1 macrophages, in which mevastatin reduced ABCA1 and ABCG1 expression in normal cells, with this effect reversed by cholesterol loading.²¹⁷ Atorvastatin may therefore increase LXR activity,

both to repress or activate its target genes depending on oxysterol levels. Statins have been shown to improve HDL-C in humans clinical trials,²¹⁸ an increase in LXR activity resulting in upregulated ABCA1 expression could account for this effect. Sodium butyrate affects ABCA1 levels in disparate ways, lowering it in all tested conditions except in BRIN-BD11 cells. The fact that the decrease in ABCA1 was apparent after only 8 h and was also observed in THP-1 macrophages despite the lack of cholesterol lowering seen in these cells suggests that butyrate can impact ABCA1 independent of cholesterol levels. As TSA also lowered ABCA1 protein from 8 h, HDAC inhibition is likely the contributing mechanism. It has been shown that LXR is acetylated and deacetylation by SIRT1 increases its activity;²¹⁹ HDAC inhibition could therefore decrease LXR activity and thus ABCA1 expression. However, ABCG1 and SREBP-1 are also LXR targets,³⁹ and their mRNA expression was increased by sodium butyrate in both HepG2 and BRIN-BD11 cells. It is unclear why ABCA1 is upregulated in BRIN-BD11 cells, especially as cellular cholesterol in these cells was reduced to a similar extent as in HepG2 cells. The downregulation of ABCA1 in macrophages could impair cholesterol efflux to HDL and thus increase cholesterol accumulation in these cells, while the upregulation of both ABCA1 and ABCG1 in insulin secreting cells suggests increased cholesterol efflux. It would be interesting to examine the effects of HDAC inhibition on ABCA1 in other peripheral cells, especially adipocytes and myocytes, while cholesterol import/export experiments, specifically using HDL-C, in multiple cell types could help elucidate how these alterations in ABCA1 impact cholesterol efflux and thus RCT.

SRB1 acts as an HDL receptor and thus facilitates bidirectional movement of cholesterol between cells and HDL.²²⁰ There is uncertainty over the transcriptional regulation of SRB1. Its promoter contains potential SREBP-1a, SREBP-2,²²¹ and LXR²²² binding sites. In macrophages, SRB1 levels have been reported to inversely correlate with cholesterol levels, while neither expression of constitutively active SREBPs nor knockout of LXR affected SRB1 expression.²²³ It is also unclear how the increased macrophage SRB1 observed with butyrate may affect overall cholesterol levels; Ji et al.²⁰⁸ found that SRB1 contributed to both efflux and influx of cholesterol between mouse bone marrow-derived macrophages and HDL and as a result, there was no overall difference in cholesterol contents between wild type and SRB1 knockout cells. Conversely, evidence suggests that in hepatic cells, SRB1 mediated cholesterol uptake exceeds cholesterol efflux; liver specific SRB1 overexpressing mice have lower plasma HDL and increased hepatic uptake of

cholesterol esters from HDL,²²⁴ while liver specific SRB1 knockout mice have increased plasma HDL due to reduced hepatic uptake.²²⁵ The downregulation of hepatic SRB1 and ABCA1 seen with sodium butyrate may therefore result in impaired RCT, which could be examined by performing cholesterol uptake experiments using only HDL-C.

4.5.3. Effects of butyrate and atorvastatin on expression of cholesterol genes

Butyrate is a pan-HDAC inhibitor, inhibiting both Class I and Class II HDACs and thus increasing histone acetylation.²¹⁰ As histone acetylation is considered to be a sign of active gene expression, it is unsurprising that butyrate is associated with altered expression of a significant number of genes; as mentioned previously, it has been suggested that approximately 10% of genes are significantly altered by butyrate.¹⁶⁰ While we only examined at a small selection of cholesterol related genes, we found butyrate significantly altered the expression of seven out of 13 genes (five upregulated and two downregulated) in HepG2 cells. Upregulated genes include genes involved in cholesterol transport (ABCG1, SCARB1), cholesterol and fatty acid synthesis (SREBF1), lipid rafts (CAV1) and HDL-C (APOA1), while genes involved in bile acid synthesis (CYP7A1) and cholesterol efflux (ABCA1) were downregulated. In BRIN-BD11 cells, genes involved in cholesterol transport and efflux (ABCA1, ABCG1) and cholesterol and fatty acid synthesis (SREBF1) were upregulated while LDLR, involved in cholesterol uptake, was downregulated. As later discussed in Chapter 6, we found no significant changes to histone modifications detected in proximity to ABCA1, ABCG1 or LDLR in butyrate treated HepG2 cells (see section 6.4.6), therefore suggesting that other pathways are more likely to be involved, such as via influences on non-histone acetylation of LXR and SREBP-2. It may also be that there are histone modifications in proximity to these genes that were not detected with the specific modifications and genomic regions examined.

Conversely atorvastatin, as an HMGCR inhibitor, has narrower effects on the expression of the examined genes, only significantly upregulating *HMGCR* and downregulating *SREBF1* in HepG2 cells, while *LDLR* was upregulated and *ABCA1* and *ABCG1* were downregulated in BRIN-BD11 cells. These same changes are also evident in the HepG2 cells, albeit below the threshold for statistical significance. These effects are associated with the mechanism of statins, i.e., inhibition of HMGCR and thus cholesterol biosynthesis. All the genes altered by atorvastatin treatment are

those activated by SREBP-2 (*LDLR*, *HMGCR*), or silenced by LXR (*ABCA1*, *ABCG1*, *SREBF1*) in response to low cellular cholesterol levels.³⁹ Similar results have been obtained with human samples using microarrays to interrogate thousands of genes. For example, Obeidat et al.²²⁶ compared gene expression in the blood cells of statin users to that of non-users with a microarray containing probes for 28000 genes, and found 25 (<0.01%) with altered gene expression, including upregulation of *LDLR* and downregulation of *ABCG1*. These were significantly enriched in cholesterol pathways and 18 of these 25 genes were bound by SREBP-2.

Table 4.2 summarises the influence of butyrate and atorvastatin on gene expression, including mRNA and protein levels, and the roles of those genes on cholesterol metabolism. While it is tempting to look into the genes altered by butyrate and speculate how this could affect cholesterol metabolism, it should be noted that gene expression does not always reflect protein levels or activity. For example, although protein levels of SRB1, encoded by the *SCARB* gene, were decreased by butyrate in HepG2 cells, *SCARB* mRNA was increased, possibly as a compensatory mechanism. Furthermore, the 10-fold increase in *SREBF1* mRNA, encoding SREBP-1 which activates both fatty acid and cholesterol synthesis,²²⁷ did not result in an increase in cellular triglyceride or cholesterol synthesis. Studies relying only on mRNA data should thus be treated with caution.

Gene	Butyrate		Atorvastatin		Role in cholesterol metabolism
HepG2	qPCR	WB	qPCR	WB	
ABCA1	Ļ	Ļ	ns	Ļ	Efflux cholesterol from cells to HDL.
LDLR	ns	Ļ	ns	1	Uptake of LDL-C into cells.
SCARB1	1	Ļ	ns	ns	Bidirectional transfer of cholesterol
(SRB1)					between cells and HDL.
SREBF2	ns	ns	ns	ns	Transcription factor, increases cholesterol
					synthesis and uptake.
NR1H3	ns		ns		Transcription factor, increases cholesterol
(LXR)					efflux.
ABCG1	1	ns	ns	ns	Transport of cholesterol, intracellular and to
					HDL.

Table 4.2Influence of butyrate and atorvastatin on a selection of genesinvolved in cholesterol metabolism.

Gene	Butyra	te	Atorvastatin		Role in cholesterol metabolism
HMGCR	ns	ns	1	ns	Cholesterol biosynthesis.
SREBF1	1		Ļ		Transcription factor for fatty acid synthesis
					and cholesterol metabolism.
APOA1	1	ns	ns	ns	Component of HDL.
CYP7A1	Ļ		ns		Bile acid synthesis.
MVK	ns		ns		Cholesterol biosynthesis.
CAV1	1	ns	ns	ns	Component of lipid rafts in plasma
					membrane.
АРОВ	ns	ns	ns	ns	Component of LDL.
BRIN-BD	11				
Abca1	1	1	↓	↓	Efflux cholesterol from cells to HDL.
Abcg1	1		Ļ		Transport of cholesterol, intracellular and to
					HDL.
Ldlr	↓	↓	1	1	Uptake of LDL-C into cells.
Srebf1	1		ns		Transcription factor for fatty acid synthesis
					and cholesterol metabolism.

4.6. Conclusion

This chapter describes the most comprehensive analysis of the effects of butyrate on cholesterol metabolism, in particular comparing and contrasting it with the commonly used LDL-C lowering drug atorvastatin. Previous animal studies have reported reduced serum cholesterol with butyrate supplementation, and it has been suggested that butyrate inhibits cholesterol biosynthesis *in vitro*. However, lowering of cellular cholesterol by butyrate has not previously been demonstrated. Furthermore, the results presented here suggest HDAC related influences, rather than FFAR activation, as the mechanism, with a time-dependent decrease in SREBP-2 signalling the most likely exacting mechanism. In contrast to statins, the cellular cholesterol lowering by butyrate does not increase uptake of cholesterol by hepatic cells and thus calls into question whether butyrate could lower serum LDL-C in a similar manner.

The effects of butyrate on proteins involved in RCT were also in contrast to statins, where noted effects were directly related to cholesterol levels. The effects of butyrate on these proteins differed according to cell type and as such it is difficult to draw conclusions on how this may affect HDL metabolism and RCT at the organismal level. The effects of statins on gene expression are limited and directly tied to cellular cholesterol levels, whereas butyrate alters the expression of significantly more genes to a greater degree, reflecting their different mechanisms as HMGCR inhibitors and HDAC inhibitors respectively. Combined with the differences in cholesterol metabolism between humans and rodents, and the poor systemic availability of oral butyrate, there remains numerous doubts and questions over how butyrate may affect cholesterol metabolism in humans.

Chapter 5. Glucose metabolism

5.1. Abstract

The widely used cholesterol lowering drugs statins have been found to increase the risk of developing T2D in humans. Previous studies by our group have found that altering cellular cholesterol, whether by statins or other means, impairs insulin secretion and signalling. Conversely, HDAC inhibitors have been reported to reduce both cholesterol and diabetes risk in animal models of metabolic disease. This chapter explores the effects of statins and HDAC inhibitors on insulin secretion from pancreatic cells and on insulin signalling in hepatic cells, while also examining effects on the expression of genes implicated in T2D in these cells. Insulin secreting cells were treated with HDAC inhibitors and stimulated with glucose and alanine. Insulin was quantified from the 24 h treatment media and from the stimulation media. Hepatic cells were treated with statins and HDAC inhibitors and stimulated with insulin. Insulin stimulation was assessed by immunoblotting of proteins extracted from stimulated and non-stimulated cells. Gene expression was assessed bv immunoblotting and qPCR. HDAC inhibition decreased chronic 24 h insulin secretion and the insulin content of cells. Robust HDAC inhibition with 5 mM butyrate or TSA for 24 h in insulin secreting cells decreased basal insulin secretion and content, as well as insulin secretion in response to acute stimulation. Treatment with butyrate also increased expression of the disallowed gene hexokinase I, possibly explaining the impairment to insulin secretion, and of TXNIP, which may increase oxidative stress and β cell apoptosis. In contrast to robust HDAC inhibition (>70% after 24 h), low dose and acute high dose treatment with butyrate enhanced nutrient stimulated insulin secretion. All the genes implicated in insulin secretion and diabetes altered by stating are controlled by cellular cholesterol, adding evidence to the theory that the diabetogenic effects of statins are a result of cholesterol lowering rather than an off-target effect. For insulin signalling, butyrate decreased AKT phosphorylation and tended to increase non-stimulated IRS1 phosphorylation, consistent with previous studies showing phosphorylation of these proteins is impacted by their acetylation. The *in vitro* effects of butyrate on insulin secretion and signalling contrast with the reported *in vivo* effects, which may be more reflective of effects on other tissues, such as GLP-1 secretion from intestinal cells.

5.2. Objectives

2. Investigate how statins and established epigenetic modifiers affect insulin secretion and signalling.

4a. Identify genes involved in cholesterol metabolism or implicated in T2D whose expression is altered by these treatments.

5.3. Introduction

Dyslipidaemia and T2D are closely linked,¹¹ and *in vitro*, increased cholesterol has been found to impair insulin secretion in β cells⁶³ and glucose uptake in muscle cells.²²⁸ Reduced plasma LDL-C in humans, whether through the use of statins¹⁴ or genetic mutations,⁷⁰ has also been associated with an increased risk of T2D. Conversely, although results in section 4.4.2 indicate HDAC inhibition also decreases cellular cholesterol, animal studies have suggested that HDAC inhibition may in fact be protective against diabetes.^{165,167,180,229}

Previous studies in our group assessed the effects of altered cholesterol by statins and other means, namely using methyl- β -cyclodextrins to add or remove cholesterol from plasma membranes, on insulin secretion from β cells⁷² and insulin binding and signalling in a number of cell types.⁷¹ These studies found that both increasing and decreasing cellular cholesterol impaired insulin secretion and action. In this chapter, the effects of HDAC inhibition, another method of cholesterol depletion, on insulin secretion in β cells and insulin signalling in hepatic cells are examined and compared to those of statins. The mechanisms behind how statins and HDAC inhibition alter insulin secretion and potentially T2D risk were explored by examining the effects of the HDAC inhibitor sodium butyrate and the statin atorvastatin on numerous candidate genes implicated in T2D. In BRIN-BD11 cells this included genes involved in insulin secretion and other aspects of β cell function and identity, while the use of hepatic cells enabled the examination of genes involved in insulin signalling, glucose uptake, glucose sensing and gluconeogenesis.

5.4. Results

5.4.1. HDAC inhibition impairs insulin secretion and reduces insulin content

Previous investigation by our laboratory found that statins blunted insulin secretion stimulated by glucose and alanine in BRIN-BD11 cells without significant impact on 24 h secretion.⁷² A series of experiments were conducted in order to determine if HDAC inhibition, which has a similar effect on cellular cholesterol, impacts insulin secretion in a similar manner. Treatment of BRIN-BD11 cells with 5 mM sodium butyrate (Figure 5-1A) and 2.5 µM TSA (Figure 5-1B) reduced chronic insulin secretion into culture media over the 24 h treatment period, conducted in the present work in media containing 11.1 mmol/L glucose as is standard.²³⁰ Subsequent acute stimulation with 16.7 mM glucose and 10 mM alanine (a widely accepted positive control condition)²³¹ caused significant increases in insulin secretion in control cells and in cells treated for 24 h with 1 mM sodium butyrate, but this response was significantly blunted in cells treated with 5 mM sodium butyrate or $2.5 \,\mu\text{M}$ TSA for 24 h. The insulin content of these cells was also reduced (Figure 5-1C). The percentage of cellular insulin content secreted by cells acutely stimulated with glucose and alanine was also decreased by TSA (Figure 5-1D), thus the reduction in insulin secretion was not completely accounted for by a decrease in the insulin content of the cells. By contrast, 1 mM sodium butyrate significantly increased the percentage of insulin secreted.





The effect of 24 h exposure to A) sodium butyrate (SB) and B) 2.5 μ M trichostatin A (TSA) on 24 h (chronic) insulin secretion and in response to acute stimulation by 16.7 mM glucose and 10 mM alanine for 20 min (stimulated) using the method described in section 3.7. C) The effect of 24 h treatment of sodium butyrate and TSA on BRIN-BD11 cellular insulin content, determined using the method described in section 3.7.2, and D) the percentage of total insulin secreted in response to stimulation per the insulin content of unstimulated cells. Results are mean combined from at least three independent experiments + SEM. * p<0.05, ** p<0.01, *** p<0.001, **** p<0.001 compared to the relative untreated control. Analysed by ANOVA.

5.4.2. Butyrate acts as an acute β cell secretagogue

The previous results demonstrated that 24 h treatment of BRIN-BD11 cells with 5 mM sodium butyrate impaired their ability to respond to acute stimulation by glucose and alanine. Conversely, in the absence of any prior treatment with butyrate, inclusion of 5 mM butyrate in the stimulation media enhanced the secretory response to alanine and glucose (Figure 5-2A). Butyrate also tended to increase insulin secretion in the absence of alanine and glucose, however this was not significant (p=0.09). To determine whether this may be due to effects of HDAC inhibition on gene expression, cells were treated with butyrate for 40 min, before addition of the

stimulation media without butyrate. This short-term butyrate treatment failed to significantly enhance stimulated insulin secretion (p=0.43) (Figure 5-2B). The lack of stimulation following the removal of butyrate suggests that the acute stimulatory effect is unlikely to be due to epigenetic effects and may instead be due to effects mediated via FFAR binding and activation, or due to metabolism of butyrate as an energy source.





5.4.3. Butyrate interferes with insulin signalling in hepatic cells

Reduced insulin sensitivity, or insulin resistance, usually precedes impaired insulin secretion in the development of T2D.²³² The effect of HDAC inhibitors and statins on insulin signalling was thus investigated in HepG2 hepatic cells with or without insulin stimulation (Figure 5-3). Sodium butyrate impaired insulin stimulated AKT phosphorylation and tended to decrease baseline (unstimulated) phosphorylation (p=0.2). On the other hand, butyrate tended to increase the baseline phosphorylation of insulin receptor substrate-1 (IRS-1) (p=0.066), an effect not seen in insulin stimulated cells.

Atorvastatin tended to increase insulin stimulated IRS-1 (p=0.09) but this was not statistically significant compared to the DMSO vehicle control.

In general, the quality of the blots was variable, and the error bars are often large which may have contributed to the lack of statistically significant results.





Immunoblots were completed using the method described in section 3.5 for HepG2 cells treated with 5 mM sodium butyrate or 10 μ M atorvastatin, showing the phosphorylation of IRS-1, IR β and AKT with or without 7 min stimulation with 100 nM insulin as described in section 3.8. The graphs represent the mean combined density readings as normalised to GAPDH from at least three independent experiments, with error bars representing SEMs. ** p<0.01. Analysed by ANOVA.

5.4.4. Butyrate increases protein levels of the disallowed gene hexokinase I in HepG2 and BRIN-BD11 cells

Hexokinases phosphorylate glucose, and other hexoses, and for glucose this creates glucose 6-phosphate in the key first step of cellular glucose utilisation. Hexokinase I (HK1), which has a low Km, is usually silenced in β cells and liver cells in favour of glucokinase, which has a higher Km,⁴⁷ although some residual expression is a feature of BRIN-BD11 cells.²³³ 24 h exposure to 5 mM butyrate caused a substantial increase in HK1 levels in both HepG2 (Figure 5-4A) and BRIN-BD11 cells (Figure 5-5A). Atorvastatin had no effect. Neither sodium butyrate nor atorvastatin significantly altered protein levels of glucokinase in either HepG2 (Figure 5-4B) or BRIN-BD11

cells (Figure 5-5B). Furthermore, neither sodium butyrate nor atorvastatin significantly altered protein levels of the glucose transporter GLUT2 (Figure 5-4C) or the gluconeogenic enzyme PEPCK (Figure 5-4D) in HepG2 cells.

Signs of increased expression of HK1 in HepG2 cells were apparent at 8 h of exposure to 5 mM butyrate and with 16 h treatment of TSA, although this was not statistically significant as the degree of upregulation varied between replicate experiments, reflected by the large SEMs (Figure 5-6).

Protein levels of cholesterol related genes which have also been implicated in diabetes were altered by both butyrate and atorvastatin; butyrate reduced protein levels of LDLR, SRB1 and ABCA1 in HepG2 cells and reduced LDLR but increased ABCA1 in BRIN-BD11 cells while atorvastatin increased LDLR and reduced ABCA1 in HepG2 and BRIN-BD11 cells (see section 4.4).



Figure 5-4 Influence of butyrate and atorvastatin on glucose metabolism proteins in HepG2 cells

Immunoblots were completed using the method described in section 3.5 for A) hexokinase I, B) glucokinase, C) GLUT2 and D) PEPCK in HepG2 cells following treatment with 1 mM (SB1) or 5 mM (SB5) sodium butyrate or 10 μ M atorvastatin for 24 h. The graphs represent the mean combined density readings as normalised to GAPDH from three independent experiments, with error bars representing SEMs. ** *p*<0.01. Analysed by ANOVA.



Figure 5-5 Influence of butyrate and atorvastatin on hexokinases in BRIN-BD11 cells

Immunoblots were completed using the method described in section 3.5 for A) hexokinase I and B) glucokinase in BRIN-BD11 cells following treatment with 1 mM (SB1) or 5 mM (SB5) sodium butyrate or 10μ M atorvastatin for 24h. The graphs represent the mean combined density readings as normalised to GAPDH from three independent experiments, with error bars representing SEMs. * p<0.05. Analysed by ANOVA.



Figure 5-6 Influence of HDAC inhibition on hexokinase I in HepG2 cells at 8 h and 16 h

Immunoblots were completed using the method described in section 3.5 for hexokinase I in HepG2 cells following treatment with 1 mM (SB1) or 5 mM (SB5) sodium butyrate for 8 or 16 h or with

 $2.5 \,\mu$ M TSA for 16 h. The graph represents the mean combined density readings as normalised to GAPDH from three independent experiments, with error bars representing SEMs. Analysed by ANOVA and Student's *t* test.

5.4.5. But yrate significantly alters expression of numerous T2D candidate genes, atorvastatin has more limited effect on gene expression

5 mM sodium butyrate had significant effect on the mRNA expression of numerous genes that have been linked to T2D and/or glucose metabolism as determined by RT qPCR. Notably, hexokinase was upregulated more than 100-fold in HepG2 cells (Figure 5-7A) and was also upregulated in BRIN-BD11 cells, albeit to a lesser degree (Figure 5-7B). Furthermore, glucokinase was significantly downregulated by butyrate in HepG2 cells. *SLC2A2*, the gene encoding GLUT2, was upregulated by butyrate in HepG2 cells, as were the lipid related genes implicated in glucose metabolism, *ABCG1, CAV, SREBF1, SCARB1* and *APOA1*, as well as *CPT1A*, which regulates mitochondrial long-chain fatty acid oxidation. The lipid related genes *ABCA1* and *CYP7A1* were significantly downregulated by butyrate in HepG2 cells, 5 mM sodium butyrate also significantly upregulated *Hnf4a, Txnip, Abca1, Abcg1, Srebf1,* and insulin (*Ins*) mRNA, all of which have been implicated in β cell function and insulin secretion. The important β cell identity gene *Pdx1* was marginally upregulated (*p*=0.06). *LdIr* was downregulated.

Conversely, in HepG2 cells 10 μ M atorvastatin only significantly altered expression of genes implicated in T2D but primarily involved in lipid metabolism, namely the transcription factor *PPARG* and the cholesterol synthesis gene *HMGCR* were upregulated, and *SREBF1* was downregulated (Figure 5-7C). Furthermore, atorvastatin upregulated *Ldlr* expression and downregulated *Abca1* and *Abcg1* expression in BRIN-BD11 cells, while having no significant effect on the other genes examined in this cell type (Figure 5-7D).



Figure 5-7 Influence of butyrate and atorvastatin on diabetes-related gene expression

Reverse-transcription qPCR was conducted as described in section 3.6 for A) HepG2 cells and B) BRIN-BD11 cells treated with 5 mM sodium butyrate; relative gene expression in C) HepG2 cells and D) BRIN-BD11 cells treated with 10 μ M atorvastatin compared to the untreated control HepG2 cell gene expression is normalised to reference genes *GAPDH*, *YWHAZ* and *RPL13A*. BRIN-BD11 gene expression is normalised to reference genes *Actb* and *Rpl13a*. Atorvastatin data is shown relative to the untreated control, but statistical significance is compared to the DMSO vehicle control. Results represent the mean combined from three independent experiments, with error bars representing SEMs. * p<0.05 ** p<0.01, *** p<0.001, **** p<0.0001. Analysed by ANOVA.

5.5. Discussion

5.5.1. Effects of butyrate and statins on the expression of genes involved in diabetes and glucose metabolism

Sodium butyrate altered the expression of a significant number of genes that have been associated with glucose metabolism and T2D, while atorvastatin altered the expression of a smaller subset of genes. These are summarised in

Table 5.1. Of the 23 candidate genes analysed in HepG2 cells, butyrate significantly altered the mRNA expression of 11 (eight upregulated, three downregulated). In BRIN-BD11 cells, butyrate altered the mRNA expression of nine out of ten candidate genes (eight upregulated, one downregulated). Conversely, atorvastatin only altered the mRNA expression of three genes in HepG2 cells, increasing *HMGCR* and *PPARG* expression and decreasing *SREBF1* expression, and three genes in BRIN-BD11 cells, decreasing *Abcg1* and *Abca1* expression and increasing *Ldlr* expression, all of which have been implicated in T2D but are primarily involved in lipid metabolism. The disparity between butyrate and statins is likely due to their different mechanisms of action as discussed in Chapter 4 for cholesterol genes.

Also as discussed in Chapter 4, the effects on mRNA expression are not always linked to protein levels or phenotypic changes. For instance, many of the changes in gene expression by 5 mM sodium butyrate in BRIN-BD11 cells would be expected to increase β cell function and insulin secretion, however, the insulin secretion experiments indicate the opposite occurred.

Specific genes whose altered expression may contribute to observed effects on insulin secretion are discussed in the next two sections on HDAC inhibition and statins respectively.

Table 5.1	Influence of but	tyrate and	atorvastatin	on a	selection	of	genes
involved in glu	ucose metabolism	and T2D					

Gene	Butyrate		Atorv		Role in glucose metabolism and T2D
HepG2	qPCR	WB	qPCR	WB	
ABCA1	Ļ	Ļ	ns	Ļ	Cholesterol efflux, downregulation impairs insulin secretion. ²³⁴
LDLR	ns	Ļ	ns	1	LDL receptor, may play a role in the diabetogenic effects of statins, mutations decrease T2D risk. ²³⁵

Gene	Buty	rate	Atorv		Role in glucose metabolism and T2D
SCARB1	1	↓	ns	ns	HDL receptor, upregulated in T2D, ²³⁶
(SRB1)					polymorphisms alter T2D risk. ²³⁷
HK1	1	1	ns	ns	High-affinity hexokinase, alters
					glucose sensing, repressed in healthy
					β cells and hepatocytes. ⁴⁷
PPARG	ns	ns	↑	ns	Increases insulin sensitivity in
					peripheral cells, increases glucose sensing in β cells and henatocytes 2^{38}
ARCG1	↑	ns	ns	ns	Cholesterol transporter reduced
incor		113	115	115	expression in T2D. ²³⁹ downregulation
					impairs insulin secretion. ²³⁴
GK	\downarrow	ns	ns	ns	Glucose sensor in β cells and
					hepatocytes, mutations cause
					diabetes. ²⁴⁰
HMGCR	ns	ns	↑	ns	Cholesterol biosynthesis, mutations
	•				increase T2D risk. ²⁴¹
SREBFI	T T		↓↓		Transcription factor, downregulates
					gluconeogenesis and increases
					secretion. ²⁴³
APOA1	1	ns	ns	ns	Component of HDL, increases insulin
					secretion, ²⁴⁴ low levels increase T2D
					risk. ²⁴⁵
CYP7A1	\downarrow		ns		Bile acid sythesis, increased by
					glucose and insulin, overexpression
SICAA	↑	na	na	n 0	Chapter transporter in 6 cells and
SLC2A2		115	115	115	hepatocytes, important for GSIS. ²⁴⁷
CAV1	1	ns	ns	ns	Protein of plasma membrane
					caveolae which contain insulin
					receptors, important for insulin
					signalling. ²⁴⁸
CPT1A	↑		ns		Controls hepatic mitochondrial fatty
					acid β -oxidation, inhibitors decrease
BRIN-BD11	<u> </u>		I		Sideoneogenesis.
Ins1	1		ns		Insulin gene.
Abcal	1	↑		1	Cholesterol efflux reduced in
			*	*	T2D, ²⁵⁰ downregulation impairs
					insulin secretion. ²³⁴
	qPCR	WB	qPCR	WB	
Hk1	1	↑	ns	ns	High-affinity hexokinase, alters
					glucose sensing, repressed in healthy
D1 1	•				β cells and hepatocytes. ⁴⁷
Paxi	Γ		ns		p cell specific transcription factor,
					function. ²⁵¹

Gene	Butyrate		Atory		Role in glucose metabolism and T2D
Abcg1	Î		Ļ		Cholesterol transporter, reduced in T2D, ²³⁹ downregulation impairs insulin secretion. ²³⁴
Ldlr	Ļ	Ļ	Î	↑	LDL receptor, may play a role in the diabetogenic effects of statins, mutations decrease T2D risk. ²³⁵
Srebf1	Î		ns		Transcription factor, downregulates gluconeogenesis and increases glycogen synthesis, ²⁴² impairs insulin secretion. ²⁴³
Txnip	↑		ns		Inhibits thioredoxin which protects against oxidative stress, induced by glucose and promotes β cell apoptosis. ²⁵²
Hnf4a	1		ns		Transcription factor important in islet and liver function, mutations cause diabetes. ²⁵³

5.5.2. Effects of HDAC inhibition on insulin secretion

The effect of statins and altered cellular cholesterol on insulin secretion were previously studied in-depth by our group.⁷² Briefly, either reducing or increasing BRIN-BD11 cellular cholesterol content reduced stimulated insulin secretion. As butyrate and TSA both reduce cellular cholesterol (see section 4.4.2, note that the effect of TSA on the cellular cholesterol content of BRIN-BD11 cells was not examined), it is possible that the reduced insulin secretion observed here was also a result of reduced cellular cholesterol. Statins, however, did not impair chronic insulin secretion was greater with HDAC inhibitors than with statins despite comparable cholesterol lowering at the doses used. Thus, cholesterol lowering in and of itself does not account for the defects in insulin secretion seen with HDAC inhibition.

Potent inhibition of HDAC activity in BRIN-BD11 cells over 24 h with 5 mM sodium butyrate or 2.5 uM TSA significantly impaired both 24 h insulin secretion and secretion in response to acute (20 min) stimulation with well-established potent secretagogues. This is likely due to defects in both insulin production and secretion, since insulin content was decreased by 24 h HDAC inhibition, and the percentage of total insulin content secreted was also lower in cells subject to potent HDAC inhibition. It should be noted that insulin secretion was not increased, and in fact was decreased, during the treatment period, so the reduced content was not due to high
levels of insulin secretion depleting stores. As insulin mRNA was increased by 5 mM butyrate, it is unlikely that altered insulin gene expression contributed to the reduced cellular insulin content, although it is possible that expression was lowered at an earlier time point as seen with *HMGCR* in section 4.4.5. If not insulin gene expression, insulin content could be affected by influences on translation, processing or degradation. The insulin gene encodes preproinsulin, which is cleaved in the endoplasmic reticulum to form proinsulin, which is folded and transported to secretory vesicles where it is again cleaved to form mature insulin.²⁵⁴ The insulin ELISA used in this study only recognises mature insulin and thus would not detect changes to the precursors preproinsulin and proinsulin. Many proteins are involved in the translation and processing of insulin, with glucose sensing upregulating the process.

The activation of hexokinase I expression, observed both in this study and in studies of butyrate treated RINm5F insulin-secreting cells,^{182,183} may contribute to the detrimental effects of butyrate on both insulin content and GSIS. Furthermore, although protein levels of glucokinase, the 'glucose sensor' in β cells and hepatocytes,²⁵⁵ were not significantly altered by butyrate treatment, glucokinase mRNA was significantly downregulated in HepG2 cells. Hexokinase I is silenced in healthy β cells and hepatocytes as it phosphorylates glucose at low concentrations, whereas glucokinase has a low affinity for glucose, as thus is only active when glucose concentrations are high.⁴⁷ As a result, hexokinase I expression would lead to insulin secretion and glycogen synthesis when blood glucose is low, thereby causing hypoglycaemia. Furthermore, unlike glucokinase, other hexokinases are inhibited by their product, thereby limiting their activity in high glucose environments. Therefore, hexokinase I expression would result in inappropriate response to both high and low glucose (Figure 5-8). Feeding hamsters high sucrose diets has been found to increase islet hexokinase I expression, leading to hyperinsulinaemia,256 and hexokinase I expression was found to increase with age, obesity and diabetes development in rat islets.²⁵⁷ While hexokinase I expression has been associated with β cell dedifferentiation, 258 the increased expression of transcription factors relevant to β cell identity, such as Pdx1 and Hnf4a, as well as the insulin gene itself, suggest that more generalised dedifferentiation has not occurred.



Figure 5-8 Hexokinase and glucokinase activity

Exposing islets to high glucose (27 mM) for 48 h has been shown to impair insulin secretion,²⁵⁹ associated with an increase in the proinsulin:insulin ratio.²⁶⁰ Although cellular insulin content is decreased by glucose overstimulation, this only partially explains the impaired insulin secretion. It seems plausible that increased glucose sensing due to hexokinase activity could have similar effects, especially as an increase in the proinsulin:insulin ratio would explain the increased the discord between increased insulin mRNA but decreased insulin content seen in butyrate treated cells. This is comparable to the glucotoxicity thought to occur in T2D, where prolonged hyperglycaemia and subsequent demand for insulin leads to a decline in β cell function and an eventual increase in β cell apoptosis, possibly due to oxidative stress.²⁶¹

To explore the mechanism by which HDAC inhibitors increase hexokinase expression, ChIP experiments were conducted to examine histone acetylation in proximity to the *HK1* gene in HepG2 cells (see section 6.4.6). Using ENCODE data, an H3K9ac site in proximity to intron 1 of HK1 with low peak strength in HepG2 cells compared to other cell types was identified. ChIP experiments found no change in H3K9 or H4K8 acetylation in this region in butyrate treated HepG2 cells. Thus, the exact mechanism by which butyrate upregulates hexokinase I remains unknown. Interestingly, it has been shown that the repression of hexokinase I in healthy β cells is epigenetically controlled; Dhawan et al.²⁶² demonstrated increased DNA

methylation of the *Hk1* promoter in β cells from adult mice compared to immature β cells from neonatal mice, with increased binding of the DNA methyltransferase DNMT3A in this region, Histone modifications, namely decreased H3K9 acetylation and increased H3K27 methylation, have been found to play a role in the repression of two other genes disallowed in β cells, *Mct1* and *Ldha*.²⁶³ ChIP-seq across the *Hk1* locus in both β cells and hepatocytes could elucidate the mechanism by which HDAC inhibition activates hexokinase I expression.

Butyrate treatment also increased expression of thioredoxin interacting protein (*Txnip*) in BRIN-BD11 cells. TXNIP inhibits thioredoxin, a thiol oxidoreductase that reduces oxidised proteins produced by reactive oxygen species.¹⁰² The increase in *Txnip* mRNA thus provides further evidence that HDAC inhibition can increase oxidative stress in β cells; TSA has also been previously found to increase reactive oxygen species in insulin secreting cells.²⁶⁴ In β cells, TXNIP is strongly upregulated by glucose and induces apoptosis. As a result, TXNIP has been implicated as the link between chronic hyperglycaemia and β cell death.¹⁰² While the observed increase in *Txnip* mRNA could be due to increased glucose sensing as a result of hexokinase I expression, there is also evidence that *TXNIP* is epigenetically controlled. Glucose increases histone acetylation in proximity to the *TXNIP* gene and knockdown or inhibition of the histone acetyltransferase p300 ameliorates the glucose induced TXNIP upregulation.²⁶⁵ The increased *Txnip* expression could therefore be a result of HDAC inhibition and subsequent increase taxing expression.^{264,266-268}

The effect of chronic butyrate exposure on insulin secretion are summarised in Figure 5-9.



Figure 5-9 Summary of effects of chronic butyrate treatment on insulin secretion

24 h treatment with butyrate inhibits HDAC activity, which leads to both decreased insulin content and impaired insulin secretion. This may be due to the effects of HDAC inhibition on gene activity, namely upregulation of *HK1* and *TXNIP*, leading to increased glucose phosphorylation and increased oxidative stress respectively.

Although 24 h treatment with butyrate was detrimental to insulin secretion, acute exposure to butyrate enhanced the secretory response of BRIN-BD11 cells to alanine and glucose. In agreement with this, a historical study by Manns et al.¹⁷⁹ in 1967 reported that infusion of butyrate directly into the pancreatic artery of sheep resulted in a marked increase in plasma insulin within 3 min. More recently, Lin et al.²⁶⁹ found increased plasma insulin 10 min after orally administering sodium butyrate to mice. This acute secretion is unlikely to be due to changes in gene expression, i.e., due to HDAC inhibition, as short term (40 min) butyrate treatment prior to stimulation failed to significantly enhance insulin secretion in our study. Instead, butyrate may act as a secretagogue through FFAR activation or through direct metabolism of butyrate as an

energy source. Butyrate is the primary energy source for colonocytes,²⁷⁰ while, *in vitro*, butyric acid increased GSIS in MIN6 cells but not in FFAR2 knockdown cells, suggesting it may be the actions of butyrate as a FFAR agonist that can enhance insulin secretion.²⁷¹ Similar results with other SCFAs were reported by Pingitore et al.²⁷² and Priyadarshini et al.²⁷³ however, other studies have suggested that FFAR activation impairs insulin secretion.^{274,275} thus uncertainty remains regarding the role of FFARs in insulin secretion. It may be that the stimulatory effects of the 1 mM butyrate after 24 h are also a result of FFAR activation or butyrate metabolism, while the detrimental effects of HDAC inhibition are stronger at higher doses. Conversely, it may be that positive effects on gene expression, such as upregulation of *Pdx1*, *Hnfa* and insulin, have a greater impact on GSIS than the negative effects of upregulation of *Hk1* and *Txnip* at this lower dose.

Butyrate and TSA are known to have poor systemic availability and are rapidly metabolised.²⁷⁶⁻²⁷⁸ Thus, the treatment times and doses used in this, and other *in vitro*, studies are unlikely to reflect the *in vivo* effects of oral treatment. Most notably, rather than impairing insulin secretion and β cell function, in high fat diet (HFD) fed rodents, butyrate supplementation reduces fasting glucose, 118,211,279 decreases β cell hyperplasia,^{180,279} and reduces signs of inflammatory response in islets.²⁷⁹ In an *ex* vivo study of pancreatic islets from mice fed a HFD with or without butyrate supplementation, butyrate increased GSIS by increasing glucose-stimulated (16.7 mM) insulin secretion while reducing basal (2.8 mM glucose) insulin secretion.180 These in vivo and ex vivo effects may not be a direct effect of HDAC inhibition in the β cell, but rather the result of butyrate affecting other tissues. Butyrate stimulates the release of glucagon-like peptide 1 (GLP-1) from intestinal cells both in vitro and in vivo,^{280,281} as also observed in a human trial,⁶ and GLP-1 is an established enhancer of β cell function and insulin secretion.²⁸² It may also be that cells are exposed to low doses of butyrate for a short time period, and thus the low dose and acute butyrate treatments may be more reflective of *in vivo* effects than the supraphysiological 24 h treatments.

5.5.3. Cholesterol lowering by statins as a mechanism for impaired insulin secretion

The genes altered by atorvastatin treatment can all be linked to decreased cholesterol levels, as they are either activated by SREBP-2 (*LDLR*, *HMGCR*, *PPARG*),²⁸³ or silenced by LXR (*ABCA1*, *ABCG1*, *SREBF1*) when cellular cholesterol levels are

low.³⁹ There were no changes in the examined diabetes related genes that cannot be explained by cholesterol lowering. There are multiple lines of evidence that suggest it is cholesterol lowering rather than an off-target effect that accounts for the diabetogenic effects of statins. Lowering cellular cholesterol with methyl-βcyclodextrin, an oligosaccharide with a strong affinity for cholesterol that extracts cholesterol from cellular membranes,²⁸⁴ impairs both insulin secretion⁷² and signalling⁷¹ in vitro, and mutations that increase⁶⁹ or decrease⁷⁰ plasma LDL-C decrease and increase the risk of T2D respectively. Furthermore, numerous studies have shown that silencing of the cholesterol transporters ABCA1234,285-287 and ABCG1,^{64,234} both of which were downregulated by atorvastatin in BRIN-BD11 cells, impairs β cell function and insulin secretion. The defects seen in the ABCA1 deficient cells were attributed to lack of cholesterol efflux leading to intracellular cholesterol accumulation; depleting the ABCA1 deficient cells of cholesterol corrected the impairment.²⁸⁶ ABCA1 downregulation is thus unlikely to be the cause of impaired insulin secretion in statin treated BRIN-BD11 cells these cells had depleted cellular cholesterol (see section 4.4.2). Conversely, ABCG1 deficiency was found to not alter total cellular cholesterol nor cholesterol efflux, and addition of cholesterol to deficient cells rescued insulin secretion.⁶⁴ It was found that ABCG1 localised to insulin secretory granules and that ABGC1 deficient cells had reduced granule membrane cholesterol and altered granule morphology, namely an increase in cross-sectional area. The membranes of insulin secretory granules are approximately 35% cholesterol, similar to the plasma membrane,²⁸⁸ and depleting secretory vesicles of cholesterol, by the use of M β CD, has been shown to reduce membrane fusion.²⁸⁹ This may be because SNARE proteins, which facilitate the fusion of secretory vesicles with the plasma membrane, are found in cholesterol dependant clusters; cholesterol depletion by MBCD causes dispersion of these clusters and reduced exocytosis.²⁹⁰ There is also evidence that ABCG1 deficiency and reduced insulin secretory granule cholesterol increases the lysosomal degradation of granules.²⁸⁸ Reduced ABCG1 expression, and subsequent reduced granule cholesterol, may therefore contribute to the impairments in insulin secretion seen in statin treated cells and thus may contribute to the diabetogenic effects of statins.

Interestingly, DNA hypermethylation of *ABCG1* in blood cells has been associated with T2D,⁹³⁻⁹⁷ with low levels of HDL-C,⁸⁹⁻⁹² and with statin use.¹⁵⁴⁻¹⁵⁶ As DNA methylation was not examined in this thesis it cannot be ruled out that epigenetics contributes to the reduced *Abcg1* expression in statin treated BRIN-BD11 cells.

Similarly to histone modifications induced by SREBP-2 (see section 6.5.3), statins may induce epigenetic changes indirectly through LXR activity. Sandoval-Hernández et al.²⁹¹ found that the LXR agonist GW3965 reduced DNA methylation at numerous LXR target genes in mice, although they did examine effects on *ABCA1* or *ABCG1*. Similarly, Tristán-Flores et al.²⁹² found a differentially methylated region in the *Alu* retrotransposon in atherosclerotic plaques that contained an LXR binding site was hypermethylated by GW3965. Whether LXR activity alters DNA methylation of *ABCG1* does not appear to have been examined and would be an interesting area of further research.

There is also evidence that statins may alter *ABCG1* expression through histone modifications. In adipocytes, statins reduced *ABCG1* expression while increasing levels of HDAC9,²⁹³ and HDAC9 deficient macrophages have increased *ABCG1* expression and H3K9 acetylation.²⁹⁴ Figure 6-10 showed a trend towards decreased H3K9 acetylation in proximity to *ABCG1* in atorvastatin treated HepG2 cells but this was not significant. It should be noted that this was in HepG2 cells, where *ABCG1* expression was not significantly altered by atorvastatin, whereas ChIP was not performed in BRIN-BD11 cells due to budgetary constraints.

5.5.4. Effects of butyrate and statins on insulin signalling

Peripheral insulin resistance typically precedes impaired insulin secretion in the development of T2D. Some clinical trials have suggested a worsening of insulin sensitivity with statin treatment, 295,296 while other trials found either no effect 297,298 or an improvement in insulin sensitivity.^{299,300} A previous study by our group found either increasing or decreasing cellular cholesterol using cyclodextrins decreased the binding of insulin to the insulin receptor in a number of cell lines, most likely due to changes in plasma membrane cholesterol.⁷¹ Furthermore, reducing cellular cholesterol with cyclodextrins reduced the phosphorylation of AKT and IR β in response to insulin. Atorvastatin, however, did not have the same effects on insulin binding or signalling. In fact, in HepG2 cells, but not the Chinese hamster ovary CHO cell line or differentiated myotubes, atorvastatin increased the phosphorylation of AKT and IRβ in response to insulin.⁷¹ Conversely, results shown in this chapter did not find a significant effect of atorvastatin on insulin signalling, possibly due to a shorter treatment time (24 h v 48 h), although it did find a trend towards increased insulin-stimulated IRβ and IRS-1 phosphorylation but decreased AKT

phosphorylation. It should be noted AKT phosphorylation is not unique to insulin signalling and is involved in numerous other pathways involved in cell growth, metabolism and survival.³⁰¹ Regardless, the effects on statins on insulin signalling remain unclear.

The HDAC inhibitor sodium butyrate decreased AKT phosphorylation while tending to increase non-stimulated IRS-1 phosphorylation. This could be due to effects on non-histone lysine acetylation. In addition to impacting gene expression by reducing histone acetylation, HDACs can also remove acetyl marks from numerous other proteins, including those in insulin signalling pathways. In particular, IRS-1 has been found to associate with HDAC2 and HDAC inhibition with TSA increases lysine acetylation of IRS-1 and subsequently increases IRS-1 tyrosine phosphorylation.³⁰² AKT is also acetylated at lysine residues by HATs and deacetylated by the HDAC SIRT1.³⁰³ Unlike IRS-1, acetylation of AKT reduces its ability to be phosphorylated, consistent with the results presented here. Multiple other proteins involved in insulin signalling have been shown to be acetylated, with acetylation again acting both to increase (Rictor and TRB3) and decrease (IRS-2 and PDK1) insulin signalling.³⁰⁴ As a result, it is difficult to summarise the overall effect of HDAC inhibition on insulin sensitivity.

It is also unclear how HDAC inhibition affects insulin signalling *in vivo*. Zhang et al.³⁰⁵ reported that sodium butyrate supplementation reduced liver AKT phosphorylation while increasing GSK3 α and GSK3 β phosphorylation in db/db mice. Increased IRS-1 phosphorylation has been reported in the liver¹⁸⁴ and skeletal muscle¹⁶⁵ of butyrate treated rodents. Increased AKT phosphorylation has also been reported in liver¹¹⁸ and skeletal muscle¹⁸⁴ of butyrate treated rodents. Similarly to insulin secretion, *in vivo* effects on insulin signalling may not be due to direct effects on HDAC inhibitors on the genes and proteins involved in the insulin signalling pathway but may be downstream effects of changes to genes and proteins in other metabolic pathways, such as enhanced insulin secretion due to GLP-1 secretion.

5.6. Conclusion

Statin use has been linked to an increased risk of T2D, while HDAC inhibitors have been suggested to have positive effects on glucose metabolism in animal models of diabetes. The results presented here suggest that *in vitro*, statins and HDAC inhibitors both impair insulin secretion and may interfere with insulin signalling. The effects of statins are most likely secondary to cholesterol lowering, as cholesterol depletion through other means has similar effects. In particular, the decrease in expression of ABCG1 in response to lowered cholesterol may impair insulin secretion, possibly by reducing the cholesterol content of insulin secretory granules. HDAC inhibition may impair β cell function and insulin secretion through activation of hexokinase I, causing cells to respond to low glucose and leading to β cell exhaustion, and TXNIP, increasing oxidative stress. Conversely, low concentrations of butyrate may increase stimulated insulin secretion and butyrate may directly act as a secretagogue to acutely increase secretion. Butyrate may interfere with insulin signalling by increasing acetylation of proteins involved in the insulin signalling cascade, which can either increase (IRS-1) or decrease (AKT) their ability to be phosphorylated. As the uptake of HDAC inhibitors such as butyrate into the systemic circulation is poor, these *in vitro* effects may not reflect *in vivo* effects, which may be more reflective of effects on intestinal cells, in particular increased secretion of GLP-1.

Chapter 6. Epigenetics

6.1. Abstract

Epigenetic modifications have been linked to dyslipidaemia, CVD and T2D. Statins have also been reported to have epigenetic effects, in particular inhibition of HDACs, however this has rarely been investigated in the context of diabetes or cholesterol lowering. HDAC inhibitors such as the SCFA butyrate have been reported to reduce cholesterol and diabetes risk in animal models, however the majority of studies do not report epigenetic modifications at target genes that may be involved. This chapter investigates the effect of statin drugs and established epigenetic modifiers on epigenetic modifications in multiple cell lines, including hepatic cells, insulinsecreting cells, macrophages and breast carcinoma cells, as well as liver extracts from statin-treated C57B1/6J mice. Cells or cell extracts were treated with statins and with established epigenetic modulators, and HDAC, HAT, and DNMT activities were quantified. Histone acetylation was assessed on a global level by immunoblotting and at specific genomic locations by ChIP-qPCR. Statins altered neither HDAC nor HAT activity. Accordingly, global acetylation of histones H3 and H4 was unchanged with statin treatment. Atorvastatin did tend to increase DNMT activity. Butyrate significantly inhibited HDAC activity and increased global histone acetylation in all experiments. Despite the lack of effect on HDACs, HATs and global histone acetylation, statins tended to increase histone acetylation in proximity to SREBP-2 target sites, significantly for HMGCR. The effects of butyrate on histone modifications at targeted genomic regions are unclear, as ChIP results were significantly impacted by the data normalisation method used. Increased H4K8 acetylation in proximity to the gene encoding the glucose transporter GLUT2 was consistent across all methods.

6.2. Objectives

3. Investigate how statins and established epigenetic modifiers influence epigenetic modifying enzymes and alter genome-wide epigenetics.

4b. Investigate if epigenetic modifications contribute to the altered expression of target genes.

6.3. Introduction

There has been considerable research linking epigenetics to dyslipidaemia, ^{15,16} insulin resistance and T2D,¹⁷⁻²⁰ and CVD.²¹⁻²³ As a result, there is interest in the use of epigenetic modifiers, in particular histone deacetylase (HDAC) inhibitors, to prevent and treat these disorders.^{24,25} Butyrate, an HDAC inhibitor produced by the gut microbiome, has shown promising results in pre-clinical experiments and has reached clinical trials for lipid lowering and T2D.^{6,26,27} Statins, the widely used cholesterol lowering drugs, also have reported epigenetic effects, as reviewed in Allen and Mamotte²⁸ with the majority of studies conducted in the context of cancer. The effect of statins on epigenetic enzyme activity and global histone modifications has not been investigated in the context of dyslipidaemia, the primary indication for statin use, or type 2 diabetes, which has been linked to statin use.¹⁴ This chapter investigates the effects of statins and sodium butyrate on global and local histone modifications and on the activity of the major classes of enzymes that enact epigenetic changes in cell types implicated in the pathology of CVD and T2D, namely HepG2 hepatic cells, BRIN-BD11 insulin-secreting cells and THP-1 macrophages, with confirmatory experiments conducted in MDA-MB-231 breast cancer cells to compare to previous studies conducted in cancer cell lines.

6.4. Results

6.4.1. Statins do not inhibit HDAC activity

Sodium butyrate inhibited HDAC activity in live HepG2 cells with an IC₅₀ of 445 μ M (± 126 μ M SE) (Figure 6-1A). Conversely, after 24 h treatment, no statin (at 10 μ M) inhibited HDAC activity in live HepG2 cells (Figure 6-1B). The potent HDAC inhibitor TSA also significantly inhibited HDAC activity (*p*<0.0001). Atorvastatin, chosen as a model statin due to its potency and widespread use, was tested at a range of concentrations and did not inhibit HDAC activity at up to and including 50 μ M (Figure 6-2), a concentration sufficiently high to impact cell viability (see section 4.4.1), and more than sufficient to reduce cellular cholesterol content (see section 4.4.2).



Figure 6-1 Influence of statins and butyrate on HDAC activity in HepG2 cells

HDAC activity was determined in live HepG2 cells as determined using the methods described in section 3.9.1 following 24 h treatment with A) varying concentrations of sodium butyrate and B) 10 μ M statins, 5 mM sodium butyrate or 10 μ M TSA. HDAC activity was also determined following direct application of C) varying concentrations of sodium butyrate and D) 100 μ M statins or 10 μ M TSA to HepG2 nuclear extracts. Results represent the mean combined from at least three independent experiments, with error bars representing SEMs. **** p<0.0001 compared to control. Analysed by non-linear regression (A and C) and ANOVA (B and D).



Figure 6-2 Influence of atorvastatin on HDAC activity in HepG2 cells

HDAC activity was determined in live HepG2 cells following 24 h treatment with varying concentrations of atorvastatin. Results represent the mean combined from three independent experiments, with error bars representing SEMs. Analysed by ANOVA.

As statins have been reported to directly inhibit HDAC activity in A549 lung carcinoma nuclear extracts,¹⁴⁵ we examined the direct effect of statins and sodium butyrate on HDAC activity in HepG2 nuclear extracts. Sodium butyrate inhibited HDAC activity with an IC₅₀ of 104 μ M (± 21 μ M SE) (Figure 6-1C). Conversely, as in live cells, atorvastatin and pravastatin did not inhibit HDAC activity (Figure 6-1D). This was done with very high statin concentrations, 100 μ M, a dose significantly higher than the 30 μ M used by Lin et al.¹⁴⁵

Furthermore, as previous studies of HDAC inhibition by statins were conducted primarily in the context of cancer, experiments were conducted using MDA-MB-231 breast carcinoma cells and found HDAC inhibition by sodium butyrate (IC₅₀ of 403 μ M) (Figure 6-3A). Experiments on live cells (Figure 6-3B) and on nuclear extracts showed no effect by statins (Figure 6-3C). Statins also failed to inhibit HDAC activity in live BRIN-BD11 insulin secreting cells, while sodium butyrate and TSA showed significant inhibition (Figure 6-3D).



Figure 6-3 Influence of statins and butyrate on HDAC activity in MDA-MB-231 cells and BRIN-BD11 cells

HDAC activity was determined in A) live MDA-MB-231 cells following 24 h treatment with 10 μ M statins or 5 mM sodium butyrate, B) live MDA-MB-231 cells following 24 h treatment with varying of sodium butyrate; C) MDA-MB-231 nuclear extracts exposed to varying concentrations of atorvastatin; D) BRIN-BD11 cells following 24 h treatment with 10 μ M statins or trichostatin A, or 5 mM sodium butyrate. Results represent the mean combined from three independent experiments, with error bars representing SEMs. **** p<0.0001 compared to control. Analysed by ANOVA.

Additionally, experiments were conducted using the Fluor De Lys® HDAC fluorometric activity assay kit used in previous studies,^{145,306,307} which again showed

HDAC inhibition with TSA and sodium butyrate but not statins in both HepG2 nuclear extracts and the HeLa nuclear extract provided with the kit (Figure 6-4).



Figure 6-4 Influence of statins and HDAC inhibitors on HDAC activity using the Fluor De Lys HDAC activity kit

HDAC activity of HepG2 and HeLa nuclear extracts treated with 200 μ M statins, 1 mM sodium butyrate or 4 μ M TSA was determined using the Fluor De Lys HDAC activity kit (Enzo Life Sciences) as outlined in section 3.9.1. Results represent the mean combined from three independent experiments, with error bars representing SEMs. ****p<0.0001 compared to control. Analysed by ANOVA.

Finally, liver samples were utilised from a previous animal study in which male C57B1/6J mice were fed normal chow or high fat diet and treated with atorvastatin or water (vehicle control) for 12 weeks. HDAC activity was measured in nuclear extracts isolated from the liver of control animals fed a normal or HFD and from atorvastatin animals fed a normal or HFD. Neither diet nor treatment had a statistically significant effect on HDAC activity (Figure 6-5).



Figure 6-5 Influence of statin and diet on HDAC activity in mice livers Mice were fed a normal diet (ND) or high fat diet (HFD) and treated with a torvastatin (A) or water (V) for 12 weeks (n=3 per group) as outlined in section 3.2.2 and HDAC activity was measured as described in section 3.9.1, with error bars representing SEMs. Analysed by ANOVA.

6.4.2. Statins do not inhibit HAT activity

In addition to HDAC inhibition, histone acetylation levels can be altered through influences on HATs. Curcumin was used as a control inhibitor and strongly inhibited the HAT activity of HepG2 nuclear extracts; 100 μ M curcumin inhibited HAT activity by 70% (± 6% SE) (Figure 6-6). Conversely, high dose (100 μ M) atorvastatin had no statistically significant effect on HAT activity.



Figure 6-6 Influence of atorvastatin on HAT activity

 $100 \,\mu$ M curcumin, a torvastatin or DMSO vehicle control was a pplied to HepG2 nuclear extracts obtained using the method described in section 3.2.3, and HAT activity was measured using the

method described in section 3.9.2. Results represent the mean combined from three independent experiments, with error bars representing SEMs. * p<0.05. Analysed by ANOVA.

6.4.3. Sodium butyrate, but not statins, increase global histone acetylation

Global levels of histone modifications were determined by immunoblotting using antibodies against modified histones, namely against acetylated or trimethylated H3K9 and multiple acetylated residues of H4. H3K9 and H4 acetylation are commonly associated with open chromatin and gene expression,^{308,309} while H3K9 trimethylation is associated with compact chromatin.³¹⁰ Protein levels of HDAC3 were also examined as this been implicated in the development of diabetes.³¹¹ H3K9 and H4 acetylation was increased in a dose dependant manner by sodium butyrate treatment (Figure 6-7A, B). Conversely, acetylation levels were unchanged in cells treated with atorvastatin or curcumin. No treatment had significant effects on H3K9 trimethylation or on the levels of HDAC3 in HepG2 cells (Figure 6-7C, D). Similar results were seen in BRIN-BD11 cells (Supplementary Figure S4).



Figure 6-7 Influence of butyrate, atorvastatin and curcumin on global histone modifications in HepG2 cells

Relative protein levels of A) a cetylated H3K9, B) a cetylated H4, C) trimethylated H3K9 and D) HDAC3. Immunoblots obtained using the method described in section 3.5, using cell extracts obtained using RIPA buffer as described in section 3.2.4. The graph represents the mean combined density readings as normalised to GAPDH from at least three independent experiments, with error bars representing SEMs. * p<0.05, ***p<0.001. Analysed by ANOVA.

The bands for the acetylated histones were faint in control HepG2 and BRIN-BD11 cells. The untreated bands were denser from THP-1 macrophages, and this was further increased by sodium butyrate treatment (Figure 6-8). Atorvastatin did not alter histone acetylation in THP-1 macrophages.

In order to determine that the lack of lipoproteins in the serum did not affect results, H3K9 acetylation was also determined in cells treated in media containing high lipoprotein serum. As in LPDS, sodium butyrate significantly increased acetylation above baseline levels, whereas atorvastatin had no effect (Supplementary Figure S5).



Figure 6-8 Influence of butyrate and atorvastatin on global histone modifications in THP-1 cells

Relative protein levels of A) acetylated H3K9 and B) acetylated H4. The graph represents the mean combined density readings as normalised to GAPDH from three independent experiments, with error bars representing SEMs. ***p<0.001, ****p<0.0001. Analysed by ANOVA.

6.4.4. Statins do not directly inhibit DNMT activity

Statins have also been reported to inhibit DNMT activity. When applied directly to whole cell extracts, the control DNMT inhibitor curcumin inhibited DNMT activity by 65% (\pm 8% SE) compared to the control (DMSO vehicle) (Figure 6-9A). This was significantly greater than an equal dose of the rationally designed DNMT inhibitor

RG108,³¹² which inhibited DNMT activity by 25% (\pm 9% SE) and was nominally significant (*p*=0.057). 200 µM atorvastatin did not significantly inhibit, and in fact tended to increase, DNMT activity (*p*=0.064). Similar results were obtained with BRIN-BD11 cell extracts (Supplementary Figure S6A).

In a separate series of experiments, cells were treated for 24 h with atorvastatin or curcumin prior to extraction and subsequent measurement of DNMT activity. Activity was variable in these experiments but DNMT activity in atorvastatin cells was in fact higher than the DMSO vehicle control, although it did not differ significantly from untreated control cells (p=0.3) (Figure 6-9B). DNMT activity of cells treated with curcumin did not differ from those treated with DMSO. In BRIN-BD11 cells, there were no significant differences in DNMT activity between untreated, DMSO treated, atorvastatin treated and curcumin treated cells (Supplementary Figure S6B).



extracts

A) DNMT activity in whole cell extracts exposed to $200 \,\mu$ M atorvastatin, RG108, curcumin or DMSO vehicle control; B) DNMT activity in whole cell extracts from cells treated for 24 h with $10 \,\mu$ M atorvastatin, curcumin or DMSO vehicle control. DNMT activity was measured using the method described in section 3.9.3, cell extracts were obtained using RIPA buffer as outlined in section 3.2.4. Results represent the mean combined from three independent experiments, with error bars representing SEMs. * p < 0.05 * * * p < 0.001. Analysed by ANOVA.

6.4.5. Statins alter histone acetylation in specific genomic regions using ChIP-qPCR

Chromatin immunoprecipitation using antibodies against modified histones, namely acetylated and methylated H3K9 and acetylated H4K8, followed by qPCR was used to determine the effect of treatment on histone modifications in proximity to selected target genes. Data were normalised using the percentage of input method, where the qPCR signals are divided by signals from a portion of chromatin put aside before immunoprecipitation.³¹³ Treatment of HepG2 cells with 10 uM atorvastatin for 24 h significantly increased histone H4K8 acetylation upstream of the HMGCR transcription start site in proximity to a SREBP binding site, while non-significantly increasing H3K9 acetylation in this region (p=0.21) (Figure 6-10A). A similar pattern, though not statistically significant, was observed in proximity to fellow SREBP-2 targets, LDLR (p=0.10) (Figure 6-10B) and PPARG (p=0.21) (Figure 6-10C). When results from the three SREBP-2 target sites were combined, there was a significant increase in H4K8 acetylation (p=0.001) (Figure 6-10 Influence of atorvastatin on histone acetylation in proximity to specific genomic regionsF). There were no significant differences observed at the control gene GAPDH (Figure 6-10D), or in proximity to ABCG1 (Figure 6-10E), a non-SREBP-2 target whose expression is decreased when cholesterol levels are low through LXR signalling. In these experiments, FBS was used instead of LPDS in both treated and control cells to reduce SREBP-2 signalling in control cells.



Figure 6-10 Influence of atorvastatin on histone acetylation in proximity to specific genomic regions

ChIP qPCR using the method described in section 3.10 from HepG2 cells treated with 10 μ M atorvastatin or DMSO vehicle control for 24 h in media containing FBS in proximity to A) *HMGCR*, B) *LDLR*, C) *PPARG*, D) *GAPDH* or E) *ABCG1*, or F) the combined results from the SREBP-2 target sites in proximity to *HMGCR*, *LDLR* and *PPARG*. Results represent the mean combined from three independent experiments, with error bars representing SEMs. * *p*<0.05 Analysed by Student's *t* test.

There is no agreed upon method to normalise ChIP-qPCR data, although percentage of input DNA and fold enrichment are the most commonly cited.³¹³ Fold enrichment involves dividing signals by the background (beads only) signal, however the

background in these experiments was so low to be negligible. Other normalisation methods include relative to nucleosome density (total H3) or to a positive control sequence.³¹³ When normalised to either total H3 signal or to the positive control region of GAPDH (see section 6.4.6), the increase in H4K8 acetylation in proximity to *HMGCR* was still observed but no longer significant (p=0.29 and p=0.15 respectively), possibly due to increased standard error found using these methods (Supplementary Figure S7).

6.4.6. Butyrate increases ChIP signals and H4 acetylation at numerous genomic regions

Butyrate treated cells had higher histone H3 signals as a percentage of input DNA compared to control cells at all genomic regions examined (Figure 6-11A), including genes whose expression is upregulated, downregulated and unchanged by butyrate treatment (see section 5.4.5). This effect was not observed with atorvastatin treatment, where there were no significant differences between treated cells and DMSO control cells at any genomic region (Supplementary Figure S8) and was consistent across four independent experiments so is unlikely to be a result of technical error. As a result, signals for modified histones were similarly increased in butyrate treated cells when results from all genomic regions examined were combined (Figure 6-11B), a pattern that was also observed for each individual gene (Supplementary Figure S9). Similar results were observed with the raw data (Supplementary Figure S10).



Figure 6-11Influence of butyrate on histone signals in ChIP-qPCRChIP qPCR from HepG2 cells treated with 5 mM butyrate for 24 h in media containing LPDS. A)ChIP qPCR H3 signals and B) combined ChIP qPCR signals from all genomic regions examined.

Results represent the mean combined from four independent experiments, with error bars representing SEMs. * p < 0.05, ** p < 0.01, **** p < 0.001. Analysed by Student's *t* test.

When signals were presented as a modified histone: total H3 ratio, only H4K8 acetylation remained elevated in butyrate treated cells when results from all genomic regions examined were combined (Figure 6-12). For individual genes, this increase in H4K8 acetylation was significant in proximity to SLC2A2 (GLUT2) (Figure 6-13A) and HMGCR (Figure 6-13B). There were no statistically significant changes to histone modifications at any other genomic region when analysed in this manner (Supplementary Figure S11). The overall increase in H4K8 acetylation could be a real effect, as butyrate does increase global H4 acetylation as shown in section 6.4.3, but it could also be due to altered binding efficiency. Data were therefore also normalised to the average result from all regions examined. When analysed in this manner, the increase in H4K8 acetylation in proximity to SLC2A2 (Figure 6-13C), but not HMGCR (Figure 6-13D), remained significant. There were no other statistically significant changes to histone modifications at any other genomic region when analysed in this manner (Supplementary Figure S12). Another method of normalisation is to use the signals of a positive control gene which is known to have high levels of histone acetylation.³¹³ A region of exon 2 of GAPDH was used for this purpose. When normalised to GAPDH, SLC2A2 H4K8 acetylation was increased approximately three-fold in butyrate treated cells, however this was not significant due to interexperiment variability in the strength of the difference (p=0.09)(Supplementary Figure S13). No other histone modification at any genomic region examined had borderline significance (p < 0.2). An increase in H4K8 acetylation could provide a mechanism for the increased SLC2A2 expression observed in butyrate treated cells (see section 5.4.5).



Figure 6-12 Influence of butyrate on histone signals in ChIP-qPCR relative to total H3

ChIP qPCR from HepG2 cells treated with 5 mM butyrate for 24 h in media containing LPDS presented as a modified histone: total H3 ratio of combined ChIP qPCR signals from all genomic regions examined. Results represent the mean combined from four independent experiments, with error bars representing SEMs. **** p < 0.0001. Analysed by Student's *t* test.





ChIP qPCR from HepG2 cells treated with 5 mM butyrate for 24 h in media containing LPDS presented as a modified histone: total H3 ratio of combined ChIP qPCR signals in proximity to A) SLC2A2 (GLUT2) and B) HMGCR, and as normalised to average results from all genomic regions examined, in proximity to C) SLC2A2 (GLUT2) and D) HMGCR. Results represent the mean combined from four independent experiments, with error bars representing SEMs. * p<0.05, ** p<0.01. Analysed by Student's t test.

6.5. Discussion

6.5.1. Sodium butyrate and curcumin inhibit epigenetic modifying enzymes

These results confirm the inhibitory effects of sodium butyrate and TSA on HDAC activity, and of curcumin on HAT and DNMT activity. Sodium butyrate is a well-established HDAC inhibitor with a reported IC₅₀ values ranging from 90 μ M in HT-29 nuclear extracts to 1.13 mM in live HeLa cells.¹⁵⁸ The results presented here fit into this range, with IC50s of ~100 μ M in nuclear extracts and ~400 μ M in live cells,

with the greater IC50 in live cells postulated to be a result of the cellular metabolism of butyrate over the 24 h treatment period.

Curcumin, the bright yellow compound in the spice turmeric, has purported benefits in a wide variety of disorders.³¹⁴ Research suggests that curcumin inhibits a variety of epigenetic modifying enzymes, including the HAT p300, HDACs 1, 4, 5, 6 and 8, and DNMTs 1, 3a and 3b.¹ Results confirmed a strong inhibitory effects on HAT and DNMT activity in cell extracts, but no significant change in DNMT activity in cells that had been treated with curcumin for 24 h. Furthermore, curcumin treatment did not result in a significant change in the acetylation of residues on histones H3 and H4 after 24 h. Curcumin is highly unstable in solution; Wang et al.² found 90% of curcumin degrades within 30 min in serum free media. This instability, combined with the poor bioavailability of curcumin in human trials; oral intake of 10 g or more results in serum concentrations of 50 ng/mL,³¹⁵ leads to uncertainty over the epigenetic effects of curcumin *in vivo*. Thus, curcumin was deemed unsuitable for use in experiments on cholesterol and glucose metabolism in cells.

6.5.2. Statins do not directly inhibit epigenetic modifying enzymes

As opposed to butyrate and curcumin, and contrary to the reported literature, statins failed to inhibit epigenetic enzymatic activity. Studies of various cell lines, 145-148 and a rabbit model of atherosclerosis¹⁴⁹ reported that statin treatment can cause hyperacetylation of histones H3 and/or H4. The literature suggests this may be due to inhibitory effects of statins on HDAC activity^{306,307} and expression.¹⁴⁶ The results presented here on HDAC activity in live cells differ from those reported in the literature. This was demonstrated on a variety of different cells lines, namely HepG2, MDA-MB-231 and BRIN-BD11 cells, in addition to livers of mice treated with statins for 12 weeks, leading us to conclude that neither acute nor chronic statin treatment alters cellular HDAC activity. After preliminary experiments with live Hep-G2 and BRIN-BD11 cells failed to find HDAC inhibition, we hypothesised that in live cells, the stating may be unable to enter the nucleus, or some other element of cellular activity prevents them from interacting with HDAC proteins, as studies by other groups were conducted either using recombinant HDAC proteins³⁰⁶ or cell extracts.^{145,307} As a result, we conducted experiments using nuclear extracts, which again failed to show HDAC inhibition by statins. The negative results of these experiments are unlikely to be due to an inadequate dose as concentrations of up to

500 µM statins were used in the nuclear extract experiments. This concentration is significantly higher than the 10 µM doses reported to inhibit HDAC activity by Lin et al.145 and well above the 1-15 nM serum concentration of statin users.316 Furthermore, cholesterol assays (see section 4.4.2) showed that the statins did lower cellular cholesterol, indicating that the statins are able to enter the cells and inhibit HMG-CoA reductase. The difference could also be due to the different cell types (A549 lung carcinoma cells and primary glomerular mesangial cells) and HDAC substrates used (all reported studies used the Fluor-de-Lys[™] HDAC Activity Assay Kit [Enzo Life Sciences]). Because the majority of studies reporting epigenetic effects have been conducted in the context of cancer, we used the aggressive breast cancer cell line MDA-MB-231 and found no effect in either live cells or nuclear extracts. We also obtained the Fluor-de-Lys[™] assay kit and found no effect with statins on HDAC activity in HepG2 or HeLa nuclear extracts. The accuracy of using fluorogenic substrates to measure HDAC activity has been questioned and the Fluor-de-Lys[™] substrate in particular tends to overestimate inhibition of HDAC1 and HDAC6 while failing to detect inhibition of HDAC8.³¹⁷ Similar experiments have not been reported for the cell permeable substrate used in the Sigma-Aldrich assay kit.

One potential contributor to the lack of an effect is the form of statins used. In the body, statins are present in both an active acid form and in a lactone form; in plasma, about 50% of atorvastatin and simvastatin are present in lactone form, whereas less than 10% of pravastatin, rosuvastatin and fluvastatin are present as lactones at any time.^{318,319} Lactone statins are 1000-2000 times more lipophilic than their acid counterpoints and may be more important in side effects including muscle toxicity³¹⁸ and drug interactions.³²⁰ The study by Lin et al.¹⁴⁵ did not state which form of statin they used. Acid salt statins were used in the experiments presented here as they are the active form, and of the statins used, only simvastatin is administered in the lactone form. Interestingly, in PBS and DMEM, used here in the nuclear extract and cell-based experiments respectively, it has been shown that lactone statins are largely converted to the acid form but not vice-versa.³²¹ The interconversion of statins by HepG2 cells does not appear to have been reported.

The lack of HDAC inhibition *in vitro* is supported by the immunoassay results showing no increase in acetylation of histone H3K9 or multiple residues of histone H4. In fact, despite reporting that lovastatin and atorvastatin directly inhibited the activity of recombinant HDAC1, HDAC2 and HDAC6 proteins with IC₅₀ values less

than 30 μ M, Chen et al.³⁰⁶ failed to show increased histone H3 acetylation following 24 h treatment with 30-50 μ M lovastatin, although Lin et al.¹⁴⁵ did report increased H3 acetylation following 16 h of treatment with 10 μ M statins (lovastatin, simvastatin, pravastatin, fluvastatin and atorvastatin). Furthermore, although Lüthje et al.¹⁴⁸ reported increased H3 acetylation in telomerase-immortalised uroepithelial cells (TERT-NHUC) treated with 1 nM simvastatin for 24 h, this effect was marginal and there was no effect at 10 nM, nor with 1-10 nM atorvastatin or in simvastatin-treated HEKa keratinocytes.

Similarly, the results presented here failed to find an inhibitory effect of atorvastatin on DNMT activity, either in cells that had been treated for 24 h or when applied directly to cell extracts; in contrast, the opposite was found with an increase in DNMT activity in extracts from HepG2 cells treated with statins for 24 h. Conversely, Kodach et al.¹⁵² reported reduced DNMT activity in extracts from lovastatin treated HCT116 colon cancer cells. Several differences in experimental design between our study and theirs may possibly account for the disparate results. Firstly, they used lovastatin, which we chose not to use. Instead, we chose to use atorvastatin as our exemplar statin as it was the most commonly prescribed drug in Australia.⁵ Additionally, they used a different cell type and a longer treatment time, 48 h versus 24 h. As they did not investigate if lovastatin directly inhibited DNMT activity, it cannot be ruled out that the observed decrease in activity is a result of changed phenotype, as they did observe increased differentiation and reduced 'stemness' of the cancer cells. The differences between HepG2 hepatoma cells and HCT116 colon cancer cells may therefore account for the different results. Finally, different commercial kits were used, although both relied on similar ELISA type principles. Unfortunately, there is no current protocol to measure DNMT activity in live cells as there is for HDAC activity, a limitation of both studies as how enzymes behave in cell-free extracts may not reflect their actual activity in live cells.

The effect of statins on HAT activity has not been previously reported. As with HDAC and DNMT activity, results presented here failed to find a direct effect of atorvastatin on HAT activity in HepG2 cell extracts. Thus, the results indicate that statins are unlikely to directly inhibit the three major classes of epigenetic modifying enzymes.

6.5.3. Statins may impact epigenetic modifications in specific genomic regions

Despite having no direct effect on epigenetic modifying enzymes nor increasing global histone H3K9 or H4 acetylation, atorvastatin significantly increased histone H4K8 acetylation in proximity to the SREBP-2 binding site of the HMGCR gene. This is most likely not due to a direct epigenetic effect of atorvastatin, but an indirect effect of cholesterol lowering and subsequent SREBP-2 activation, as similar, though non-significant, effects were found in proximity to the SREBP-2 binding sites of LDLR and PPARG, while no effects were noted in proximity to non-SREBP targets GAPDH and ABCG1. Specifically, the literature suggests that SREBP-2 recruits the HATs p300 and CBP to SREBP-2 target sites in proximity to genes such as HMGCR, increasing histone acetylation at these sites and thereby activating gene expression (Figure 6-14). Oliner et al.³²² found that SREBP-2 interacts with p300 and CBP, and that CBP enhances the transcriptional activity of SREBP-2 on a reporter construct containing the LDLR promoter. Furthermore, in a study of Chinese hamster ovary CHO-7 cells incubated in media containing LPDS with or without cholesterol supplementation, cells in cholesterol deficient media displayed increased H3 acetylation in proximity to the HMGCR and LDLR promoters compared to the cholesterol supplemented cells.¹¹¹ Neither of these studies used statins to activate SREBP-2, this finding of an epigenetic effect of statin treatment is thus a novel finding.



Figure 6-14 Postulated mechanism by which statins may impact histone acetylation

Statins inhibit HMGCR, leading to decreased cellular cholesterol. This leads to the activation of SREBP-2, which recruits p300/CBP to SREBP-2 target sites in the promoter regions of genes such as *HMGCR* and *LDLR*, increasing histone acetylation and activating gene expression.

6.5.4. Results of ChIP-qPCR experiments of butyrate treated cells depend on analysis and normalisation methods

The results of this study outline some of the difficulties in interpreting ChIP-qPCR data, particularly with treatments that may cause widespread changes to chromatin. The most widely used methods are the percentage input method and the fold enrichment method, with other methods including relative to nucleosome density and relative to control sequences.^{313,323,324} The fold enrichment method divides the ChIP signal by the 'beads only' signal. As the 'beads only' signal was uniformly low (0.30% for control cells and 0.29% for butyrate treated cells), this method is unsuitable for this data as any small technical variation may result in large differences in the transformed data that would be unlikely to reflect real differences.³¹³ In the percentage input method (i.e. relative to total DNA), the signals of the immunoprecipitated samples are divided by the signals from a portion of chromatin put aside before the ChIP reaction and is used in the majority of ChIP-qPCR

experiments. In context of histone modifications this method assumes equal nucleosome density, which was the case for the atorvastatin ChIP experiments but not the butyrate experiments, which had uniformly higher H3 signals than the control, and subsequently higher signals of all modified histones. Similar results were found with the raw data. For this reason, ChIP signals were also normalised relative to nucleosome density (total H3). This method removed the differences between control and butyrate treatment for H3K9 acetylation and trimethylation, while H4K8 acetylation remained significantly higher for butyrate treated cells, significant for SLC2A2 (GLUT2) and HMGCR. This may be a real effect, as butyrate does increase global H4 acetylation (see section 6.4.3), but it may be an effect of differential antibody binding efficiency. Another normalisation method is relative to a control gene, in this case GAPDH. An advantage of this technique is that it eliminates technical error, in that it directly compares the same sample whereas other methods compare different antibodies or controls which are handled separately following chromatin isolation.³¹³ This method assumes that the treatment has no effect on histone modifications at the control gene, which may not be the case with broad epigenetic modifiers, such as HDAC inhibitors, that drastically alter global histone acetylation. As a result, data were also normalised to average values, where the signal was divided by the average signal of that antibody across all genomic regions. With these methods, the increase in H4K8 acetylation at SLC2A2 remained (p=0.09 and p=0.03 for the control sequence and average value method respectively), while there was no difference in H4K8 acetylation at HMGCR (p=0.45 and p=0.85). Thus, an increase in H4K8 acetylation in proximity to SLC2A2 is the only histone modification observed consistently across different normalisation methods. Of course, the use of multiple normalisation methods does increase uncertainty and it's possible some real effects have been masked.

The H3 antibody used in these experiments (ab1791) is considered the gold standard for epigenomic experiments³²⁵ and immunoblotting with this antibody found that butyrate had no effect on overall histone H3 levels (Supplementary Figure S14). Increased H3 occupancy is expected to correlate with inaccessible chromatin,³²⁵ however, the increased H3 signal was seen in all genomic regions tested, including in the proximity of *HK1* and *ABCG1*, both of which were upregulated more than 100-fold by butyrate (see section 5.4.5), and thus is unlikely to be a result of chromatin compaction. It may be that instead butyrate relaxed chromatin compaction and thus increases the availability of histones to the antibodies. Genome-wide experimental

methods such as ChIP-seq and Assay of Transposase Accessible Chromatin sequencing (ATAC-seq), which maps accessible chromatin,³²⁶ would be useful to determine if this increased H3 signal occurs across the genome and how it correlates with chromatin compaction.

The increase in H4K8 acetylation in proximity to *SLC2A2* with butyrate treatment is a novel finding. *SLC2A2* gene expression has previously been shown to be upregulated by butyrate in intestinal cells *in vitro*³²⁷ and *in vivo*,³²⁸ however, these studies did not examine histone modifications. Although the increase in *SLC2A2* gene expression here did not result in a noticeable increase in GLUT2 protein levels (see section 5.4.4), butyrate has been reported to increase GLUT2 protein levels in the livers of HFD fed rodents^{118,211} and in insulin-resistant HepG2 cells.¹¹⁸ Whether this epigenetic modification results in a phenotypic change, i.e., increased glucose uptake, remains an area of further research.

6.6. Conclusion

In contrast to the published literature, the results presented here indicate that statins are unlikely to inhibit the three major classes of epigenetic modifying enzymes, namely HDACs, HATs, and DNMTs, and in fact may increase DNMT activity. This represents the most comprehensive study of HDAC activity, using multiple cell types, in both live cells and nuclear extracts and using two different HDAC substrates to demonstrate no effect with statin treatment. While our in vitro studies indicate it is unlikely that stating directly inhibit these epigenetic enzymes, it is feasible that through effects on cell metabolism and/or phenotype, statins may have indirect effects on specific genomic regions. In fact, while the experiments on the livers of statin treated mice would also seem to rule out effects of prolonged treatment on HDAC activity, direct or otherwise, atorvastatin did increase H4K8 acetylation in proximity to the SREBP-2 binding site of the HMGCR gene. It is likely that while statins do not directly impact epigenetic changes, they instead indirectly impact histone modifications in proximity to SREBP-2 target genes as a result of their cholesterol lowering activity, the activation of SREBP-2 and the subsequent recruitment of HATs to SREBP-2 target sites.

Conversely, butyrate greatly inhibited HDAC activity and increased global H3K9 and H4 acetylation in all experiments. The noted differences in gene expression in Chapters 4 and 5 reflect these differences in epigenetic mechanisms, with butyrate

impacting the expression of many genes involved in different pathways, presumably due to direct effects of HDAC inhibition on the genes and proteins themselves, while all the noted effects of atorvastatin, including the epigenetic effect described in this chapter, seem to occur as a result of cellular cholesterol lowering, rather than a direct effect of the statin itself, and hence are more targeted towards cholesterol metabolism pathways. However, the specific effects of butyrate at the gene level remain unclear as ChIP results varied significantly with different normalisation methods. It is likely that butyrate increases *SLC2A2* expression by increasing H4K8 acetylation, presumably as a result of HDAC inhibition, a finding that has not been previously reported.

Chapter 7. Concluding remarks

This thesis is the first study to compare the effects of the statin class of cholesterol lowering drugs, which had previously been reported to inhibit HDACs, and HDAC inhibitors, which have been reported to alter cholesterol metabolism, on epigenetic modifications, cholesterol metabolism and glucose metabolism. This includes the most comprehensive examination of the epigenetic effects of statins yet conducted as well as significantly expanding on the existing literature on the effects of HDAC inhibitors on cholesterol and glucose metabolism *in vitro*.

The path taken by this thesis was driven by the data, firstly that in literature and then by the findings of preliminary experiments. As discussed in Allen and Mamotte²⁸ and in section 2.5.3, several studies had suggested statins have epigenetic effects, most notably HDAC inhibition, but none were conducted in the context of lipid disorders or diabetes, which are areas of interest for our laboratory group. Thus, the original plan was to explore how epigenetics may contribute to the diabetogenic effects of statins. Preliminary experiments showed that statins had no effect on HDAC activity, while butyrate, originally used as a control HDAC inhibitor, was shown to lower cellular cholesterol to a similar extent as the stating, leading us to examine the potential for butyrate to treat or prevent lipid disorders and diabetes, as reviewed in Bridgeman et al.¹⁶³ Metformin, a very widely used anti-diabetic drug, was also briefly investigated as it has been reported to have numerous epigenetic effects, as reviewed in Bridgeman et al.⁷⁷ Preliminary experiments with butyrate showed that the HDAC inhibitor had widespread effects worthy of further investigation and perhaps it would be more important to include another HDAC inhibitor, namely trichostatin A, rather than an unrelated extra treatment like metformin. The scope of the study was thus to compare and contrast the effects of statins and established HDAC inhibitors, primarily butyrate, on epigenetic modifications, cholesterol metabolism and insulin signalling and secretion. This is reflected in the literature review (Chapter 2), which evolved to include not only a thorough review of the epigenetic effects of statins, but contralaterally the potential beneficial effects of butyrate on lipid metabolism and the metabolic syndrome. The emphasis in the experimental chapters was thus on how butyrate influences cholesterol metabolism and how statins influence epigenetic modifications, as well as how both treatments may impact β cell function and potentially diabetes risk.

Possibly the most significant novel finding was that butyrate lowers cellular cholesterol content in both a hepatic and an insulin-secreting cell line and that this is most likely due to its actions as an HDAC inhibitor, since the potent HDAC inhibitor TSA also reduced cholesterol content, as outlined in Chapter 4. In particular, the data suggests that HDAC inhibition lowers cellular cholesterol by inhibiting SREBP-2 activity in a time-dependant manner, without altering SREBF2 mRNA expression. The mechanism by which butyrate alters SREBP-2 activity is currently unknown and is under investigation by our group, specifically by looking at the localisation and acetylation of the SREBP-2 protein. As a result of the inhibition of SREBP-2 activity and subsequent reduced LDLR expression, butyrate did not increase cellular cholesterol uptake whereas statins did, and thus butyrate may not be able to lower plasma LDL-C in vivo as statins do. This chapter also explored how butyrate and atorvastatin affect proteins involved in reverse cholesterol transport (ABCA1, ABCG1, SRB1), with results differing according to cell type and amount of lipoprotein in the serum, making it difficult to speculate on how these treatments may impact reverse cholesterol transport in vivo.

An interesting finding regarding β cell insulin secretion, described in Chapter 5, is that substantial HDAC inhibition (5 mM butyrate or 2.5 μ M TSA x 24 h) impairs insulin secretion while acute or low dose (1 mM) sodium butyrate enhances insulin secretion stimulated by high glucose and alanine. HDAC inhibition reduced both 24 h insulin secretion in cell media and the insulin content of cells; this was significant for 5 mM butyrate and TSA. The impaired β cell function could be explained by the noted upregulation of hexokinase I, which would interfere with glucose sensing at both low and high glucose, and of *TXNIP*, which would increase oxidative stress. At the same time, the data presented in this chapter suggests that statin-induced β cell dysfunction is a direct result of cellular cholesterol lowering rather than an off-target effect, as all genes linked to β cell function and diabetes with statin-altered expression are regulated directly by transcription factors whose activity is dependent on cellular cholesterol levels.

In addition to β cell insulin secretion, Chapter 5 also examined insulin signalling in hepatic cells, finding that butyrate reduced AKT phosphorylation while tending to increase unstimulated IRS-1 phosphorylation. Interestingly, both of these findings can be explained by increased acetylation of the respective proteins, as the literature suggests that acetylation of IRS-1 increases its phosphorylation while acetylation of
AKT inhibits its ability to be phosphorylated. Similarly, the literature suggests other proteins in the insulin signalling pathway can be acetylated and whether acetylation acts to increase or decrease insulin signalling varies from protein to protein. As a result, it is difficult to speculate how HDAC inhibition may impact insulin sensitivity.

Contrary to the published literature, summarised in Allen and Mamotte,²⁸ the findings in Chapter 6 demonstrate that statins do not directly inhibit the activity of epigenetic modifying enzymes, namely HDACs, HATs and DNMTs, or increase global histone acetylation. Statins may however alter histone acetylation at certain genomic sites, namely SREBP-2 target sites, through the activation of SREBP-2 and its subsequent recruitment of histone acetyltransferases, reflected in the finding that atorvastatin significantly increased H4K8 acetylation in proximity to the SREBP-2 target site in *HMGCR*. Unfortunately, the effects of butyrate on histone acetylation at target genes varied greatly depending on the normalisation method used. It does seem that butyrate upregulates *SLC2A2* (GLUT2) gene expression by increasing H4K8 acetylation in its vicinity, a finding that has not previously been reported despite other studies finding that butyrate upregulates GLUT2 expression and may consequently increase glucose uptake.

7.1. Future directions and perspectives

In research, there are always numerous further experiments that could be conducted as one set of results opens many rabbit holes that one could dive into. To start, experiments that ideally would have been included in this thesis but were not due to time and budgetary constraints include the use of primary cells rather than cell lines, ChIP analysis of BRIN-BD11 cells and bisulfite sequencing to examine changes in DNA methylation. Methods to assess the effect of HDAC inhibition on genome-wide chromatin changes such ChIP-seq and ATAC-seq would be ideal to resolve the uncertainty regarding the ChIP-qPCR results of butyrate treated cells. Repeating experiments with additional statins, particularly the hydrophilic rosuvastatin and pravastatin, would help elucidate if the findings with atorvastatin can apply to statins in general. Due to the often-opposing effects of atorvastatin and butyrate, for example on SREBP-2 signalling, it would also be interesting to conduct more experiments using HDAC inhibitor and statin co-treatments; this may have important clinical implications due to the high percentage of people with metabolic disorders already using statins. Other experiments could be used to further support the hypotheses ascertained from the results. The hypothesis that statins indirectly alter histone acetylation through SREBP-2 recruitment of HATs to target genes could be explored by statin treatment of SREBP-2 knock-out cells and by co-treating with the p300/CBP HAT inhibitor A-485.³²⁹ Whether butyrate alters non-histone acetylation of SREBP-2, IRS-1 and AKT could be determined using immunoprecipitation with antibodies against the protein in question followed by immunoblotting with anti-acetyl lysine antibodies.^{112,302} The importance of ABCG1 in the impairment of insulin secretion by statins could be examined by statin treatment of ABCG1 knock-out and constitutively expressed cells and by co-treating with LXR agonists such as T0901317.³³⁰

Future animal models could be designed to ascertain whether the systemic effects of butyrate *in vivo* are a result of HDAC inhibition or effects on intestinal cells such as GLP-1 secretion. This could be accomplished by giving one group of HFD fed animals oral butyrate and another intravenous butyrate at a dose that gives comparable butyrate concentrations in peripheral circulation and comparing their metabolic parameters.

On a broader note, there is a need for well-designed placebo-controlled clinical trials for the use of butyrate for dyslipidaemia and T2D. The only studies reported so far lack statistically significant results, possibly due to their small sample sizes.^{6,7,166} It may be that the uptake of butyrate is not sufficient to exert significant effects on nonintestinal tissues. For this reason, butyrate pro-drugs designed to improve the cellular uptake of butyrate have been developed. Of these, AN-9 (pivaloyloxymethyl butyrate, also known as Pivanex)³³¹ and tributyrin (a triglyceride with three esterified butyrate molecules)^{332,333} have reached clinical trials, but in a very different context, namely for treatment of cancers. Interestingly, these trials reported hyperglycaemia as an adverse effect, although as they were not placebo-controlled it is unknown if this is an actual side effect. Nevertheless, it is in line with the results in section 5.4.1 suggesting that high-dose HDAC inhibition impairs β cell function. Further studies, including animal models, at appropriate doses are needed to ascertain if these prodrugs may be useful in metabolic disorders.

Due to the wide-ranging effects of epigenetic modifying enzymes on gene expression across the genome, the future of altering epigenetic modifications for therapeutic reasons may not lie with altering the activity of these enzymes but instead in targeting epigenetic modifications at specific genomic regions. One approach is using a modified CRISPR-Cas9 system, in which the Cas9 nuclease is deactivated (dCas9) and fused to an epigenetic modifying enzyme.³³⁴ Hilton et al.³³⁵ fused the HAT p300 to CRISPR-dCas9 to selectively activate numerous genes using appropriate guide RNAs in live cells and a similar approach using DNMT3a has been used to selectively silence genes.³³⁶ While this technology has a long way to go, it has proven effective in animal models. In a mouse model, a transactivation complex designed to selectively target *Pdx1* increased histone acetylation at *Pdx1* and partially ameliorated STZ-induced diabetes.³³⁷ The possibilities are intriguing. For example, could we see a future where statins users at risk of T2D have *ABCG1* epigenetically activated by CRISPR-dCas9 technology?

7.2. Limitations

Any research project will have its limitations, and the work in this thesis is no exception. In particular, the findings outlined in the thesis are limited by the fact that, with the exception of HDAC activity in the livers of statin-treated mice, all experiments were performed on cell lines either derived from human cancers (HepG2, THP-1, MDA-MB-231) or from the electrofusion of rat pancreatic islet cells and the insulinoma cell line RINm5F (BRIN-BD11).³³⁸ Thus by definition these are not normal hepatocytes, macrophages or β cells. Ideally results would be validated by the use of additional cell lines or primary cells. It was decided that rather than using multiple cell types of the same origin tissue, we would obtain a wider picture by using cell types from different tissues that are important in dyslipidaemia and/or T2D.

Furthermore, like the vast majority of *in vitro* studies, the doses used in experiments are supraphysiological compared to concentrations reported in human serum. For butyrate, average concentrations of close to 100 μ M have were reported in the portal vein following butyrate enemas;²⁷⁷ the concentrations used in this study were 10 to 50-fold higher. For statins, the mean concentrations in serum of therapeutic doses is 1-15 nM,³¹⁶ around 1000 fold less than the concentrations used in this and most other *in vitro* studies. It is thus unknown how relevant the novel findings of this thesis are to the use of statins and butyrate as human therapeutics.

As mentioned in section 1.4, this study was designed to be mechanistic in nature and thus relied on *in vitro* experiments, as there have been numerous *in vivo* studies on the effects of butyrate and statins on metabolic disorders but the majority lack evidence on the exacting mechanisms. In particular, for butyrate it is unknown what

proportion of the noted phenotypic changes are the result of HDAC inhibition on the relevant tissues and how much is the result of FFAR activation in the intestine and the subsequent release of hormones such as GLP-1. While *in vitro* experiments enable controlled experiments to be conducted without the complications caused by effects on other organs, without confirming that the novel findings outlined earlier in this chapter also occur *in vivo*, their relevance is unknown.

7.3. Clinical relevance

Statins are very widely used to lower LDL-C, especially in advanced economies. However, they are associated with a small but significant increased risk of developing type 2 diabetes, although the benefits in reducing cardiovascular events greatly outweigh the risks.³³⁹ Exploring the mechanism behind the diabetogenic effects of statins may help reduce these risks or identify people in which the risk is highest, enabling early detection and intervention. The findings of this project suggest that statins do not have direct epigenetic effects, and all noted effects of statins are directly explained by cellular cholesterol lowering. This adds further evidence that the increased risk of diabetes with statins is related to cholesterol lowering, rather than an off-target effect. In particular, the lowering of cholesterol in β cells and the downstream downregulation of ABCG1 may impair insulin secretion. People without significantly elevated cholesterol or with diabetes risk factors may reduce this risk by using a hydrophilic statin such as pravastatin which is likely to have lower uptake in non-hepatic tissues.¹⁴²

HDAC inhibitors such as butyrate are of interest as therapeutics to protect against diabetes and the metabolic syndrome, however they are limited by their poor uptake.¹⁶³ The data in this study suggests that HDAC inhibitors have detrimental effects on β cell function and insulin secretion, and that attempts to increase butyrate uptake may have unfavourable consequences. Instead of focusing on effects of HDAC inhibitors in general on peripheral tissues, it may be best to focus on the action of butyrate on intestinal cells and GLP-1 secretion. It may indeed be as a GLP-1 secretagogue that butyrate may one day be used as a therapeutic to protect against T2D.

7.4. Conclusion

This project demonstrated that both statins and HDAC inhibitors reduce cellular cholesterol in numerous cell types and impair β cell insulin secretion, although they act through different mechanisms. All noted effects of statins can be linked directly to their primary mechanism of reducing cholesterol biosynthesis through HMGCR inhibition, resulting in the activation of SREBP-2 target genes and the repression of LXR target genes. HDAC inhibitors conversely have more widespread and varied effects on genes and proteins involved in cholesterol and glucose metabolism, most likely due to both histone and non-histone lysine acetylation. This does include downregulation of *HMGCR* expression, explaining the lowered cellular cholesterol. In contrast to statins, HDAC inhibition downregulates SREBP-2 activity and LDLR levels, thus may not reduce circulating LDL-C as statins do.

Similarly, both statins and HDAC inhibitors have been shown to impair insulin secretion from β cell lines. The results of this study suggest that the detrimental effects of statins are a direct result of cellular cholesterol lowering, with *ABCG1* downregulation potentially playing an important role. Conversely, HDAC inhibitors altered the expression of numerous genes involved in insulin secretion and β cell function, with the upregulation of *HK1* and *TXNIP* potentially leading to oxidative stress and β cell exhaustion.

The postulated mechanisms by which statins and butyrate alter cellular cholesterol and insulin secretion are summarised in Figure 7-1.



Figure 7-1 Postulate mechanisms by which statins and butyrate impact cellular cholesterol and insulin secretion

Statins inhibit HMGCR activity and thus reduce cellular cholesterol synthesis, and all noted effects can be explained by their lowering of cellular cholesterol. Low levels of cholesterol activate SREBP-2, enabling it to recruit HATs to HMGCR and LDLR, increasing their expression. Increased LDLR expression leads to increased cholesterol uptake. However, lowered cellular cholesterol results in the repression of ABCA1 and ABCG1, regulated by LXR. Decreased ABCG1 in particular may impair insulin secretion. Conversely, HDAC inhibition by butyrate inhibits SREBP-2 activity, leading to decreased HMGCR and LDLR expression and thus reduced cellular cholesterol. Butyrate also increases HK1 and TXNIP expression, leading to increased glucose sensing and increased oxidate stress respectively, both of which may result in impaired insulin secretion.

In stark contrast to the literature, the results presented here show that statins are unlikely to impact epigenetic modifications directly and noted epigenetic effects may be due to the downstream effects of cholesterol lowering and subsequent effects on transcription factors, such as SREBP-2, which interact with epigenetic modifying enzymes.

HDAC inhibitors have been suggested as treatments for diabetes and other metabolic disorders due to the recent surge of research linking epigenetic modifications to these diseases and have shown promising results in animal models. This project highlights the danger of broad-spectrum epigenetic modifiers due to significant effects across the genome that may have detrimental and unpredictable consequences.

Chapter 8. References

- Boyanapalli SSS, Kong A-NT. Curcumin, the king of spices: Epigenetic regulatory mechanisms in the prevention of cancer, neurological, and inflammatory diseases. *Curr Pharmacol Rep.* 2015;1(2):129-39. doi:10.1007/s40495-015-0018-x
- Wang Y-J, Pan M-H, Cheng A-L, et al. Stability of curcumin in buffer solutions and characterization of its degradation products. *J Pharm Biomed*. 1997;15(12):1867-76. doi:10.1016/S0731-7085(96)02024-9
- 3. Medical Expenditure Panel Survey. Prescribed Drug Estimates: 2015: Agency for Healthcare Research and Quality; Available from: https://meps.ahrq.gov/data_stats/summ_tables/hc/drugs/2015/hcdrugest_totp ur2015.shtml accessed 13 Feb 2019.
- 4. Prescribing and Medicine Team. Prescriptions Dispensed in the Community— Statistics for England, 2007–2017: NHS Digital; Available from: https://digital.nhs.uk/data-andinformation/publications/statistical/prescriptions-dispensed-in-thecommunity/prescriptions-dispensed-in-the-community-england---2007---2017 accessed Feb 13 2019.
- 5. PBS Information Management Section. Expenditure and prescriptions twelve months to 30 June 2016. Department of Health. Canberra, 2016.
- 6. Roshanravan N, Mahdavi R, Alizadeh E, et al. Effect of butyrate and inulin supplementation on glycemic status, lipid profile and glucagon-like peptide 1 level in patients with type 2 diabetes: a randomized double-blind, placebocontrolled trial. *Horm Metab Res.* 2017;49(11):886-91. doi:10.1055/s-0043-119089
- 7. Roshanravan N, Mahdavi R, Jafarabadi MA, et al. The effects of sodium butyrate and high-performance inulin supplementation on the promotion of gut bacterium Akkermansia muciniphila growth and alterations in miR-375 and KLF5 expression in type 2 diabetic patients: A randomized, double-blind, placebo-controlled trial. *Eur J Integr Med.* 2018;18:1-7. doi: 10.1016/j.eujim.2017.12.011

- 8. World Health Organisation. Global health estimates 2016: Estimated deaths by cause and region, 2000 and 2016. Geneva, 2018.
- 9. Saeedi P, Petersohn I, Salpea P, et al. Global and regional diabetes prevalence estimates for 2019 and projections for 2030 and 2045: Results from the International Diabetes Federation Diabetes Atlas, 9th edition. *Diabetes Res Clin Pract.* 2019;157 doi:10.1016/j.diabres.2019.107843
- Saeedi P, Salpea P, Karuranga S, et al. Mortality attributable to diabetes in 20-79 years old adults, 2019 estimates: Results from the International Diabetes Federation Diabetes Atlas, 9th edition. *Diabetes Res Clin Pract*. 2020;162 doi: 10.1016/j.diabres.2020.108086
- Barr ELM, Magliano DJ, Zimmet PZ, et al. AusDiab 2005: The Australian Diabetes, Obesity and Lifestyle Study. Tracking the Accelerating Epidemic: Its Causes and Outcomes. Melbourne, Australia: International Diabetes Institute, 2006.
- 12. High Cholesterol Statistics. Melbourne: National Heart Foundation of Australia, 2012.
- Taylor F, Huffman MD, Macedo AF, et al. Statins for the primary prevention of cardiovascular disease. *Cochrane Database Syst Rev.* 2013(1) doi:10.1002/14651858.CD004816.pub5
- 14. Sattar N, Preiss D, Murray HM, et al. Statins and risk of incident diabetes: A collaborative meta-analysis of randomised statin trials. *Lancet*. 2010;375(9716):735-42. doi:10.1016/S0140-6736(09)61965-6
- Braun KV, Voortman T, Dhana K, et al. The role of DNA methylation in dyslipidaemia: a systematic review. *Prog Lipid Res.* 2016;64:178-91. doi:10.1016/j.plipres.2016.10.002
- Meaney S. Epigenetic regulation of cholesterol homeostasis. *Front Genet*. 2014;5:311. doi:10.3389/fgene.2014.00311
- Burgio E, Lopomo A, Migliore L. Obesity and diabetes: from genetics to epigenetics. *Mol Biol Rep.* 2015;42(4):799-818. doi:10.1007/s11033-014-3751-z
- Gilbert ER, Liu D. Epigenetics: the missing link to understanding beta-cell dysfunction in the pathogenesis of type 2 diabetes. *Epigenetics*. 2012;7(8):841-52.
- 19. Ling C, Groop L. Epigenetics: a molecular link between environmental factors and type 2 diabetes. *Diabetes*. 2009;58 doi:10.2337/db09-1003

- 20. Sommese L, Zullo A, Mancini FP, et al. Clinical relevance of epigenetics in the onset and management of type 2 diabetes mellitus. *Epigenetics*. 2017:1-15. doi:10.1080/15592294.2016.1278097
- 21. Webster ALH, Yan MS-C, Marsden PA. Epigenetics and cardiovascular disease. *Can J Cardiol.* 2013;29(1):46-57. doi:10.1016/j.cjca.2012.10.023
- 22. Abi Khalil C. The emerging role of epigenetics in cardiovascular disease. *Ther Adv Chronic Dis.* 2014;5(4):178-87. doi: 10.1177/2040622314529325
- 23. Ordovas JM, Smith CE. Epigenetics and cardiovascular disease. *Nat Rev Cardiol*. 2010;7(9):510-19.
- Sharma S, Taliyan R. Histone deacetylase inhibitors: future therapeutics for insulin resistance and type 2 diabetes. *Pharmacol Res.* 2016;113(Pt A):320-26. doi:10.1016/j.phrs.2016.09.009
- 25. Chittur SV, Sangster-Guity N, McCormick PJ. Histone deacetylase inhibitors: A new mode for inhibition of cholesterol metabolism. *BMC Genomics*. 2008;9:507-07. doi:10.1186/1471-2164-9-507
- Chuengsamarn S, Rattanamongkolgul S, Luechapudiporn R, et al. Curcumin extract for prevention of type 2 diabetes. *Diabetes Care*. 2012;35(11):2121-27. doi:10.2337/dc12-0116
- 27. Qin S, Huang L, Gong J, et al. Efficacy and safety of turmeric and curcumin in lowering blood lipid levels in patients with cardiovascular risk factors: a meta-analysis of randomized controlled trials. *Nut J.* 2017;16(1):68. doi:10.1186/s12937-017-0293-y
- 28. Allen SC, Mamotte CDS. Pleiotropic and adverse effects of statins: do epigenetics play a role? J Pharmacol Exp Ther 2017;362(2):319-26. doi:10.1124/jpet.117.242081
- Clarke R, Baigent C. Cholesterol and Lipids. In: Quah SR, ed. International Encyclopedia of Public Health (Second Edition). Oxford: Academic Press 2017:9-21.
- 30. Arguello G, Balboa E, Arrese M, et al. Recent insights on the role of cholesterol in non-alcoholic fatty liver disease. *Biochim Biophys Acta Mol Basis Dis.* 2015;1852(9):1765-78. doi:10.1016/j.bbadis.2015.05.015
- Brunham LR, Kruit JK, Verchere CB, et al. Cholesterol in islet dysfunction and type 2 diabetes. *J Clin Investig.* 2008;118(2):403-08. doi:10.1172/JCI33296
- 32. Afonso MS, Machado RM, Lavrador M, et al. Molecular Pathways Underlying Cholesterol Homeostasis. *Nutrients*. 2018;10(6):760.

- McNamara DJ. Cholesterol: sources, absorption, function, and metabolism. In: Caballero B, ed. *Encyclopedia of Human Nutrition* (Third Edition). Waltham: Academic Press 2013:341-45.
- Buhaescu I, Izzedine H. Mevalonate pathway: a review of clinical and therapeutical implications. *Clin Biochem.* 2007;40(9):575-84. doi:10.1016/j.clinbiochem.2007.03.016
- 35. Belinsky SA, Grimes MJ, Picchi MA, et al. Combination therapy with vidaza and entinostat suppresses tumor growth and reprograms the epigenome in an orthotopic lung cancer model. *Cancer Res.* 2011;71 doi:10.1158/0008-5472.can-10-3184
- 36. Jeon H, Blacklow SC. Structure and physiologic function of the low-density lipoprotein receptor. Annu Rev Biochem. 2005;74(1):535-62. doi:10.1146/annurev.biochem.74.082803.133354
- Eberlé D, Hegarty B, Bossard P, et al. SREBP transcription factors: master regulators of lipid homeostasis. *Biochimie*. 2004;86(11):839-48. doi:10.1016/j.biochi.2004.09.018
- Edwards PA, Kennedy MA, Mak PA. LXRs: oxysterol-activated nuclear receptors that regulate genes controlling lipid homeostasis. Vascul Pharmacol. 2002;38(4):249-56. doi:10.1016/S1537-1891(02)00175-1
- Li AC, Glass CK. PPAR- and LXR-dependent pathways controlling lipid metabolism and the development of atherosclerosis. J Lipid Res. 2004;45(12):2161-73. doi:10.1194/jlr.R400010-JLR200
- 40. Olson RE. Discovery of the lipoproteins, their role in fat transport and their significance as risk factors. J Nutr. 1998;128(2):439S-43S. doi:10.1093/jn/128.2.439S
- 41. Baigent C, Blackwell L, Emberson J, et al. Efficacy and safety of more intensive lowering of LDL cholesterol: a meta-analysis of data from 170,000 participants in 26 randomised trials. *Lancet*. 2010;376(9753):1670-81. doi:10.1016/s0140-6736(10)61350-5
- 42. Linton MF, Yancey PG, Davies SS, et al. The role of lipids and lipoproteins in atherosclerosis. In: De Groot LJ, G. C, Dungan K, eds. *Endotext*. MDText.com, Inc, 2019. South Dartmouth (MA). Accessed April 14, 2020. https://www.endotext.org/

- 43. Chistiakov DA, Bobryshev YV, Orekhov AN. Macrophage-mediated cholesterol handling in atherosclerosis. J Cell Mol Med, 2016;20(1):17-28. doi:10.1111/jcmm.12689
- Vance JE. Dysregulation of cholesterol balance in the brain: contribution to neurodegenerative diseases. *Dis Model Mech.* 2012;5(6):746-55. doi:10.1242/dmm.010124
- 45. Röder PV, Geillinger KE, Zietek TS, et al. The role of SGLT1 and GLUT2 in intestinal glucose transport and sensing. *PLoS ONE*. 2014;9(2):e89977-e77. doi:10.1371/journal.pone.0089977
- 46. Iynedjian PB. Mammalian glucokinase and its gene. *Biochem J.* 1993;293(Pt 1):1-13. doi:10.1042/bj2930001
- 47. Lemaire K, Thorrez L, Schuit F. Disallowed and allowed gene expression: two faces of mature islet beta cells. *Annu Rev Nutr.* 2016;36(1):45-71. doi:10.1146/annurev-nutr-071715-050808
- 48. Huang S, Czech MP. The GLUT4 glucose transporter. *Cell Metab*. 2007;5(4):23752. doi:10.1016/j.cmet.2007.03.006
- 49. Han H-S, Kang G, Kim JS, et al. Regulation of glucose metabolism from a livercentric perspective. *Exp Mol Med.* 2016;48(3):e218-e18. doi:10.1038/emm.2015.122
- 50. Baltrusch S, Tiedge M. Glucokinase regulatory network in pancreatic β-cells and liver. *Diabetes* 2006;55(S2):S55-S64. doi:10.2337/db06-S008
- 51. Briant L, Salehi A, Vergari E, et al. Glucagon secretion from pancreatic α-cells. Ups J Med Sci. 2016;121(2):113-19. doi:10.3109/03009734.2016.1156789
- Petersen KF, Laurent D, Rothman DL, et al. Mechanism by which glucose and insulin inhibit net hepatic glycogenolysis in humans. *J Clin Invest*. 1998;101(6):1203-09. doi:10.1172/JCI579
- 53. Zhang X, Yang S, Chen J, et al. Unraveling the regulation of hepatic gluconeogenesis. *Front Endocrin.* 2019;9(802) doi:10.3389/fendo.2018.00802
- 54. Aronoff SL, Berkowitz K, Shreiner B, et al. Glucose metabolism and regulation: beyond insulin and glucagon. *Diabetes Spectr.* 2004;17(3):183-90. doi:10.2337/diaspect.17.3.183
- 55. Bazotte RB, Silva LG, Schiavon FP. Insulin resistance in the liver: deficiency or excess of insulin? *Cell cycle*. 2014;13(16):2494-500. doi:10.4161/15384101.2014.947750

- 56. Cersosimo E, Triplitt C, Solis-Herrera C, et al. Pathogenesis of type 2 diabetes Mellitus. In: Feingold KR AB, Boyce A, et al., eds. *Endotext*. MDText.com, Inc, 2018. South Dartmouth (MA). Accessed April 14, 2020. https://www.endotext.org/
- 57. Cerf ME. Beta cell dysfunction and insulin resistance. *Front Endocrinol*. 2013;4:37-37. doi:10.3389/fendo.2013.00037
- Fowler MJ. Microvascular and macrovascular complications of diabetes. *Clin Diabetes*. 2008;26(2):77-82. doi:10.2337/diaclin.26.2.77
- 59. Tuomi T, Santoro N, Caprio S, et al. The many faces of diabetes: a disease with increasing heterogeneity. Lancet 2014;383(9922):1084-94. doi:10.1016/s0140-6736(13)62219-9
- 60. Bitzur R, Cohen H, Kamari Y, et al. Triglycerides and HDL cholesterol: stars or second leads in diabetes? *Diabetes Care*. 2009;32(suppl 2):S373-S377. doi:10.2337/dc09-S343
- Rutti S, Ehses JA, Sibler RA, et al. Low- and high-density lipoproteins modulate function, apoptosis, and proliferation of primary human and murine pancreatic beta-cells. *Endocrinology* 2009;150(10):4521-30. doi:10.1210/en.2009-0252
- Roehrich ME, Mooser V, Lenain V, et al. Insulin-secreting beta-cell dysfunction induced by human lipoproteins. J Biol Chem. 2003;278(20):18368-75. doi: 10.1074/jbc.M300102200
- Hao M, Head WS, Gunawardana SC, et al. Direct effect of cholesterol on insulin secretion: a novel mechanism for pancreatic beta-cell dysfunction. *Diabetes*. 2007;56(9):2328-38.
- 64. Sturek JM, Castle JD, Trace AP, et al. An intracellular role for ABCG1-mediated cholesterol transport in the regulated secretory pathway of mouse pancreatic β cells. J Clin Invest. 2010;120(7):2575-89. doi:10.1172/JCI41280
- 65. Ishikawa M, Iwasaki Y, Yatoh S, et al. Cholesterol accumulation and diabetes in pancreatic beta-cell-specific SREBP-2 transgenic mice: a new model for lipotoxicity. J Lipid Res. 2008;49(12):2524-34. doi:10.1194/jlr.M800238-JLR200
- 66. Llanos P, Contreras-Ferrat A, Georgiev T, et al. The cholesterol-lowering agent methyl-β-cyclodextrin promotes glucose uptake via GLUT4 in adult muscle fibers and reduces insulin resistance in obese mice. Am J Physiol Endocrinol Metab. 2015;308(4):E294-E305. doi:10.1152/ajpendo.00189.2014

- Drew BG, Duffy SJ, Formosa MF, et al. High-density lipoprotein modulates glucose metabolism in patients with type 2 diabetes mellitus. *Circulation*. 2009;119(15):2103-11. doi:10.1161/CIRCULATIONAHA.108.843219
- Siebel AL, Heywood SE, Kingwell BA. HDL and glucose metabolism: current evidence and therapeutic potential. *Front Pharmacol.* 2015;6:258-58. doi:10.3389/fphar.2015.00258
- Besseling J, Kastelein JP, Defesche JC, et al. Association between familial hypercholesterolemia and prevalence of type 2 diabetes mellitus. *JAMA*. 2015;313(10):1029-36. doi:10.1001/jama.2015.1206
- 70. Ference BA, Robinson JG, Brook RD, et al. Variation in PCSK9 and HMGCR and risk of cardiovascular disease and diabetes. N Engl J Med. 2016;375(22):2144-53. doi: 10.1056/NEJMoa1604304
- 71. Sabapathy T. The Interaction of Insulin with the Insulin Receptor in a Membrane Environment (Thesis). Curtin University, 2018.
- 72. Ellison G. The Influence of Cholesterol on β -cell Function (Thesis). Curtin University, 2019.
- 73. Kuroda A, Rauch TA, Todorov I, et al. Insulin gene expression is regulated by
 DNA methylation. *PLoS ONE*. 2009;4(9):e6953.
 doi:10.1371/journal.pone.0006953
- 74. Kanwal R, Gupta K, Gupta S. Cancer epigenetics: an introduction. *Methods Mol Biol.* 2015;1238:3-25. doi:10.1007/978-1-4939-1804-1
- 75. Lillycrop KA. Effect of maternal diet on the epigenome: implications for human metabolic disease. *Proc Nutr Soc.* 2011;70(1):64-72. doi:10.1017/S0029665110004027
- 76. Csoka AB, Szyf M. Epigenetic side-effects of common pharmaceuticals: A potential new field in medicine and pharmacology. *Med Hypotheses*. 2009;73(5):770-80. doi:10.1016/j.mehy.2008.10.039
- 77. Bridgeman SC, Ellison GC, Melton PE, et al. Epigenetic effects of metformin: From molecular mechanisms to clinical implications. *Diabetes Obes Metab*. 2018;20(7):1553-62. doi:10.1111/dom.13262
- 78. Clouaire T, Stancheva I. Methyl-CpG binding proteins: specialized transcriptional repressors or structural components of chromatin? *Cell Mol Life Sci.* 2008;65(10):1509-22. doi:10.1007/s00018-008-7324-y

- 79. Babenko VN, Chadaeva IV, Orlov YL. Genomic landscape of CpG rich elements in human. BMC Evol Biol. 2017;17(Suppl 1):19. doi10.1186/s12862-016-0864-0
- 80. Jeltsch A, Ehrenhofer-Murray A, Jurkowski TP, et al. Mechanism and biological role of Dnmt2 in nucleic acid methylation. *RNA Biol.* 2017;14(9):1108-23. doi:10.1080/15476286.2016.1191737
- 81. Jin B, Robertson KD. DNA methyltransferases (DNMTs), DNA damage repair, and cancer. Adv Exp Med Biol. 2013;754:3-29. doi:10.1007/978-1-4419-9967-2_1
- 82. Barau J, Teissandier A, Zamudio N, et al. The DNA methyltransferase DNMT3C protects male germ cells from transposon activity. *Science*. 2016;354(6314):909-12. doi:10.1126/science.aah5143
- Bannister AJ, Kouzarides T. Regulation of chromatin by histone modifications. Cell Res. 2011;21(3):381-95. doi10.1038/cr.2011.22
- 84. Lavu S, Boss O, Elliott PJ, et al. Sirtuins: novel therapeutic targets to treat ageassociated diseases. *Nat Rev Drug Discov*. 2008;7(10):841-53.
- Rossetto D, Avvakumov N, Côté J. Histone phosphorylation: a chromatin modification involved in diverse nuclear events. *Epigenetics* 2012;7(10):1098-108. doi:10.4161/epi.21975
- 86. Lara E, Calvanese V, Fernandez AF, et al. Techniques to study DNA methylation and histone modification. In: Roach HI, Bronner F, Oreffo ROC, eds. *Epigenetic Aspects of Chronic Diseases*. 2011:21-39. London: Springer London
- 87. Au SL-K, Wong CC-L, Lee JM-F, et al. EZH2-Mediated H3K27me3 is involved in epigenetic repression of deleted in liver cancer 1 in human cancers. *PLoS ONE*. 2013;8(6):e68226. doi:10.1371/journal.pone.0068226
- Shi Y, Lan F, Matson C, et al. Histone demethylation mediated by the nuclear amine oxidase homolog LSD1. *Cell*. 2004;119(7):941-53. doi:10.1016/j.cell.2004.12.012
- Braun KVE, Dhana K, de Vries PS, et al. Epigenome-wide association study (EWAS) on lipids: the Rotterdam Study. *Clin Epigenetics*. 2017;9:15. doi:10.1186/s13148-016-0304-4
- 90. Pfeiffer L, Wahl S, Pilling LC, et al. DNA methylation of lipid-related genes affects blood lipid levels. *Circ Cardiovasc Genet*. 2015;8(2):334-42. doi:10.1161/circgenetics.114.000804

- 91. Sayols-Baixeras S, Subirana I, Lluis-Ganella C, et al. Identification and validation of seven new loci showing differential DNA methylation related to serum lipid profile: an epigenome-wide approach. The REGICOR study. *Hum Mol Genet.* 2016;25(20):4556-65. doi:10.1093/hmg/ddw285
- 92. Dekkers KF, van Iterson M, Slieker RC, et al. Blood lipids influence DNA methylation in circulating cells. *Genome Biol.* 2016;17(1):138. doi:10.1186/s13059-016-1000-6
- 93. Walaszczyk E, Luijten M, Spijkerman AMW, et al. DNA methylation markers associated with type 2 diabetes, fasting glucose and HbA1c levels: a systematic review and replication in a case–control sample of the Lifelines study. *Diabetologia* 2018;61(2):354-68. doi:10.1007/s00125-017-4497-7
- 94. Wahl S, Drong A, Lehne B, et al. Epigenome-wide association study of body mass index, and the adverse outcomes of adiposity. *Nature*. 2016;541:81. doi:10.1038/nature20784
- 95. Kriebel J, Herder C, Rathmann W, et al. Association between DNA methylation in whole blood and measures of glucose metabolism: KORA F4 study. *PLoS ONE*. 2016;11(3):e0152314. doi:10.1371/journal.pone.0152314
- 96. Kulkarni H, Kos MZ, Neary J, et al. Novel epigenetic determinants of type 2 diabetes in Mexican-American families. *Hum Mol Genet*. 2015;24(18):5330-44. doi:10.1093/hmg/ddv232
- 97. Hidalgo B, Irvin MR, Sha J, et al. Epigenome-wide association study of fasting measures of glucose, insulin, and HOMA-IR in the Genetics of Lipid Lowering Drugs and Diet Network study. *Diabetes.* 2014;63(2):801-7. doi:10.2337/db13-1100
- 98. Cardona A, Day FR, Perry JRB, et al. Epigenome-wide association study of incident type 2 diabetes in a British population: EPIC-Norfolk Study. *Diabetes.* 2019;68(12):2315-26. doi:10.2337/db18-0290
- 99. Chambers JC, Loh M, Lehne B, et al. Epigenome-wide association of DNA methylation markers in peripheral blood from Indian Asians and Europeans with incident type 2 diabetes: a nested case-control study. *Lancet Diabetes Endocrinol.* 2015;3(7):526-34. doi:10.1016/s2213-8587(15)00127-8
- 100. Al Muftah WA, Al-Shafai M, Zaghlool SB, et al. Epigenetic associations of type
 2 diabetes and BMI in an Arab population. *Clin Epigenetics*. 2016;8:13-13. doi:10.1186/s13148-016-0177-6

- 101. Meeks KAC, Henneman P, Venema A, et al. Epigenome-wide association study in whole blood on type 2 diabetes among sub-Saharan African individuals: findings from the RODAM study. *Int J Epidemiol.* 2019;48(1):58-70. doi:10.1093/ije/dyy171
- 102. Alhawiti NM, Al Mahri S, Aziz MA, et al. TXNIP in metabolic regulation: physiological role and therapeutic outlook. *Curr Drug Targets*. 2017;18(9):1095-103. doi:10.2174/1389450118666170130145514
- 103. Guay S-P, Brisson D, Munger J, et al. ABCA1 gene promoter DNA methylation is associated with HDL particle profile and coronary artery disease in familial hypercholesterolemia. *Epigenetics*. 2012;7(5):464-72. doi:10.4161/epi.19633
- 104. Xu L, Zheng D, Wang L, et al. GCK gene-body hypomethylation is associated with the risk of coronary heart disease. *Biomed Res Int.* 2014;2014:7. doi:10.1155/2014/151723
- 105. Tang L, Ye H, Hong Q, et al. Elevated CpG island methylation of GCK gene predicts the risk of type 2 diabetes in Chinese males. *Gene*. 2014;547(2):329-33. doi:10.1016/j.gene.2014.06.062
- 106. Rönn T, Ling C. DNA methylation as a diagnostic and therapeutic target in the battle against Type 2 diabetes. *Epigenomics*. 2015;7(3):451-60. doi:10.2217/epi.15.7
- 107. Remsberg JR, Ediger BN, Ho WY, et al. Deletion of histone deacetylase 3 in adult beta cells improves glucose tolerance via increased insulin secretion. *Mol Metab.* 2017;6(1):30-37. doi:10.1016/j.molmet.2016.11.007
- 108. Lundh M, Galbo T, Poulsen SS, et al. Histone deacetylase 3 inhibition improves glycaemia and insulin secretion in obese diabetic rats. *Diabetes Obes Metab*. 2015;17(7):703-07. doi:doi:10.1111/dom.12470
- 109. Lundh M, Christensen DP, Damgaard Nielsen M, et al. Histone deacetylases 1 and 3 but not 2 mediate cytokine-induced beta cell apoptosis in INS-1 cells and dispersed primary islets from rats and are differentially regulated in the islets of type 1 diabetic children. *Diabetologia*. 2012;55(9):2421-31. doi:10.1007/s00125-012-2615-0
- 110. Nunes MJ, Moutinho M, Gama MJ, et al. Histone deacetylase inhibition decreases cholesterol levels in neuronal cells by modulating key genes in cholesterol synthesis, uptake and efflux. *PLoS ONE*. 2013;8(1):e53394-e94. doi:10.1371/journal.pone.0053394

- 111. Bennett MK, Osborne TF. Nutrient regulation of gene expression by the sterol regulatory element binding proteins: Increased recruitment of gene-specific coregulatory factors and selective hyperacetylation of histone H3 in vivo. *PNAS*. 2000;97(12):6340-44. doi:10.1073/pnas.97.12.6340
- 112. Giandomenico V, Simonsson M, Gronroos E, et al. Coactivator-dependent acetylation stabilizes members of the SREBP family of transcription factors. *Mol Cell Biol*. 2003;23(7):2587-99. doi: 10.1128/mcb.23.7.2587-2599.2003
- 113. Ponugoti B, Kim DH, Xiao Z, et al. SIRT1 deacetylates and inhibits SREBP-1C activity in regulation of hepatic lipid metabolism. J Biol Chem. 2010;285(44):33959-70. doi:10.1074/jbc.M110.122978
- 114. Wondisford AR, Xiong L, Chang E, et al. Control of Foxo1 gene expression by co-activator P300. J Biol Chem. 2014;289(7):4326-33. doi:10.1074/jbc.M113.540500
- 115. Yao W, Wang T, Huang F. p300/CBP as a key nutritional sensor for hepatic energy homeostasis and liver fibrosis. *Biomed Res Int.* 2018;2018:8168791-91. doi:10.1155/2018/8168791
- 116. Oiso H, Furukawa N, Suefuji M, et al. The role of class I histone deacetylase (HDAC) on gluconeogenesis in liver. *Biochem Biophys Res Commun*. 2011;404(1):166-72. doi:10.1016/j.bbrc.2010.11.086
- 117. Sharma S, Taliyan R, Ramagiri S. Histone deacetylase inhibitor, trichostatin A, improves learning and memory in high-fat diet-induced cognitive deficits in mice. J Mol Neurosci. 2015;56(1):1-11. doi10.1007/s12031-014-0461-x
- 118. Mollica MP, Mattace Raso G, Cavaliere G, et al. Butyrate regulates liver mitochondrial function, efficiency, and dynamics in insulin-resistant obese mice. *Diabetes* 2017;66(5):1405-18. doi:10.2337/db16-0924
- 119. Gu Q, Paulose-Ram R, Burt VL, et al. Prescription cholesterol-lowering medication use in adults aged 40 and over: United States, 2003–2012. In: Centers for Disease Control and Prevention, ed. Hyattsville, MD: National Center for Health Statistics, 2014.
- 120. Naci H, Brugts JJ, Fleurence R, et al. Comparative benefits of statins in the primary and secondary prevention of major coronary events and all-cause mortality: a network meta-analysis of placebo-controlled and activecomparator trials. *Eur J Prev Cardiol.* 2013;20(4):641-57. doi:10.1177/2047487313480435

- 121. Goldstein JL, Brown MS. A century of cholesterol and coronaries: from plaques to genes to statins. *Cell*. 2015;161(1):161-72. doi:10.1016/j.cell.2015.01.036
- 122. Naci H, Brugts J, Ades T. Comparative tolerability and harms of individual statins. *Circ Cardiovasc Qual Outcomes*. 2013;6(4):390-99. doi:10.1161/CIRCOUTCOMES.111.000071
- 123. Weng TC, Yang YH, Lin SJ, et al. A systematic review and meta-analysis on the therapeutic equivalence of statins. *J Clin Pharm Ther.* 2010;35(2):139-51. doi: 10.1111/j.1365-2710.2009.01085.x
- 124. Cholesterol Treatment Trialists' Collaborators, Mihaylova B, Emberson J, et al. The effects of lowering LDL cholesterol with statin therapy in people at low risk of vascular disease: meta-analysis of individual data from 27 randomised trials. *Lancet.* 2012;380(9841):581-90.doi: 10.1016/S0140-6736(12)60367-5
- 125. Barter PJ, Brandrup-Wognsen G, Palmer MK, et al. Effect of statins on HDL-C: a complex process unrelated to changes in LDL-C: analysis of the VOYAGER Database. J Lipid Res. 2010;51(6):1546-53. doi:10.1194/jlr.P002816
- 126. Yada T, Nakata M, Shiraishi T, et al. Inhibition by simvastatin, but not pravastatin, of glucose-induced cytosolic Ca2+ signalling and insulin secretion due to blockade of L-type Ca2+ channels in rat islet beta-cells. Br J Pharmacol. 1999;126(5):1205-13.
- 127. Ishikawa M, Okajima F, Inoue N, et al. Distinct effects of pravastatin, atorvastatin, and simvastatin on insulin secretion from a beta-cell line, MIN6 cells. *J Atheroscler Thromb.* 2006;13(6):329-35.
- 128. Zúñiga-Hertz JP, Rebelato E, Kassan A, et al. Distinct pathways of cholesterol biosynthesis impact on insulin secretion. J Endocrinol. 2015;224(1):75-84. doi:10.1530/joe-14-0348
- 129. Zhao W, Zhao SP. Different effects of statins on induction of diabetes mellitus: an experimental study. *Drug Des Devel Ther.* 2015;9:6211-23. doi:10.2147/dddt.s87979
- 130. Ganesan S, Ito M. Coenzyme Q10 ameliorates the reduction in GLUT4 transporter expression induced by simvastatin in 3T3-L1 adipocytes. *Metab Syndr Relat Disord*. 2013;11(4):251-55.
- 131. Chamberlain LH. Inhibition of isoprenoid biosynthesis causes insulin resistance in 3T3-L1 adipocytes. *FEBS Lett.* 2001;507(3):357-61. doi:10.1016/S0014-5793(01)03007-1

- 132. Khan T, Hamilton MP, Mundy DI, et al. Impact of simvastatin on adipose tissue:
 pleiotropic effects in vivo. *Endocrinology*. 2009;150(12):5262-72.
 doi:10.1210/en.2009-0603
- 133. Nakata M, Nagasaka S, Kusaka I, et al. Effects of statins on the adipocyte maturation and expression of glucose transporter 4 (SLC2A4): implications in glycaemic control. *Diabetologia*. 2006;49(8):1881-92. doi:10.1007/s00125-006-0269-5
- 134. Nowis D, Malenda A, Furs K, et al. Statins impair glucose uptake in human cells.
 BMJ Open Diabetes Res Care. 2014;2(1):e000017. doi:10.1136/bmjdrc-2014-000017
- 135. Takaguri A, Satoh K, Itagaki M, et al. Effects of atorvastatin and pravastatin on signal transduction related to glucose uptake in 3T3L1 adipocytes. J Pharmacol Sci. 2008;107(1):80-9.
- 136. Pramfalk C, Parini P, Gustafsson U, et al. Effects of high-dose statin on the human hepatic expression of genes involved in carbohydrate and triglyceride metabolism. J Intern Med. 2011;269(3):333-39. doi:10.1111/j.1365-2796.2010.02305.x
- 137. Wang HJ, Park JY, Kwon O, et al. Chronic HMGCR/HMG-CoA reductase inhibitor treatment contributes to dysglycemia by upregulating hepatic gluconeogenesis through autophagy induction. *Autophagy*. 2015;11(11):2089-101. doi:10.1080/15548627.2015.1091139
- 138. Chen ZY, Liu SN, Li CN, et al. Atorvastatin helps preserve pancreatic beta cell function in obese C57BL/6 J mice and the effect is related to increased pancreas proliferation and amelioration of endoplasmic-reticulum stress. *Lipids Health Dis.* 2014;13:98. doi:10.1186/1476-511X-13-98
- 139. de las Heras N, Valero-Muñoz M, Ballesteros S, et al. Factors involved in rosuvastatin induction of insulin sensitization in rats fed a high fat diet. *Nutr Metab* Cardiovasc Dis. 2013;23(11):1107-14. doi:10.1016/j.numecd.2012.11.009
- 140. Naples M, Federico LM, Xu E, et al. Effect of rosuvastatin on insulin sensitivity in an animal model of insulin resistance: evidence for statin-induced hepatic insulin sensitization. *Atherosclerosis*. 2008;198(1):94-103. doi:10.1016/j.atherosclerosis.2007.11.003

- 141. Lalli CA, Pauli JR, Prada PO, et al. Statin modulates insulin signaling and insulin resistance in liver and muscle of rats fed a high-fat diet. *Metabolism*. 2008;57(1):57-65. doi:10.1016/j.metabol.2007.07.021
- 142. Climent E, Benaiges D, Pedro-Botet J. Hydrophilic or lipophilic statins? *Front Cardiovasc Med.* 2021;8:687585-85. doi:10.3389/fcvm.2021.687585
- 143. Navarese EP, Buffon A, Andreotti F, et al. Meta-analysis of impact of different types and doses of statins on new-onset diabetes mellitus. Am J Cardiol 2013;111(8):1123-30. doi:10.1016/j.amjcard.2012.12.037
- 144. Paseban M, Butler AE, Sahebkar A. Mechanisms of statin-induced new-onset diabetes. J Cell Physiol 2019;234(8):12551-61. doi:10.1002/jcp.28123
- 145. Lin YC, Lin JH, Chou CW, et al. Statins increase p21 through inhibition of histone deacetylase activity and release of promoter-associated HDAC1/2. *Canc Res.* 2008;68(7):2375-83. doi:10.1158/0008-5472.CAN-07-5807
- 146. Karlic H, Thaler R, Gerner C, et al. Inhibition of the mevalonate pathway affects epigenetic regulation in cancer cells. *Canc Genet*. 2015;208(5):241-52. doi:10.1016/j.cancergen.2015.03.008
- 147. Feig JE, Shang Y, Rotllan N, et al. Statins promote the regression of atherosclerosis via activation of the CCR7-dependent emigration pathway in macrophages. *PLoS ONE*. 2011;6(12):e28534.
- 148. Lüthje P, Walker S, Kamolvit W, et al. Statins influence epithelial expression of the anti-microbial peptide LL-37/hCAP-18 independently of the mevalonate pathway. *Clin Exp Immunol.* 2019;195(2):265-76. doi:10.1111/cei.13217
- 149. Tikoo K, Patel G, Kumar S, et al. Tissue specific up regulation of ACE2 in rabbit model of atherosclerosis by atorvastatin: role of epigenetic histone modifications. *Biochem Pharmacol.* 2015;93(3):343-51. doi:10.1016/j.bcp.2014.11.013
- 150. Cooney CA. Drugs and supplements that may slow aging of the epigenome. Drug Discov Today. 2010;7(3–4):57-64. doi:10.1016/j.ddstr.2011.03.001
- 151. Ishikawa S, Hayashi H, Kinoshita K, et al. Statins inhibit tumor progression via an enhancer of zeste homolog 2-mediated epigenetic alteration in colorectal cancer. *Int J Canc.* 2014;135(11):2528-36.
- 152. Kodach LL, Jacobs RJ, Voorneveld PW, et al. Statins augment the chemosensitivity of colorectal cancer cells inducing epigenetic reprogramming and reducing colorectal cancer cell 'stemness' via the bone morphogenetic protein pathway. *Gut.* 2011;60(11):1544-53.

- 153. Kim YC, Kim KK, Shevach EM. Simvastatin induces Foxp3+ T regulatory cells by modulation of transforming growth factor-beta signal transduction. *Immunology*. 2010;130(4):484-93.
- 154. Ochoa-Rosales C, Portilla-Fernandez E, Nano J, et al. Epigenetic link between statin therapy and type 2 diabetes. *Diabetes Care.* 2020:dc191828. doi:10.2337/dc19-1828
- 155. Liu Y, Shen Y, Guo T, et al. Statin use associates with risk of type 2 diabetes via epigenetic patterns at ABCG1. *Front Genet*. 2020;11(622) doi:10.3389/fgene.2020.00622
- 156. Yao C, Hwang S-J, Mendelson M, et al. Abstract 20759: DNA Methylation Signature of Statin Therapy. *Circulation*. 2017;136(S1):A20759-A59. doi:10.1161/circ.136.suppl_1.20759
- 157. Bourassa MW, Alim I, Bultman SJ, et al. Butyrate, neuroepigenetics and the gut microbiome: Can a high fiber diet improve brain health? *Neurosci Lett.* 2016;625(SC):56-63. doi:10.1016/j.neulet.2016.02.009
- 158. Waldecker M, Kautenburger T, Daumann H, et al. Inhibition of histonedeacetylase activity by short-chain fatty acids and some polyphenol metabolites formed in the colon. J Nutr Biochem. 2008;19(9):587-93. doi:10.1016/j.jnutbio.2007.08.002
- 159. Rada-Iglesias A, Enroth S, Ameur A, et al. Butyrate mediates decrease of histone acetylation centered on transcription start sites and down-regulation of associated genes. *Genome Res.* 2007;17(6):708-19. doi:10.1101/gr.5540007
- 160. Daly K, Shirazi-Beechey SP. Microarray analysis of butyrate regulated genes in colonic epithelial cells. *DNA Cell Biol.* 2006;25(1):49-62. doi: 10.1089/dna.2006.25.49
- 161. McNabney SM, Henagan TM. Short chain fatty acids in the colon and peripheral tissues: a focus on butyrate, colon cancer, obesity and insulin resistance. *Nutrients.* 2017;9(12) doi:10.3390/nu9121348
- 162. Steinert RE, Feinle-Bisset C, Asarian L, et al. Ghrelin, CCK, GLP-1, and PYY(3–36): secretory controls and physiological roles in eating and glycemia in health, obesity, and after RYGB. *Physiol Rev.* 2017;97(1):411-63. doi:10.1152/physrev.00031.2014
- 163. Bridgeman SC, Northrop W, Melton PE, et al. Butyrate, generated by gut microbiota, and its therapeutic role in metabolic syndrome. *Pharmacol Res.* 2020:105174. doi:10.1016/j.phrs.2020.105174

- 164. Zhang L, Du J, Yano N, et al. Sodium butyrate protects against high fat dietinduced cardiac dysfunction and metabolic disorders in type II diabetic mice. *J Cell Biochem.* 2017;118(8):2395-408. doi:10.1002/jcb.25902
- 165. Gao Z, Yin J, Zhang J, et al. Butyrate improves insulin sensitivity and increases energy expenditure in mice. *Diabetes*. 2009;58(7):1509-17. doi:10.2337/db08-1637
- 166. Bouter KEC, Bakker GJ, Levin E, et al. Differential metabolic effects of oral butyrate treatment in lean versus metabolic syndrome subjects. *Clin Transl Gastroenterol.* 2018;9(5):155. doi:10.1038/s41424-018-0025-4
- 167. Khan S, Jena G. Sodium butyrate reduces insulin-resistance, fat accumulation and dyslipidemia in type-2 diabetic rat: a comparative study with metformin. *Chem Biol Interact.* 2016;254:124-34. doi:10.1016/j.cbi.2016.06.007
- 168. Zhao Y, Liu J, Hao W, et al. Structure-specific effects of short-chain fatty acids on plasma cholesterol concentration in male Syrian hamsters. J Agric Food Chem. 2017;65(50):10984-92. doi:10.1021/acs.jafc.7b04666
- 169. Marcil V, Delvin E, Garofalo C, et al. Butyrate impairs lipid transport by inhibiting microsomal triglyceride transfer protein in Caco-2 cells. J Nutr. 2003;133(7):2180-83. doi:10.1093/jn/133.7.2180
- 170. Alvaro A, Solà R, Rosales R, et al. Gene expression analysis of a human enterocyte cell line reveals downregulation of cholesterol biosynthesis in response to short-chain fatty acids. *IUBMB Life.* 2008;60(11):757-64. doi:10.1002/iub.110
- 171. Kaptein A, de Wit ECM, Princen HMG. Butyrate stimulates the secretion of apolipoprotein B-100-containing lipoproteins from HepG2 cells by inhibiting the intracellular degradation. *Biochim Biophys Acta Mol Cell Biol Lipids*. 1994;1213(3):349-56. doi:10.1016/0005-2760(94)00064-6
- 172. Marcil V, Delvin E, Seidman E, et al. Modulation of lipid synthesis, apolipoprotein biogenesis, and lipoprotein assembly by butyrate. Am J Physiol Gastrointest Liver Physiol. 2002;283(2):G340-G46. doi:10.1152/ajpgi.00440.2001
- 173. Nazih H, Nazih-Sanderson F, Krempf M, et al. Butyrate stimulates ApoA-IV-containing lipoprotein secretion in differentiated Caco-2 cells: role in cholesterol efflux. J Cell Biochem. 2001;83(2):230-38. doi:doi:10.1002/jcb.1221

- 174. Kaptein A, Roodenburg L, Princen HMG. Butyrate stimulates the secretion of apolipoprotein A-I and apolipoprotein B-100 in Hep G2 cells by different mechanisms. *Clin Biochem.* 1992;25(5):317-19. doi:10.1016/0009-9120(92)80005-2
- 175. Guo Z, Yuan C, Wei-Lavery T, et al. Secretion of phospholipid transfer protein by human hepatoma cell line, Hep G2, is enhanced by sodium butyrate. J Nutr. 1999;129(11):1984-91. doi:10.1093/jn/129.11.1984
- 176. Sperker B, Mark M, Budzinski RM. The expression of human plasma cholesteryl-ester-transfer protein in HepG2 cells is induced by sodium butyrate. *Eur J Biochem.* 1993;218(3):945-50.
- 177. Skretting G, Gjernes E, Prydz H. Sodium butyrate inhibits the expression of the human lecithin: cholesterol acyltransferase gene in HepG2 cells by a post-transcriptional mechanism. *FEBS Lett.* 1997;404(1):105-10.
- 178. Chen Y, Xu C, Huang R, et al. Butyrate from pectin fermentation inhibits intestinal cholesterol absorption and attenuates atherosclerosis in apolipoprotein E-deficient mice. J Nutr Biochem. 2018;56:175-82. doi:10.1016/j.jnutbio.2018.02.011
- 179. Manns J, Boda J, Willes R. Probable role of propionate and butyrate in control of insulin secretion in sheep. Am J Physiol. 1967;212(4):756-64. doi:10.1152/ajplegacy.1967.212.4.756
- 180. Matheus V, Monteiro L, Oliveira R, et al. Butyrate reduces high-fat diet-induced metabolic alterations, hepatic steatosis and pancreatic beta cell and intestinal barrier dysfunctions in prediabetic mice. *Exp Biol Med.* 2017;242(12):1214-26. doi:10.1177/1535370217708188
- 181. Liu HK, Green BD, Flatt PR, et al. Effects of long-term exposure to nicotinamide and sodium butyrate on growth, viability, and the function of clonal insulin secreting cells. *Endocr Res.* 2004;30(1):61-68. doi:10.1081/ERC-120028485
- 182. Fernandez-Mejia C, Davidson MB. Effect of sodium butyrate on glucose transport and glucose-phosphorylating enzymes in RIN-m5F cells. *Pancreas* 1993;8(5):589-96.
- 183. Tiedge M, Lenzen S. Effects of sodium butyrate on glucose transporter and glucose-phosphorylating enzyme gene expression in RINm5F insulinoma cells. J Mol Endocrinol. 1996;17(1):19-26. doi:10.1677/jme.0.0170019

- 184. Khan S, Jena GB. Protective role of sodium butyrate, a HDAC inhibitor on betacell proliferation, function and glucose homeostasis through modulation of p38/ERK MAPK and apoptotic pathways: study in juvenile diabetic rat. *Chem Biol Interact*. 2014;213:1-12. doi:10.1016/j.cbi.2014.02.001
- 185. Chriett S, Zerzaihi O, Vidal H, et al. The histone deacetylase inhibitor sodium butyrate improves insulin signalling in palmitate-induced insulin resistance in L6 rat muscle cells through epigenetically-mediated up-regulation of Irs1. *Mol Cell Endocrinol*. 2017;439:224-32. doi:10.1016/j.mce.2016.09.006
- 186. Yan H, Ajuwon KM. Mechanism of butyrate stimulation of triglyceride storage and adipokine expression during adipogenic differentiation of porcine stromovascular cells. *PLoS ONE* 2015;10(12):e0145940. doi:10.1371/journal.pone.0145940
- 187. Mihaylova Maria M, Vasquez Debbie S, Ravnskjaer K, et al. Class IIa histone deacetylases are hormone-activated regulators of FOXO and mammalian glucose homeostasis. *Cell* 2011;145(4):607-21. doi:10.1016/j.cell.2011.03.043
- 188. Jove M, Serrano JC, Bellmunt MJ, et al. When cholesterol is not cholesterol: a note on the enzymatic determination of its concentration in model systems containing vegetable extracts. *Lipids Health Dis.* 2010;9:65. doi:10.1186/1476-511x-9-65
- 189. Rio DC, Ares M, Jr., Hannon GJ, et al. Purification of RNA using TRIzol (TRI reagent). Cold Spring Harb Protoc. 2010;2010(6):pdb.prot5439. doi:10.1101/pdb.prot5439
- 190. Chomczynski P, Mackey K. Substitution of chloroform by bromo-chloropropane in the single-step method of RNA isolation. *Anal Biochem.* 1995;225(1):163-4. doi:10.1006/abio.1995.1126
- 191. Ponchel F, Toomes C, Bransfield K, et al. Real-time PCR based on SYBR-Green I fluorescence: An alternative to the TaqMan assay for a relative quantification of gene rearrangements, gene amplifications and micro gene deletions. *BMC Biotech*. 2003;3(1):18. doi:10.1186/1472-6750-3-18
- 192. Rodrigues AC, Curi R, Genvigir FDV, et al. The expression of efflux and uptake transporters are regulated by statins in Caco-2 and HepG2 cells. Acta Pharmacol Sin. 2009;30(7):956-64. doi:10.1038/aps.2009.85

- 193. Chua SL, See Too WC, Khoo BY, et al. UBC and YWHAZ as suitable reference genes for accurate normalisation of gene expression using MCF7, HCT116 and HepG2 cell lines. *Cytotechnology* 2011;63(6):645-54. doi:10.1007/s10616-011-9383-4
- 194. Cicinnati VR, Shen Q, Sotiropoulos GC, et al. Validation of putative reference genes for gene expression studies in human hepatocellular carcinoma using real-time quantitative RT-PCR. *BMC Canc.* 2008;8:350-50. doi:10.1186/1471-2407-8-350
- 195. Brennan L, Shine A, Hewage C, et al. A nuclear magnetic resonance-based demonstration of substantial oxidative L-alanine metabolism and L-alanineenhanced glucose metabolism in a clonal pancreatic beta-cell line: metabolism of L-alanine is important to the regulation of insulin secretion. *Diabetes.* 2002;51(6):1714-21. doi:10.2337/diabetes.51.6.1714
- 196. Huang X, Pan Q, Lin Y, et al. A native chromatin immunoprecipitation (ChIP) protocol for studying histone modifications in strawberry fruits. *Plant Meth.* 2020;16(1):10. doi:10.1186/s13007-020-0556-z
- 197. RT-PCR primer design for ChIP Michigan: The University of Michigan; 2016 Available from: http://bridgeslab.sph.umich.edu/protocols/index.php/RT-PCR_primer_design_for_ChIP (accessed 30 July 2019.)
- 198. The Encode Project Consortium. An integrated encyclopedia of DNA elements in the human genome. *Nature*. 2012;489(7414):57-74. doi:10.1038/nature11247
- 199. Kent WJ, Sugnet CW, Furey TS, et al. The Human Genome Browser at UCSC. Genome Res. 2002;12(6):996-1006. doi:10.1101/gr.229102
- 200. Johansen ME, Green LA, Sen A, et al. Cardiovascular risk and statin use in the United States. Ann Fam Med. 2014;12(3):215-23. doi:10.1370/afm.1641
- 201. Matthews A, Herrett E, Gasparrini A, et al. Impact of statin related media coverage on use of statins: interrupted time series analysis with UK primary care data. *BMJ*. 2016;353 doi:10.1136/bmj.i3283
- 202. Björnsson ES. Hepatotoxicity of statins and other lipid-lowering agents. *Liver* Int. 2017;37(2):173-78. doi:10.1111/liv.13308
- 203. Yiew KH, Chatterjee TK, Hui DY, et al. Histone deacetylases and cardiometabolic diseases. *Arterioscler Thromb Vasc Biol.* 2015;35(9):1914-19. doi:10.1161/ATVBAHA.115.305046

- 204. Stancu C, Sima A. Statins: mechanism of action and effects. *J Cell Mol Med*. 2001;5(4):378-87. doi:10.1111/j.1582-4934.2001.tb00172.x
- 205. Cenedella RJ. Cholesterol synthesis inhibitor U18666A and the role of sterol metabolism and trafficking in numerous pathophysiological processes. *Lipids* 2009;44(6):477-87. doi:10.1007/s11745-009-3305-7
- 206. May P, Bock H, Nimpf J, et al. Differential glycosylation regulates processing of lipoprotein receptors by γ-secretase. J Biol Chem. 2003;278:37386-92. doi:10.1074/jbc.M305858200
- 207. Zager R, Shah V, Shah H, et al. The mevalonate pathway during acute tubular injury: selected determinants and consequences. *Am J Pathol.* 2002;161:681-92.
- 208. Ji A, Meyer JM, Cai L, et al. Scavenger receptor SR-BI in macrophage lipid metabolism. *Atherosclerosis* 2011;217(1):106-12. doi:10.1016/j.atherosclerosis.2011.03.017
- 209. Walker AK, Yang F, Jiang K, et al. Conserved role of SIRT1 orthologs in fasting-dependent inhibition of the lipid/cholesterol regulator SREBP. *Genes Dev*. 2010;24(13):1403-17. doi:10.1101/gad.1901210
- 210. Davie JR. Inhibition of histone deacetylase activity by butyrate. J Nutr. 2003;133(7):2485S-93S. doi:10.1093/jn/133.7.2485S
- 211. Mattace Raso G, Simeoli R, Russo R, et al. Effects of sodium butyrate and its synthetic amide derivative on liver inflammation and glucose tolerance in an animal model of steatosis induced by high fat diet. *PLoS ONE*. 2013;8(7):e68626. doi:10.1371/journal.pone.0068626
- 212. Hong J, Jia Y, Pan S, et al. Butyrate alleviates high fat diet-induced obesity through activation of adiponectin-mediated pathway and stimulation of mitochondrial function in the skeletal muscle of mice. *Oncotarget*. 2016;7(35):56071-82. doi:10.18632/oncotarget.11267
- 213. Yin W, Carballo-Jane E, McLaren DG, et al. Plasma lipid profiling across species for the identification of optimal animal models of human dyslipidemia. *J Lipid Res.* 2012;53(1):51-65. doi:10.1194/jlr.M019927
- 214. Jin CJ, Sellmann C, Engstler AJ, et al. Supplementation of sodium butyrate protects mice from the development of non-alcoholic steatohepatitis (NASH).
 Br J Nutr. 2015;114(11):1745-55. doi:10.1017/s0007114515003621

- 215. Zhou D, Pan Q, Xin F-Z, et al. Sodium butyrate attenuates high-fat diet-induced steatohepatitis in mice by improving gut microbiota and gastrointestinal barrier. *World J Gastroenterol.* 2017;23(1):60-75. doi:10.3748/wjg.v23.i1.60
- 216. Nagao K, Tomioka M, Ueda K. Function and regulation of ABCA1-membrane meso-domain organization and reorganization. *FEBS J.* 2011;278(18):3190-203.doi: 10.1111/j.1742-4658.2011.08170.x
- 217. Wong J, Quinn CM, Gelissen IC, et al. The effect of statins on ABCA1 and ABCG1 expression in human macrophages is influenced by cellular cholesterol levels and extent of differentiation. *Atherosclerosis*. 2008;196(1):180-89. doi:10.1016/j.atherosclerosis.2007.03.030
- 218. McTaggart F, Jones P. Effects of statins on high-density lipoproteins: a potential contribution to cardiovascular benefit. *Cardiovasc Drugs Ther.* 2008;22(4):321-38. doi:10.1007/s10557-008-6113-z
- 219. Li X, Zhang S, Blander G, et al. SIRT1 deacetylates and positively regulates the nuclear receptor LXR. *Mol Cell*. 2007;28(1):91-106. doi:10.1016/j.molcel.2007.07.032
- 220. Shen W-J, Azhar S, Kraemer FB. SR-B1: a unique multifunctional receptor for cholesterol influx and efflux. *Annu Rev Physiol.* 2018;80(1):95-116. doi:10.1146/annurev-physiol-021317-121550
- 221. Lopez D, McLean MP. Sterol regulatory element-binding protein-1a binds to cis elements in the promoter of the rat high density lipoprotein receptor SR-BI gene. *Endocrinology*. 1999;140(12):5669-81. doi:10.1210/endo.140.12.7220
- 222. Malerod L, Juvet LK, Hanssen-Bauer A, et al. Oxysterol-activated LXRalpha/RXR induces hSR-BI-promoter activity in hepatoma cells and preadipocytes. *Biochem Biophys Res Commun.* 2002;299(5):916-23.
- 223. Yu L, Cao G, Repa J, et al. Sterol regulation of scavenger receptor class B type I in macrophages. J Lipid Res. 2004;45(5):889-99. doi:10.1194/jlr.M300461-JLR200
- 224. Wang N, Arai T, Ji Y, et al. Liver-specific overexpression of scavenger receptor BI decreases levels of very low density lipoprotein ApoB, low density lipoprotein ApoB, and high density lipoprotein in transgenic mice. J Biol Chem. 1998;273(49):32920-6.

- 225. Stylianou IM, Svenson KL, VanOrman SK, et al. Novel ENU-induced point mutation in scavenger receptor class B, member 1, results in liver specific loss of SCARB1 protein. *PLoS ONE*. 2009;4(8):e6521. doi:10.1371/journal.pone.0006521
- 226. Obeidat M, Fishbane N, Nie Y, et al. The effect of statins on blood gene expression in COPD. *PLoS ONE*. 2015; 10(10). doi: 10.1371/journal.pone.0140022
- 227. Horton JD, Goldstein JL, Brown MS. SREBPs: activators of the complete program of cholesterol and fatty acid synthesis in the liver. J Clin Invest. 2002;109(9):1125-31. doi:10.1172/JCI15593
- 228. Llanos P, Contreras-Ferrat A, Osorio-Fuentealba C, et al. Cell cholesterol levels determine glucose uptake and insulin resistance in skeletal muscle. *FASEB J*. 2012;26(1_supplement):759.12-59.12.

doi:10.1096/fasebj.26.1_supplement.759.12

- 229. Christensen DP, Gysemans C, Lundh M, et al. Lysine deacetylase inhibition prevents diabetes by chromatin-independent immunoregulation and beta-cell protection. *Proc Natl Acad Sci USA*. 2014;111(3):1055-9. doi:10.1073/pnas.1320850111
- 230. Carlessi R, Chen Y, Rowlands J, et al. GLP-1 receptor signalling promotes β-cell glucose metabolism via mTOR-dependent HIF-1α activation. *Sci Rep.* 2017; 7(1). doi:10.1038/s41598-017-02838-2
- 231. Liu Z, Jeppesen PB, Gregersen S, et al. Dose- and glucose-dependent effects of amino acids on insulin secretion from isolated mouse islets and clonal INS1E beta-cells. *Rev Diabet Stud.* 2008;5(4):232-44. doi:10.1900/rds.2008.5.232
- 232. Stumvoll M, Goldstein BJ, van Haeften TW. Type 2 diabetes: principles of pathogenesis and therapy. *Lancet*. 2005;365(9467):1333-46. doi:10.1016/S0140-6736(05)61032-X
- 233. Green AD, Vasu S, Flatt PR. Cellular models for beta-cell function and diabetes gene therapy. Acta Physiol. 2018;222(3):e13012. doi:10.1111/apha.13012
- 234. Kruit JK, Wijesekara N, Westwell-Roper C, et al. Loss of both ABCA1 and ABCG1 results in increased disturbances in islet sterol homeostasis, inflammation, and impaired β-cell function. *Diabetes*. 2012;61(3):659-64. doi:10.2337/db11-1341

- 235. Yu Q, Chen Y, Xu C-B. Statins and new-onset diabetes mellitus: LDL receptor may provide a key link. *Front Pharmacol.* 2017;8:372-72. doi:10.3389/fphar.2017.00372
- 236. Rein-Fischboeck L, Krautbauer S, Eisinger K, et al. Hepatic scavenger receptor BI is associated with type 2 diabetes but unrelated to human and murine nonalcoholic fatty liver disease. *Biochem Biophys Res Commun.* 2015;467(2):377-82. doi:10.1016/j.bbrc.2015.09.149
- 237. Wamique M, Ali W, Himanshu D. Association of SRB1 and PON1 gene polymorphisms with type 2 diabetes mellitus: a case control study. Int J Diabetes Dev Ctries. 2020 doi:10.1007/s13410-019-00787-2
- 238. Kim H-i, Ahn Y-h. Role of Peroxisome Proliferator-Activated Receptor-γ in the glucose-sensing apparatus of liver and β-bells. *Diabetes*. 2004;53(S1):S60-S65. doi:10.2337/diabetes.53.2007.S60
- 239. Mauldin JP, Nagelin MH, Wojcik AJ, et al. Reduced expression of ATP-binding cassette transporter G1 increases cholesterol accumulation in macrophages of patients with type 2 diabetes mellitus. *Circulation*. 2008;117(21):2785-92. doi:10.1161/CIRCULATIONAHA.107.741314
- 240. Agius L. Targeting hepatic glucokinase in type 2 diabetes: weighing the benefits and risks. *Diabetes*. 2009;58(1):18-20. doi:10.2337/db08-1470
- 241. Swerdlow DI, Preiss D, Kuchenbaecker KB, et al. HMG-coenzyme A reductase inhibition, type 2 diabetes, and bodyweight: evidence from genetic analysis and randomised trials. *Lancet.* 2015;385(9965):351-61. doi:10.1016/s0140-6736(14)61183-1
- 242. Ruiz R, Jideonwo V, Ahn M, et al. Sterol regulatory element-binding protein-1 (SREBP-1) is required to regulate glycogen synthesis and gluconeogenic gene expression in mouse liver. J Biol Chem. 2014;289(9):5510-7. doi:10.1074/jbc.M113.541110
- 243. Shimano H, Amemiya-Kudo M, Takahashi A, et al. Sterol regulatory element– binding protein-1c and pancreatic β-cell dysfunction. *Diabetes Obes Metab*. 2007;9(s2):133-39. doi:10.1111/j.1463-1326.2007.00779.x
- 244. Rye KA, Barter PJ, Cochran BJ. Apolipoprotein A-I interactions with insulin secretion and production. *Curr Opin Lipidol.* 2016;27(1):8-13. doi:10.1097/mol.00000000000253

- 245. Wu X, Yu Z, Su W, et al. Low levels of ApoA1 improve risk prediction of type
 2 diabetes mellitus. J Clin Lipidol. 2017;11(2):362-68. doi:10.1016/j.jacl.2017.01.009
- 246. Li T, Francl JM, Boehme S, et al. Glucose and insulin induction of bile acid synthesis: mechanisms and implication in diabetes and obesity. *J Biol Chem.* 2012;287(3):1861-73. doi:10.1074/jbc.M111.305789
- 247. Thorens B. GLUT2, glucose sensing and glucose homeostasis. *Diabetologia*. 2015;58(2):221-32. doi:10.1007/s00125-014-3451-1
- 248. Strålfors P. Caveolins and caveolae, roles in insulin signalling and diabetes. In: Jasmin J-F, Frank PG, Lisanti MP, eds. *Caveolins and Caveolae: Roles in Signaling and Disease Mechanisms*. 2012:111-26. New York, NY: Springer US
- 249. Conti R, Mannucci E, Pessotto P, et al. Selective reversible inhibition of liver carnitine palmitoyl-transferase 1 by teglicar reduces gluconeogenesis and improves glucose homeostasis. *Diabetes*. 2011;60(2):644-51. doi:10.2337/db10-0346
- 250. Patel DC, Albrecht C, Pavitt D, et al. Type 2 diabetes is associated with reduced ATP-binding cassette transporter A1 gene expression, protein and function. *PLoS ONE*. 2011;6(7):e22142. doi:10.1371/journal.pone.0022142
- 251. Fujimoto K, Polonsky KS. Pdx1 and other factors that regulate pancreatic betacell survival. *Diabetes Obes Metab.* 2009;11(S4):30-37. doi:10.1111/j.1463-1326.2009.01121.x
- 252. Shalev A. Thioredoxin-interacting protein: regulation and function in the pancreatic beta-cell. *Mol Endocrinol*. 2014;28(8):1211-20. doi:10.1210/me.2014-1095
- 253. Arya VB, Rahman S, Senniappan S, et al. HNF4A mutation: switch from hyperinsulinaemic hypoglycaemia to maturity-onset diabetes of the young, and incretin response. *Diabet Med.* 2014;31(3):e11-e15. doi:10.1111/dme.12369
- 254. Fu Z, Gilbert ER, Liu D. Regulation of insulin synthesis and secretion and pancreatic Beta-cell dysfunction in diabetes. *Curr Diabetes Rev.* 2013;9(1):25-53.255. Matschinsky FM, Wilson DF. The central role of glucokinase in glucose homeostasis: a perspective 50 years after demonstrating the presence of the enzyme in islets of Langerhans. *Front Physiol.* 2019;10(148) doi:10.3389/fphys.2019.00148

- 256. Maiztegui B, Borelli MI, Massa ML, et al. Enhanced expression of hexokinase
 I in pancreatic islets induced by sucrose administration. *J Endocrinol*.
 2006;189(2):311-7. doi: 10.1677/joe.1.06662
- 257. Cockburn BN, Ostrega DM, Sturis J, et al. Changes in pancreatic islet glucokinase and hexokinase activities with increasing age, obesity, and the onset of diabetes. *Diabetes*. 1997;46(9):1434-9. doi:10.2337/diab.46.9.1434
- 258. Weir GC, Aguayo-Mazzucato C, Bonner-Weir S. β-cell dedifferentiation in diabetes is important, but what is it? *Islets*. 2013;5(5):233-7. doi:10.4161/isl.27494
- 259. Björklund A, Lansner A, Grill VE. Glucose-induced [Ca2+]i abnormalities in human pancreatic islets: important role of overstimulation. *Diabetes*, 2000;49(11):1840-48. doi:10.2337/diabetes.49.11.1840
- 260. Bjorklund A, Grill V. Enhancing effects of long-term elevated glucose and palmitate on stored and secreted proinsulin-to-insulin ratios in human pancreatic islets. *Diabetes*. 1999;48(7):1409-14. doi:10.2337/diabetes.48.7.1409
- 261. Robertson RP, Harmon J, Tran PO, et al. Glucose toxicity in β-cells: type 2 diabetes, good radicals gone bad, and the glutathione connection. *Diabetes*. 2003;52(3):581-87. doi:10.2337/diabetes.52.3.581
- 262. Dhawan S, Tschen S-I, Zeng C, et al. DNA methylation directs functional maturation of pancreatic β cells. J Clin Invest. 2015;125(7):2851-60. doi:10.1172/JCI79956
- 263. Thorrez L, Laudadio I, Van Deun K, et al. Tissue-specific disallowance of housekeeping genes: the other face of cell differentiation. *Genome Res.* 2011;21(1):95-105. doi: 10.1101/gr.109173.110
- 264. Yamato E. High dose of histone deacetylase inhibitors affects insulin secretory mechanism of pancreatic beta cell line. *Endocr Regul.* 2018;52(1):21. doi:10.2478/enr-2018-0004
- 265. Bompada P, Atac D, Luan C, et al. Histone acetylation of glucose-induced thioredoxin-interacting protein gene expression in pancreatic islets. *Int J Biochem Cell Biol.* 2016;81(Pt A):82-91. doi:10.1016/j.biocel.2016.10.022
- 266. Feingold PL, Surman DR, Brown K, et al. Induction of thioredoxin-interacting protein by a histone deacetylase inhibitor, Entinostat, is associated with DNA damage and apoptosis in esophageal adenocarcinoma. *Mol Cancer Ther*. 2018;17(9):2013-23. doi: 10.1158/1535-7163.mct-17-1240

- 267. Perrone L, Devi TS, Hosoya KI, et al. Inhibition of TXNIP expression in vivo blocks early pathologies of diabetic retinopathy. *Cell Death Dis.* 2010;1(8):e65-e65. doi:10.1038/cddis.2010.42
- 268. Elgort MG, O'Shea JM, Jiang Y, et al. Transcriptional and translational downregulation of thioredoxin interacting protein is required for metabolic reprogramming during G(1). *Genes Cancer*. 2010;1(9):893-907. doi:10.1177/1947601910389604
- 269. Lin HV, Frassetto A, Kowalik EJ, Jr., et al. Butyrate and propionate protect against diet-induced obesity and regulate gut hormones via free fatty acid receptor 3-independent mechanisms. *PLoS ONE*. 2012;7(4):e35240-e40. doi:10.1371/journal.pone.0035240
- 270. Donohoe DR, Garge N, Zhang X, et al. The microbiome and butyrate regulate energy metabolism and autophagy in the mammalian colon. *Cell Metab.* 2011;13(5):517-26. doi:10.1016/j.cmet.2011.02.018
- 271. Traisaeng S, Batsukh A, Chuang T-H, et al. *Leuconostoc mesenteroides* fermentation produces butyric acid and mediates Ffar2 to regulate blood glucose and insulin in type 1 diabetic mice. *Sci Rep.* 2020;10 doi:10.1038/s41598-020-64916-2
- 272. Pingitore A, Gonzalez-Abuin N, Ruz-Maldonado I, et al. Short chain fatty acids stimulate insulin secretion and reduce apoptosis in mouse and human islets in vitro: Role of free fatty acid receptor 2. *Diabetes Obes Metab.* 2019;21(2):330-39. doi:10.1111/dom.13529
- 273. Priyadarshini M, Villa SR, Fuller M, et al. An acetate-specific GPCR, FFAR2, regulates insulin secretion. *Mol Endocrinol.* 2015;29(7):1055-66. doi:10.1210/me.2015-1007
- 274. Priyadarshini M, Layden BT. FFAR3 modulates insulin secretion and global gene expression in mouse islets. *Islets.* 2015;7(2):e1045182. doi:10.1080/19382014.2015.1045182
- 275. Tang C, Ahmed K, Gille A, et al. Loss of FFA2 and FFA3 increases insulin secretion and improves glucose tolerance in type 2 diabetes. *Nature Med.* 2015;21(2):173-77. doi:10.1038/nm.3779
- 276. Damaskos C, Valsami S, Kontos M, et al. Histone deacetylase inhibitors: an attractive therapeutic strategy against breast cancer. *Anticancer Res.* 2017;37(1):35-46.

- 277. van der Beek CM, Bloemen JG, van den Broek MA, et al. Hepatic uptake of rectally administered butyrate prevents an increase in systemic butyrate concentrations in humans. J Nutr. 2015;145(9):2019-24. doi:10.3945/jn.115.211193
- 278. Miller AA, Kurschel E, Osieka R, et al. Clinical pharmacology of sodium butyrate in patients with acute leukemia. *Eur J Cancer Clin Oncol* 1987;23(9):1283-7. doi:10.1016/0277-5379(87)90109-x
- 279. Li H-P, Chen X, Li M-Q. Butyrate alleviates metabolic impairments and protects pancreatic β cell function in pregnant mice with obesity. *Int J Clin Exp Pathol.* 2013;6(8):1574-84.
- 280. Christiansen CB, Gabe MBN, Svendsen B, et al. The impact of short-chain fatty acids on GLP-1 and PYY secretion from the isolated perfused rat colon. Am J Physiol Gastrointest Liver Physiol. 2018;315(1):G53-G65. doi:10.1152/ajpgi.00346.2017
- 281. Yadav H, Lee JH, Lloyd J, et al. Beneficial metabolic effects of a probiotic via butyrate-induced GLP-1 hormone secretion. J Biol Chem. 2013;288(35):25088-97. doi:10.1074/jbc.M113.452516
- 282. Rowlands J, Heng J, Newsholme P, et al. Pleiotropic effects of GLP-1 and analogs on cell signaling, metabolism, and function. *Front Endocrinol.* 2018;9:672-72. doi:10.3389/fendo.2018.00672
- 283. Fajas L, Schoonjans K, Gelman L, et al. Regulation of peroxisome proliferatoractivated receptor gamma expression by adipocyte differentiation and determination factor 1/sterol regulatory element binding protein 1: implications for adipocyte differentiation and metabolism. *Mol Cell Biol.* 1999;19(8):5495-503. doi:10.1128/mcb.19.8.5495
- 284. López CA, de Vries AH, Marrink SJ. Molecular mechanism of cyclodextrin mediated cholesterol extraction. *PLoS Comput Biol.* 2011;7(3):e1002020. doi:10.1371/journal.pcbi.1002020
- 285. Kruit JK, Kremer PH, Dai L, et al. Cholesterol efflux via ATP-binding cassette transporter A1 (ABCA1) and cholesterol uptake via the LDL receptor influences cholesterol-induced impairment of beta cell function in mice. *Diabetologia.* 2010;53(6):1110-9. doi:10.1007/s00125-010-1691-2
- 286. Kruit JK, Wijesekara N, Fox JE, et al. Islet cholesterol accumulation due to loss of ABCA1 leads to impaired exocytosis of insulin granules. *Diabetes*. 2011;60(12):3186-96. doi:10.2337/db11-0081

- 287. Brunham LR, Kruit JK, Pape TD, et al. Beta-cell ABCA1 influences insulin secretion, glucose homeostasis and response to thiazolidinedione treatment. *Nat Med.* 2007;13(3):340-7. doi:10.1038/nm1546
- 288. Hussain SS, Harris MT, Kreutzberger AJB, et al. Control of insulin granule formation and function by the ABC transporters ABCG1 and ABCA1 and by oxysterol binding protein OSBP. *Mol Biol Cell*. 2018;29(10):1238-57. doi:10.1091/mbc.E17-08-0519
- 289. Churchward MA, Rogasevskaia T, Höfgen J, et al. Cholesterol facilitates the native mechanism of Ca2+-triggered membrane fusion. J Cell Sci. 2005;118(20):4833-48. doi:10.1242/jcs.02601
- 290. Lang T, Bruns D, Wenzel D, et al. SNAREs are concentrated in cholesteroldependent clusters that define docking and fusion sites for exocytosis. *Embo* J. 2001;20(9):2202-13. doi: 10.1093/emboj/20.9.2202
- 291. Sandoval-Hernández AG, Hernández HG, Restrepo A, et al. Liver X Receptor agonist modifies the DNA methylation profile of synapse and neurogenesisrelated genes in the triple transgenic mouse model of Alzheimer's Disease. J Mol Neurosci. 2016;58(2):243-53. doi:10.1007/s12031-015-0665-8
- 292. Tristán-Flores FE, Guzmán P, Ortega-Kermedy MS, et al. Liver X Receptorbinding DNA motif associated with atherosclerosis-specific DNA methylation profiles of Alu elements and neighboring CpG islands. J Am Heart Assoc. 2018;7(3):e007686. doi:10.1161/JAHA.117.007686
- 293. Khamis A, Boutry R, Canouil M, et al. Histone deacetylase 9 promoter hypomethylation associated with adipocyte dysfunction is a statin-related metabolic effect. *bioRxiv*. 2019:849695. doi:10.1101/849695
- 294. Cao Q, Rong S, Repa JJ, et al. Histone deacetylase 9 represses cholesterol efflux and alternatively activated macrophages in atherosclerosis development. *Arterioscler Thromb Vasc Biol.* 2014;34(9):1871-79. doi:10.1161/ATVBAHA.114.303393
- 295. Moutzouri E, Liberopoulos E, Mikhailidis DP, et al. Comparison of the effects of simvastatin vs. rosuvastatin vs. simvastatin/ezetimibe on parameters of insulin resistance. *Int J Clin Pract.* 2011;65(11):1141-8. doi:10.1111/j.1742-1241.2011.02779.x

- 296. Kostapanos MS, Milionis HJ, Agouridis AD, et al. Rosuvastatin treatment is associated with an increase in insulin resistance in hyperlipidaemic patients with impaired fasting glucose. *Int J Clin Pract.* 2009;63(9):1308-13. doi:10.1111/j.1742-1241.2009.02101.x
- 297. Szendroedi J, Anderwald C, Krssak M, et al. Effects of high-dose simvastatin therapy on glucose metabolism and ectopic lipid deposition in nonobese type
 2 diabetic patients. *Diabetes Care*. 2009;32(2):209-14. doi:10.2337/dc08-1123
- 298. Bellia A, Rizza S, Galli A, et al. Early vascular and metabolic effects of rosuvastatin compared with simvastatin in patients with type 2 diabetes. *Atherosclerosis.* 2010;210(1):199-201. doi:10.1016/j.atherosclerosis.2009.11.021
- 299. Paolisso G, Barbagallo M, Petrella G, et al. Effects of simvastatin and atorvastatin administration on insulin resistance and respiratory quotient in aged dyslipidemic non-insulin dependent diabetic patients. *Atherosclerosis*. 2000;150(1):121-7. doi:10.1016/s0021-9150(99)00352-4
- 300. Sugiyama S, Fukushima H, Kugiyama K, et al. Pravastatin improved glucose metabolism associated with increasing plasma adiponectin in patients with impaired glucose tolerance and coronary artery disease. *Atherosclerosis*. 2007;194(2):e43-51. doi:10.1016/j.atherosclerosis.2006.08.023
- 301. Song G, Ouyang G, Bao S. The activation of Akt/PKB signaling pathway and cell survival. J Cell Mol Med. 2005;9(1):59-71. doi:10.1111/j.1582-4934.2005.tb00337.x
- 302. Kaiser C, James SR. Acetylation of insulin receptor substrate-1 is permissive for tyrosine phosphorylation. *BMC Biol.* 2004;2:23-23. doi:10.1186/1741-7007-2-23
- 303. Sundaresan NR, Pillai VB, Wolfgeher D, et al. The deacetylase SIRT1 promotes membrane localization and activation of Akt and PDK1 during tumorigenesis and cardiac hypertrophy. *Sci Signal.* 2011;4(182):ra46. doi:10.1126/scisignal.2001465
- 304. Pirola L, Zerzaihi O, Vidal H, et al. Protein acetylation mechanisms in the regulation of insulin and insulin-like growth factor 1 signalling. *Mol Cell Endocrinol.* 2012;362(1):1-10. doi:10.1016/j.mce.2012.05.011
- 305. Zhang W-Q, Zhao T-T, Gui D-K, et al. Sodium butyrate improves liver glycogen metabolism in type 2 diabetes mellitus. J Agric Food Chem. 2019;67(27):7694-705. doi:10.1021/acs.jafc.9b02083
- 306. Chen J-B, Chern T-R, Wei T-T, et al. Design and synthesis of dual-action inhibitors targeting histone deacetylases and 3-Hydroxy-3-methylglutaryl coenzyme A reductase for cancer treatment. J Med Chem. 2013;56(9):3645-55. doi:10.1021/jm400179b.
- 307. Singh RS, Chaudhary DK, Mohan A, et al. Greater efficacy of atorvastatin versus a non-statin lipid-lowering agent against renal injury: potential role as a histone deacetylase inhibitor. *Sci Rep.* 2016;6:38034. doi:10.1038/srep38034
- 308. Dhar S, Gursoy-Yuzugullu O, Parasuram R, et al. The tale of a tail: histone H4 acetylation and the repair of DNA breaks. *Philos Trans R Soc Lond B Biol Sci.* 2017;372(1731) doi:10.1098/rstb.2016.0284
- 309. Karmodiya K, Krebs AR, Oulad-Abdelghani M, et al. H3K9 and H3K14 acetylation co-occur at many gene regulatory elements, while H3K14ac marks a subset of inactive inducible promoters in mouse embryonic stem cells. *BMC Genomics*. 2012;13(1):424. doi:10.1186/1471-2164-13-424
- 310. Nakayama J-i, Rice JC, Strahl BD, et al. Role of histone H3 lysine 9 methylation in epigenetic control of heterochromatin assembly. *Science*. 2001;292(5514):110-13. doi:10.1126/science.1060118
- 311. Meier BC, Wagner BK. Inhibition of HDAC3 as a strategy for developing novel diabetes therapeutics. *Epigenomics*. 2014;6(2):209-14. doi:10.2217/epi.14.11
- 312. Brueckner B, Garcia Boy R, Siedlecki P, et al. Epigenetic reactivation of tumor suppressor genes by a novel small-molecule inhibitor of human DNA methyltransferases. *Cancer Res.* 2005;65(14):6305-11. doi:10.1158/0008-5472.can-04-2957
- 313. Haring M, Offermann S, Danker T, et al. Chromatin immunoprecipitation: optimization, quantitative analysis and data normalization. *Plant Meth.* 2007;3:11. doi:10.1186/1746-4811-3-11
- 314. Gupta SC, Patchva S, Aggarwal BB. Therapeutic roles of curcumin: lessons learned from clinical trials. AAPS J. 2013;15(1):195-218. doi:10.1208/s12248-012-9432-8
- 315. Anand P, Kunnumakkara AB, Newman RA, et al. Bioavailability of curcumin: problems and promises. *Mol Pharm.* 2007;4(6):807-18. doi:10.1021/mp700113r

- 316. Björkhem-Bergman L, Lindh JD, Bergman P. What is a relevant statin concentration in cell experiments claiming pleiotropic effects? Br J Clin Pharmacol. 2011;72(1):164-65. doi:10.1111/j.1365-2125.2011.03907.x
- 317. Sodji QH, Kornacki JR, Mrksich M, et al. Chapter 15 In vitro histone deacetylase activity screening: making a case for better assays. In: Zheng YG, ed. *Epigenetic Technological Applications*. 2015:319-32. Boston: Academic Press
- 318. Skottheim IB, Gedde-Dahl A, Hejazifar S, et al. Statin induced myotoxicity: the lactone forms are more potent than the acid forms in human skeletal muscle cells in vitro. *Eur J Pharm Sci.* 2008;33(4-5):317-25. doi:10.1016/j.ejps.2007.12.009
- 319. Olsson AG, McTaggart F, Raza A. Rosuvastatin: a highly effective new HMG-CoA reductase inhibitor. *Cardiovasc Drug Rev.* 2002;20(4):303-28. doi:10.1111/j.1527-3466.2002.tb00099.x
- 320. Sakaeda T, Fujino H, Komoto C, et al. Effects of acid and lactone forms of eight HMG-CoA reductase inhibitors on CYP-mediated metabolism and MDR1mediated transport. *Pharm Res.* 2006;23(3):506-12. doi:10.1007/s11095-005-9371-5
- 321. Taha DA, De Moor CH, Barrett DA, et al. The role of acid-base imbalance in statin-induced myotoxicity. *Transl Res.* 2016;174:140-60.e14. doi:10.1016/j.trsl.2016.03.015
- 322. Oliner JD, Andresen JM, Hansen SK, et al. SREBP transcriptional activity is mediated through an interaction with the CREB-binding protein. *Genes Dev*. 1996;10(22):2903-11. doi:10.1101/gad.10.22.2903
- 323. Lacazette E. A laboratory practical illustrating the use of the ChIP-qPCR method in a robust model: Estrogen receptor alpha immunoprecipitation using Mcf-7 culture cells. *Biochem Mol Biol Educ*. 2017;45(2):152-60. doi:10.1002/bmb.20999
- 324. Yamaguchi N, Winter CM, Wu MF, et al. Protocols: chromatin immunoprecipitation from Arabidopsis tissues. Arabidopsis Book. 2014;12:e0170. doi:10.1199/tab.0170
- 325. Chen Y, Bravo JI, Son JM, et al. Remodeling of the H3 nucleosomal landscape during mouse aging. *Trans Med Aging*. 2020;4:22-31. doi:10.1016/j.tma.2019.12.003

- 326. Yan F, Powell DR, Curtis DJ, et al. From reads to insight: a hitchhiker's guide to ATAC-seq data analysis. *Genome Biol.* 2020;21(1):22. doi:10.1186/s13059-020-1929-3
- 327. Mangian HF, Tappenden KA. Butyrate increases GLUT2 mRNA abundance by initiating transcription in Caco2-BBe cells. JPEN J Parenter Enteral Nutr. 2009;33(6):607-17. doi:10.1177/0148607109336599
- 328. Mangian H, Mangian H, Tappenden K. Butyrate-induced upregulation of intestinal glucose transport represents a possible nutrient therapy for individuals with malabsorptive disorders (1035.14). FASEB J. 2014;28(S1):1035.14. doi:10.1096/fasebj.28.1_supplement.1035.14
- 329. Lasko LM, Jakob CG, Edalji RP, et al. Discovery of a selective catalytic p300/CBP inhibitor that targets lineage-specific tumours. *Nature*. 2017;550(7674):128-32. doi:10.1038/nature24028
- 330. Cao G, Liang Y, Broderick CL, et al. Antidiabetic action of a Liver X Receptor agonist mediated by inhibition of hepatic gluconeogenesis. J Biol Chem. 2003;278(2):1131-36. doi:10.1074/jbc.M210208200
- 331. Patnaik A, Rowinsky EK, Villalona MA, et al. A phase I study of pivaloyloxymethyl butyrate, a prodrug of the differentiating agent butyric acid, in patients with advanced solid malignancies. *Clin Cancer Res.* 2002;8(7):2142-48.
- 332. Conley BA, Egorin MJ, Tait N, et al. Phase I study of the orally administered butyrate prodrug, tributyrin, in patients with solid tumors. *Clin Cancer Res.* 1998;4(3):629-34.
- 333. Edelman MJ, Bauer K, Khanwani S, et al. Clinical and pharmacologic study of tributyrin: an oral butyrate prodrug. *Cancer Chemother Pharmacol.* 2003;51(5):439-44. doi:10.1007/s00280-003-0580-5
- 334. Xie N, Zhou Y, Sun Q, et al. Novel epigenetic techniques provided by the CRISPR/Cas9 system. Stem Cells Int. 2018;2018:7834175-75. doi:10.1155/2018/7834175
- 335. Hilton IB, D'Ippolito AM, Vockley CM, et al. Epigenome editing by a CRISPR-Cas9-based acetyltransferase activates genes from promoters and enhancers. *Nature Biotech.* 2015;33(5):510-17. doi:10.1038/nbt.3199
- 336. McDonald JI, Celik H, Rois LE, et al. Reprogrammable CRISPR/Cas9-based system for inducing site-specific DNA methylation. *Biol Open*. 2016;5(6):866-74. doi:10.1242/bio.019067

- 337. Liao H-K, Hatanaka F, Araoka T, et al. In vivo target gene activation via CRISPR/Cas9-mediated trans-epigenetic modulation. *Cell.* 2017;171(7):1495-507.e15. doi:10.1016/j.cell.2017.10.025
- 338. McClenaghan NH, Barnett CR, Ah-Sing E, et al. Characterization of a novel glucose-responsive insulin-secreting cell line, BRIN-BD11, produced by electrofusion. *Diabetes*. 1996;45(8):1132-40. doi:10.2337/diab.45.8.1132
- 339. Pinal-Fernandez I, Casal-Dominguez M, Mammen AL. Statins: pros and cons. Med Clin (Barc). 2018;150(10):398-402. doi:10.1016/j.medcli.2017.11.030

Every reasonable effort has been made to acknowledge the owners of copyright material. I would be pleased to hear from any copyright owner who has been omitted or incorrectly acknowledged

Appendix A: Supplementary figures



Supplementary Figure S1Comparison of native and cross-linked ChIPqPCR of ChIP performed on HepG2 cells with and without formaldehyde cross-linking in proximity to GAPDH. Results are from a single experiment.



Supplementary Figure S2 Influence of butyrate and atorvastatin on SREBP-2 pathway in BRIN-BD11 cells

Relative protein levels of A) LDLR and B) in BRIN-BD11 cells following treatment with 1 mM or 5 mM sodium butyrate or 10 μ M atorvastatin for 24 h. The graph represents the mean combined density readings as normalised to GAPDH from at least three independent experiments, with error bars representing SEMs. ** p<0.01.



Supplementary Figure S3 Influence of butyrate and atorvastatin on ABCA1 in THP-1 cells

Relative protein levels of ABCA1 in THP-1 cells following treatment with 5 mM sodium butyrate or 10 μ M atorvastatin for 24 h. The graph represents the mean combined density readings as normalised to GAPDH from three independent experiments with error bars representing SEMs. ** p<0.01.



Supplementary Figure S4 Influence of butyrate, atorvastatin and curcumin on global histone modifications in BRIN-BD11 cells

Relative protein levels of a) acetylated H3K9, b) acetylated H4, c) trimethylated H3K9 and d) HDAC3. The graph represents the mean combined density readings as normalised to GAPDH from three independent experiments with error bars representing SEMs. ** p<0.01.



Supplementary Figure S5 Influence of butyrate and atorvastatin on global histone acetylation in HLPS

Relative protein levels of acetylated H3K9. The graph represents the mean combined density readings as normalised to GAPDH from three independent experiments with error bars representing SEMs. *** p<0.001.



Supplementary Figure S6 Influence of atorvastatin on DNMT activity in BRIN-BD11 cell extracts

a) DNMT activity in whole cell extracts treated directly with 200 μ M atorvastatin, RG108, curcumin or DMSO vehicle control. b) DNMT activity in whole cell extracts from cells treated for 24 h with 10 μ M atorvastatin, curcumin or DMSO vehicle control. Results represent the mean combined from two independent experiments, with error bars representing SEMs. ** *p*<0.01.





ChIP qPCR H3 signals from HepG2 cells treated with 10 μ M atorvastatin or DMSO vehicle control for 24 h in media containing FBS, normalised to nucleosome density (H3) or to positive control sequence (GAPDH). Results represent the mean combined from three independent experiments, with error bars representing SEMs.



Supplementary Figure S8 Influence of atorvastatin on H3 signal in ChIPqPCR

ChIP qPCR H3 signals from HepG2 cells treated with 10 μ M atorvastatin or DMSO vehicle control for 24 h in media containing FBS. Results represent the mean combined from three independent experiments, with error bars representing SEMs.



Supplementary Figure S9 ChIP qPCR of butyrate treated cells as percentage of input

ChIP qPCR from HepG2 cells treated with 5 mM butyrate for 24 h in media containing LPDS presented as percentage of total chromatin input in proximity to a) *GAPDH*, b) *ABCA1*, c) *ABCG1*, d) *HK1*, e) *LDLR* f) *PPARG*, g) *HMGCR* and h) *SLC2A2* (GLUT2). Results represent the mean combined from four independent experiments, with error bars representing SEMs. * p<0.05, ** p<0.01, *** p<0.001.



Supplementary Figure S10 ChIP qPCR of butyrate treated cells as raw data

ChIP qPCR from HepG2 cells treated with 5 mM butyrate for 24 h in media containing LPDS as raw data in proximity to a) *GAPDH*, b) *ABCA1*, c) *ABCG1*, d) *HK1*, e) *LDLR* f) *PPARG*, g) *HMGCR* and h) *SLC2A2* (GLUT2). Results are mean combined from four independent experiments + SEM. * p<0.05, ** p<0.01, *** p<0.001.



Supplementary Figure S11 ChIP qPCR of butyrate treated cells as modified histone:H3 ratio

ChIP qPCR from HepG2 cells treated with 5 mM butyrate for 24 h in media containing LPDS presented as a modified histone: total H3 ratio of ChIP qPCR signals in proximity to a) *GAPDH*, b) *ABCA1*, c) *ABCG1*, d) *HK1*, e) *LDLR* and f) *PPARG*. Results are mean combined from four independent experiments + SEM.



Supplementary Figure S12 ChIP qPCR of butyrate treated cells normalised to average results

ChIP qPCR from HepG2 cells treated with 5 mM butyrate for 24 h in media containing LPDS normalised to average results from all genomic regions examined, in proximity to a) *GAPDH*, b) *ABCA1*, c) *ABCG1*, d) *HK1*, e) *LDLR* and f) *PPARG*. Results are mean combined from four independent experiments + SEM.



Supplementary Figure S13 ChIP qPCR of butyrate treated cells normalised to GAPDH

ChIP qPCR from HepG2 cells treated with 5 mM butyrate for 24 h in media containing LPDS normalised to *GAPDH* in proximity to a) *SLC2A2* (GLUT2), b) *ABCA1*, c) *ABCG1*, d) *HK1*, e) *LDLR* f) *PPARG* and g) *HMGCR*. Results are mean combined from four independent experiments + SEM.



Supplementary Figure S14 Influence of butyrate on global H3

Relative protein levels of H3 in HepG2 cells treated with 5 mM sodium butyrate for 24 h. The graph represents the mean combined density readings as normalised to GAPDH from three independent experiments with error bars representing SEMs.

Appendix B: qPCR optimisation

Human primers for RT-qPCR of HepG2 cells

GAPDH



GAPDH	59.4	56.9	55	53.7
1	17.58	18.00	18.30	18.42
0.3	19.32	19.89	20.27	20.68
0.1	21.14	21.61	21.94	22.23
0.03	22.84	23.24	23.76	24.00
0.01	24.34	25.01	25.21	25.31
Efficiency	0.964113	0.9402481	0.9450484	0.9599434
R2	0.9989	0.9994	0.9987	0.9937

YWHAZ



YWHAZ	61.4	59	57	55.7
1	20.36	20.32	20.27	20.26
0.3	22.53	22.25	22.14	22.05
0.1	23.90	24.00	23.72	24.19
0.03	25.71	25.73	25.55	25.59
0.01	27.09	26.86	27.03	27.19
Efficiency	0.9964699	1.003618	0.974712	0.9376052
R2	0.9957	0.9938	0.9995	0.9943

RPL13A



RPL13A	61.4	59	57	55.7
1	16.40	16.41	16.94	16.83
0.3	18.44	18.71	18.85	19.04
0.1	20.14	20.31	20.56	21.01
0.03	21.99	22.25	22.38	22.67
0.01	23.13	23.59	24.59	24.05
Efficiency	0.9672967	0.9013577	0.8436412	0.8908657
R2	0.9937	0.995	0.9973	0.9926





LDLR	61.4	59	57	55.7
1	22.75	23.05	23.48	23.38
0.3	24.38	24.86	25.53	25.54
0.1	26.14	27.00	27.28	27.54
0.03	27.97	28.50	29.24	29.22
0.01	29.55	30.05	30.92	31.29
Efficiency	0.954899	0.9209507	0.8580468	0.8042045
R2	0.9994	0.9951	0.9997	0.998

PPARG

200nM



PPARG	61.4	59	57	55.7
1	25.14	24.97	25.14	25.13
0.3	26.41	26.80	26.53	26.60
0.1	28.47	28.42	28.42	28.42
0.03	30.12	30.22	30.07	29.89
0.01	31.56	31.44	31.83	31.50
Efficiency	1.005432	1.0199418	0.9751408	1.050543
R2	0.9943	0.9973	0.9969	0.9981

100nM



PPARG	63.4	61	59	57.7
1	26.8	25.66	24.74	
0.3	28.04	26.93	26.74	26.77
0.1	30.23	29.34	29.01	29.05
0.03	31.88	30.91	30.53	31.08
0.01		33.59	33.04	32.88
Efficiency	0.950548	0.788347	0.760757	0.750344
R2	0.9865	0.9841	0.9942	0.9968

SREBF2



SREBF2	61.4	59	57	55.7
1	21.88	21.81	22.19	22.21
0.3	23.73	24.06	24.33	23.86
0.1	25.43	25.81	26.54	26.77
0.03	27.34	27.72	28.24	28.90
0.01	29.04	29.63	30.43	30.33
Efficiency	0.8993008	0.8159885	0.7592014	0.7179961
R2	1.0000	0.9991	0.9976	0.9872

NR1H3



NR1H3	61.4	59	57	55.7
1	24.72	24.69	25.33	25.22
0.3	26.25	26.25	26.68	26.91
0.1	27.95	28.28	28.41	28.35
0.03	29.69	30.23	30.25	30.71
0.01	31.39	31.98	31.90	32.24
Efficiency	0.9857055	0.859746	0.9920203	0.9058429
R2	0.9989	0.9977	0.9968	0.9953

APOB

200nM



APOB	61.4	59	57	55.7
1	16.39	16.37	16.42	16.32
0.3	18.35	18.28	18.39	18.74
0.1	20.21	20.11	20.46	20.56
0.03	22.18	22.19	22.39	22.48
0.01	23.74	23.75	23.97	24.06
Efficiency	0.8609	0.8516278	0.8275951	0.8193501
R2	0.9992	0.9992	0.9982	0.996

100nM



АРОВ	63.4	61	59	57.7
1	19.66	18.47	18.36	18.63
0.3	21.38	20.77	20.83	21.36
0.1	23.89	23.49	23.4	23.51
0.03	26.3	25.68	26.25	26.09
0.01	29.19	28.98	28.54	28.55
Efficiency	0.617119	0.559558	0.562846	0.597986
R2	0.991	0.9936	0.9994	0.9995





APOA1	62	58.4	55.5	53.5
1	16.57	16.68	16.70	16.78
0.3	18.28	18.45	18.49	18.76
0.1	20.22	20.08	20.34	20.34
0.03	22.03	21.85	22.12	22.23
0.01	23.57	23.49	23.79	23.96
Efficiency	0.913027	0.9663537	0.909281	0.9068332
R2	0.9987	1	0.9996	0.9997

GK



GK	61.4	59	57	55.7
1	20.82	20.94	20.99	20.76
0.3	22.55	22.62	22.60	22.79
0.1	24.36	24.25	24.31	24.43
0.03	26.25	26.31	26.35	26.61
0.01	27.69	27.68	28.10	28.05
Efficiency	0.9350129	0.9543705	0.8980633	0.8682931
R2	0.9988	0.9981	0.9982	0.9981

HMGCR



HMGCR	61.4	59	57	55.7
1	20.98	21.00	20.51	20.32
0.3	22.71	22.91	22.52	22.58
0.1	24.58	24.55	24.44	24.32
0.03	26.88	26.71	26.66	26.50
0.01	28.95	29.02	28.68	28.63
Efficiency	0.773543	0.7865357	0.7552778	0.7524436
R2	0.9966	0.9953	0.9996	0.9991

PEPCK1



PEPCK1	64.4	62	60	58.7
1	24.93	24.57	24.27	24.61
0.3	26.89	26.42	26.28	26.82
0.1	28.31	28.19	28.29	28.47
0.03	30.55	29.93	30.26	30.31
0.01	32.39	32.07	31.80	31.65
Efficiency	0.8578761	0.8631919	0.8304793	0.9247788
R2	0.9973	0.9979	0.9982	0.9954





SLC2A2	61.4	59	57	55.7
1	30.11	29.75	29.67	29.93
0.3	32.42	31.55	31.59	31.4
0.1	34.68	32.87	33.05	33.03
0.03	35.26	35.4	36.4	35.2
0.01		35.77		37.78
Efficiency	0.929901	1.060796	0.70843	0.805656
R2	0.9434	0.9721	0.9708	0.984

SREBF1



SREBF1	61.4	59	57	55.7
1	23.65	23.89	24.15	24.21
0.3	25.53	25.50	26.01	26.36
0.1	27.30	27.22	27.39	28.10
0.03	28.99	29.47	29.52	30.17
0.01	30.56	30.98	31.10	31.44
Efficiency	0.9468721	0.8850724	0.9365492	0.876142
R2	0.9991	0.9968	0.9979	0.9956





CAV1	61.4	59	57	55.7
1	27.99	27.68	27.08	27.28
0.3	29.89	29.72	29.76	30.62
0.1	30.85	31.10	31.52	31.16
0.03	33.04	32.35	34.16	34.66
0.01	34.82	33.52	34.21	35.10
Efficiency	0.9820708	1.2337783	0.8501961	0.7907112
R2	0.9907	0.9879	0.9495	0.9454

CYP7A1



CYP7A1	62.4	60	58	56.7
1	29.62	29.81	29.85	29.85
0.3	31.45	31.65	31.16	31.89
0.1	33.44	33.49	33.41	33.24
0.03	37.50	35.33	34.84	36.16
0.01	36.03		36.90	
Efficiency	0.8362526	0.8823473	0.9127982	0.7720439
R2	0.8624	0.9997	0.9916	0.9822

FOX01



FOXO1	61.4	59	57	55.7
1	22.30	22.38	22.35	22.43
0.3	23.92	24.11	24.35	24.31
0.1	25.89	26.02	25.86	26.03
0.03	27.30	27.44	27.30	27.52
0.01	29.00	29.02	28.73	29.12
Efficiency	0.9868355	1.0001033	1.0819479	1.0014623
R2	0.9971	0.997	0.9957	0.9982

INSR



INSR	64.5	60.9	58	56
1	24.65	24.26	24.65	25.19
0.3	26.20	26.02	26.75	27.73
0.1	28.00	27.53	28.41	29.06
0.03	29.73	29.42	30.62	31.27
0.01	31.51	31.44	31.99	33.17
Efficiency	0.9499055	0.9122157	0.8603435	0.8044025
R2	0.9986	0.9971	0.9972	0.9954



ABCG1	64.4	62	60	58.7
1	30.12	29.14	29.50	29.39
0.3	31.56	31.03	30.83	31.09
0.1	33.26	32.41	32.31	32.42
0.03	34.27	34.89	34.48	33.77
0.01	37.84	34.96	36.84	34.89
Efficiency	0.8889184	1.0979664	0.8753865	1.3194603
R2	0.9466	0.958	0.9821	0.9958

CEBPA



CEBPA	61.4	59	57
1	19.60	19.79	19.52
0.3	21.67	21.63	21.79
0.1	23.33	23.36	23.40
0.03	25.41	25.48	25.28
0.01	27.12	27.06	26.89
Efficiency	0.8455102	0.8696129	0.8795046
R2	0.9996	0.9992	0.9977





CPT1A	62.4	60	58	56.7
1	29.56	29.32	29.52	30.27
0.3	31.08	30.95	30.81	31.18
0.1	32.87	33.09	32.68	33.12
0.03	35.90	34.58	34.65	35.79
0.01	36.13	34.83	36.64	36.40
Efficiency	0.8964958	1.1924701	0.8908469	0.9786169
R2	0.96	0.95	0.9926	0.96

HNF4A



HNF4A	61.4	59	57	55.7
1	21.34	21.82	22.25	22.96
0.3	23.65	24.37	25.27	26.42
0.1	25.46	26.40	27.28	28.18
0.03	27.85	28.35	29.90	30.57
0.01	29.05	30.65	31.23	32.41
Efficiency	0.7966064	0.7026406	0.6640479	0.6463708
R2	0.9932	0.998	0.9889	0.9881



MVK	59.4	56.9	55	53.7
1	24.67	25.71	26.09	26.40
0.3	27.31	28.07	28.91	29.41
0.1	29.24	30.19	31.37	31.23
0.03	31.30	31.91	33.51	34.02
0.01	32.65	34.08	36.03	35.05
Efficiency	0.7795658	0.7493881	0.6009804	0.6899054
R2	0.9906	0.9973	0.998	0.9832

ABCA1



ABCA1	63.4	61	59	57.7
1	26.22	26.38	27.03	26.72
0.3	28.13	28.24	28.54	28.65
0.1	30.07	30.01	30.30	30.34
0.03	31.06	31.20	31.57	31.83
0.01	32.09	32.27	32.74	33.11
Efficiency	1.1906288	1.1826525	1.2182438	1.0555872
R2	0.9743	0.9851	0.9938	0.9947

MVK



HK1	62.4	60	58	56.7
1	26.37	26.41	26.04	26.13
0.3	28.02	28.37	27.59	27.72
0.1	30.32	29.86	29.61	29.21
0.03	32.00	31.60	31.74	31.24
0.01	33.06	34.04	32.57	33.04
Efficiency	0.9410328	0.865158	0.9519212	0.9412008
R2	0.9866	0.9921	0.9863	0.9968

SCARB1



SCARB1	62.4	60	58	56.7
1	20.42	20.38	20.78	20.68
0.3	22.58	22.61	22.74	22.98
0.1	24.06	24.27	24.59	24.80
0.03	26.11	26.05	26.54	26.81
0.01	27.59	27.66	28.05	28.59
Efficiency	0.9046257	0.8963609	0.8720633	0.7953614
R2	0.9977	0.9974	0.9989	0.9990

HK1
Rat primers for RT-qPCR of BRIN-BD11 cells

Actb

200nM



100nM



Actb	59	57	55.7	55
1	14.48	14.33	14.88	13.68
0.3	16.00	16.29	16.28	16.35
0.1	17.28	17.74	17.90	18.08
0.03	19.37	19.68	19.82	20.04
0.01	20.80	21.21	21.35	21.25
Efficiency	1.0527356	0.9564201	1.0135384	0.8425983
R2	0.9956	0.9991	0.9974	0.9866

Rpl13a

2 200nM







~
~
J
-

Rpl13a	63.3	61.4	59	57	55.7
1	16.42	16.15	15.98	16.18	16.84
0.3	17.60	17.51	17.70	18.03	18.22
0.1	18.95	18.49	18.63	19.07	19.40
0.03	20.83	20.43	20.50	21.14	21.33
0.01	21.97	21.81	21.86	22.45	22.75
Efficiency	1.2323495	1.2441613	1.2037336	1.0852234	1.1615314
R2	0.9939	0.9914	0.9945	0.9945	0.9949



Hkl	61.4	59	57	55.7
1	25.28	21.98	21.59	22.06
0.3	24.37	23.52	23.08	23.37
0.1	25.74	24.67	24.46	24.47
0.03	27.57	26.61	26.22	26.13
0.01	29.06	27.86	27.77	27.63
Efficiency	1.9306367	1.1717975	1.1011962	1.2899785
R2	0.8058	0.9964	0.9986	0.9951

Hnf4a



Hnf4a	61.4	59	57	55.7
1	31.23	32.04	32.37	33.19
0.3	34.23	34.77	34.96	35.55
0.1	35.33	35.63	37.41	39.31
0.03	36.91	39.34		39.58
0.01				
Efficiency	0.8935006	0.6623457	0.5794947	0.6639376
R2	0.9586	0.9573	0.999	0.9128

Hk1



Pdx1	61.4	59	57	55.7
1	20.20	19.96	19.97	20.37
0.3	22.39	21.28	21.42	21.61
0.1	22.82	22.81	22.92	23.08
0.03	24.54	24.53	24.73	24.80
0.01	25.95	26.12	25.90	26.13
Efficiency	1.3212218	1.095499	1.1350287	1.1867001
R2	0.9762	0.9972	0.9976	0.9971

Srebf1

Pdx1



Srebf1	61.4	59	57	55.7
1	27.14	27.46	28.20	28.85
0.3	29.31	30.06	30.62	31.40
0.1	31.60	32.27	33.17	33.34
0.03	33.93	34.14	35.20	35.54
0.01	35.63	35.81	37.82	38.36
Efficiency	0.70331	0.739178	0.6215402	0.6445533
R2	0.998	0.9931	0.9979	0.9958



Ins1	64.5	60.9	58	56
1	17.67	15.33	16.02	17.20
0.3	17.55	17.45	18.38	19.42
0.1	20.03	19.20	20.76	21.58
0.03	22.56	21.38	23.22	24.07
0.01	23.39	23.09	24.36	25.95
Efficiency	1.0160538	0.8068622	0.7069122	0.6814312
R2	0.9246	0.9995	0.9888	0.9992

Abcg1



Abcg1	61.4	59	57	55.7
1	22.29	22.54		
0.3	24.17	24.37	24.55	24.65
0.1	25.51	25.93	26.26	26.11
0.03	27.71	27.62	28.28	28.14
0.01	29.06	28.79	29.59	29.47
Efficiency	0.9615034	1.0762592	0.9455305	0.9967133
R2	0.9967	0.9964	0.9957	0.9962

Ins1





Abca1	61.4	59	57	55.7
1	25.25	24.45	22.88	23.05
0.3	25.45	25.50	24.64	25.02
0.1	27.45	26.13	26.27	26.39
0.03	28.40	28.04	28.40	28.19
0.01	30.06	29.18	29.36	29.27
Efficiency	1.5072278	1.6096618	0.9905646	1.0887878
R2	0.9534	0.9769	0.9925	0.9943

Txnip



Txnip	61.4	59	57	55.7
1	23.86	23.69	23.98	24.18
0.3	25.76	25.98	25.84	26.19
0.1	27.80	27.90	28.01	28.23
0.03	29.65	29.68	30.44	29.92
0.01	31.58	31.70	31.80	32.14
Efficiency	0.8146028	0.792771	0.7657868	0.7967935
R2	0.9993	0.9983	0.9948	0.9978

Ldlr



Ldlr	61.4	59	57	55.7
1	23.26	23.18	23.40	23.77
0.3	25.22	25.39	25.76	26.19
0.1	27.35	27.73	28.27	28.28
0.03	29.29	29.53	29.87	30.48
0.01	31.31	31.56	31.73	32.17
Efficiency	0.7709204	0.7352438	0.7411644	0.7262211
R2	0.9993	0.9976	0.9921	0.9977

Human primers for ChIP-qPCR of HepG2 cells

GAPDH



GAPDH	66.4	64	62	60.7	63
1	29.48	29.71	30.16	30.27	25.33
0.3	31.96	31.75	31.68	32.11	27.55
0.1	33.19	33.46	33.24	33.42	29.58
0.03	34.47	35.19	35.59	35.21	30.42
0.01	37.25	39.71	38.7	39.5	33.43
Efficiency	0.893041	0.636267	0.73255	0.708061	0.830286
R2	0.9756	0.9473	0.9716	0.9356	0.9746

HMGCR



HMGCR	65.4	63	61	59.7	
1	23.95	23.87	23.75	23.65	
0.3	0.3 25.50		25.30	25.28	
0.1	0.1 26.86		27.08	26.76	
0.03	28.94	28.74	28.52	28.63	
0.01	30.70	30.34	30.12	30.02	
Efficiency	0.9743357	1.0301579	1.0569461	1.0438689	
R2	1.00	1.00	1.00	1.00	





LDLR	65.4	63	61	59.7	
1	1 23.52		23.53	23.64	
0.3	0.3 25.13		25.19	25.30	
0.1	0.1 26.98		26.80	26.67	
0.03	28.50	28.60	29.06	28.70	
0.01	30.02	30.02	30.17	29.82	
Efficiency	1.0199225	1.0272733	0.955793	1.0746928	
R2	1.00	1.00	0.99	1.00	

PPARG



PPARG	65.4	63	61	59.7	
1	1 23.57		23.52	23.81	
0.3	25.15	25.36	25.23	25.41	
0.1	0.1 26.87		26.83	27.03	
0.03	28.53	28.81	28.60	28.49	
0.01	29.78	30.25	30.15	30.04	
Efficiency	1.0721559	0.972082	0.9969967	1.0989876	
R2	1.00	1.00	1.00	1.00	





ABCA1	65.4	63	61	59.7	
1	24.12	24.27	24.07	24.85	
0.3	0.3 25.92		25.92	25.79	
0.1	27.58	27.32 27.51		27.17	
0.03	29.02	29.27	29.10	29.49	
0.01	31.17	30.88	30.34	30.77	
Efficiency	0.9534825	0.9937994	1.0775968	1.0984366	
R2	1.00	1.00	1.00	0.98	

ABCG1



ABCG1	65.4	63	61	59.7	
1	1 24.47		25.06	24.62	
0.3	0.3 26.06		25.83	25.98	
0.1	0.1 28.36		27.49	27.34	
0.03	29.33	29.41	29.51	29.08	
0.01	31.32	31.13	30.66	30.80	
Efficiency	0.9716778	1.0161154	1.1702522	1.1085871	
R2	0.99	1.00	0.98	1.00	



НКІ	66.4	64	62	60.7
1	26.08	25.73	25.38	25.05
0.3	0.3 27.53		26.26	26.36
0.1	29.17	28.88	28.74	27.76
0.03	31.14	31.37	30.50	30.20
0.01	33.82	33.59	32.70	32.24
Efficiency	0.829128	0.8073701	0.842305	0.8820033
R2	0.98	0.99	0.98	0.98

SLC2A2



SLC2A2	65.4	63	61	59.7	
1	23.99	23.76	23.57	23.83	
0.3	25.29	25.21	25.07	25.23	
0.1	0.1 26.91		26.44	26.61	
0.03	28.57	28.57	28.47	28.34	
0.01	29.73	30.46	29.95	30.01	
Efficiency	1.1805951	0.9873649	1.0401237	1.1048289	
R2	1.00	0.99	1.00	1.00	

HK1

Appendix C: Data tables

Concentration	Atorva	Atorvastatin S			statin		Rosuvastatin			Pravastatin		
	Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	N
5	92.63	2.300	8	87.54	4.851	8	91.62	3.390	8	98.24	5.471	8
10	97.80	4.892	8	87.97	4.953	8	104.9	3.896	8	98.97	6.236	8
50	84.17	2.473	8	74.16	5.072	8	88.25	4.610	8	99.27	2.570	8
100	80.33	11.29	5	71.02	6.851	3	95.82	3.394	5	95.89	3.586	4
200	61.94	8.547	6	37.99	6.606	2	96.10	4.593	6	90.56	3.441	6
500	12.88	4.653	6	3.097	1.084	4	82.80	17.08	5	70.11	5.547	6

Table S. 1Data table for Figure 4-1A

Table S. 2Data table for Figure 4-1B

Concentration	Mean	SEM	N
0	100	1.375553	8
5	105.4508	9.61456	7
10	115.9366	7.958282	7
20	100.7191	4.811202	8

Table S. 3Data table for Figure 4-2

Control			1mM S	SB		5mM S	SB		DMSO			Atorvastatin		
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	Ν
5.88	0.71	5	6.92	1.48	3	6.35	0.87	3	6.12	0.95	3	6.41	0.90	3

Table S. 4Data table for Fig 4-3A

Control			DMSO cor	ntrol	Sodium Butyrate			Pravastat			
Mean	SEM	Ν	Mean	SEM	Ν	Mean	SEM	N	Mean	SEM	Ν
0.923508	0.07685	6	1.084472	0.111021	5	0.635663	0.089083	6	0.72072	0.09910	6
Atorvastatin			Simvastatin			Rosuvastatin					
Mean	SEM	Ν	Mean	SEM	N	Mean	SEM	N			
0.498427	0.05554	6	0.48819	0.067417	6	0.587117	0.099725	6			

Control			DMSO			Atorvastatin			
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	
0.319149	0.056971	9	0.396359	0.077689	8	0.264036	0.049961	10	
Rosuvastati	n		Sodium but	tyrate					
Mean	SEM	N	Mean	SEM	N				
0.285854	0.047104	10	0.24715	0.046143	10				

Table S. 5Data table for Figure 4-3B

Table S. 6Data table for Figure 4-3C

Control			DMSO			Sodium Butyrate			Atorvastatin		
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	N
0.1822	0.0141	9	0.2015	0.0200	9	0.1955	0.0238	9	0.2331	0.0217	9

Table S. 7	Data table for Figure 4-3D
------------	----------------------------

Control			DMSO		Atorvastatin			Ator + SB			
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	N
1.9304	0.2896	5	2.2283	0.2027	6	1.1825	0.0702	5	1.0743	0.1155	5

Table S. 8	Data table for	Figure 4-3E
------------	----------------	-------------

Control			DMSO		Atorvastatin				
Mean	SEM	N	Mean	SEM	N	Mean	SEM	Ν	
108.9469	6.04404	6	104.0208	13.17387	6	74.55724	7.314398	6	
Sodium Butyrate			Ator + SB						
Mean	SEM	N	Mean	SEM	Ν				
87.96054 4.24312 6		69.26368	13.10441	6					

Control			1mM Sodiu	ım Acetate	5mM Sodium Acetate			
Mean	SEM 1		Mean	SEM		Mean	SEM	N
2.5473	5473 0.146822 9		2.694133 0.091024 9			2.669854	0.193046	9
1uM TSA			2.5uM TSA	L	TSA + SA			
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N
2.537053 0.119734 9		9	2.024237	0.110603	9	2.249391	0.135274	6

Table S. 9Data table for Figure 4-3F

Table S. 10Data table for Figure 4-4A

Control			SB 1mM		SB 5mM			
Mean	Mean SEM N			SEM	N	Mean SEM		
81.08217	13.23381	7	70.26502	16.63544	7	85.90672	20.05331	8

Table S. 11Data table for Figure 4-4B

DMSO		Atorvastatin				
Mean	SEM	N	Mean	SEM	N	
66.83944	9.400247	7	67.481	7.298949	6	

Table S. 12Data table for Figure 4-5A

Control			DMSO		Atorvastatin			
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N
1	0.138594	7	1.16516	0.338658	7	1.959733	0.469753	8
Sodium Butyrate			U18666A					
Mean	SEM	N	Mean	SEM	N			
1.175793 0.461507 8		8	2.821978	0.296761	8			

Table S. 13Data table for Figure 4-5B

Contro	Control			DMSO			Atorvastatin			Sodium Butyrate		
Mea	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	Ν	
n												
1	0.0533	8	1.030	0.062	8	1.008	0.049	8	0.933	0.035	8	
	5		6	5		2	6		9	3		

Control	Control			Butyrate	DMSO			Atorvastatin			
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	N
1	0	3	0.878	0.114	3	1.0780	0.21	3	1.12253	0.247	3
			7	6					5	4	

Table S. 14Data table for Figure 4-6A

Table S. 15Data table for Figure 4-6B

Control			Sodium Butyrate			DMSO			Atorvastatin			
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	
1	0	3	0.7989	0.1435	3	0.9098	0.1686	3	1.0751	0.2264	3	

Table S. 16Data table for Figure 4-7A

Control			1 mM SB			5 mM SB	5 mM SB			
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N		
1	0	5	0.84523	0.04152	3	0.655183	0.051189	5		
DMSO			Atorvastatir	1						
Mean	SEM	N	Mean	SEM	N					
0.989459	0.118174	4	1.373999	0.173825	5					

Table S. 17Data table for Figure 4-7B

Control			1 mM SB			5 mM SB		
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N
1	0	5	0.946458	0.026609	3	0.991058	0.048198	5
DMSO			Atorvastatir	1				
Mean	SEM	N	Mean	SEM	N			
0.99866	0.125237	5	1.068916	0.152963	5			

Control				1 mM SB				5	mM SB		
Mean	SEM	N]	Mean	S	SEM	N	Ν	lean	SEM	N
1	0	4	(0.841128	C	0.1708	3	0	.867912	0.060888	4
DMSO				Atorvastat	in						
Mean	SEM	N	1	Mean		SEM		N			
0.989315	0.288271	4		0.913721		0.12886		4			

Table S. 18Data table for Figure 4-7C

Table S. 19Data table for Figure 4-7D

Control			Sodium Bu	ıtyrate	DMSO			Atorvastatin			
Mean	SEM	N	Mean SEM N			Mean	SEM	N	Mean	SEM	N
1	0	4	0.594407	0.045973	4	1.053937	0.10377	4	1.433487	0.365412	3

Table S. 20Data table for Figure 4-7E

Control	Control			ıtyrate	DMSO			Atorvastatin			
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	N
1	0	4	0.659747	0.089939	4	1.110957	0.21698	4	1.574868	0.494242	4

Table S. 21Data table for Figure 4-7F

Control			Sodium Bu	utyrate	DMSO			Atorvastatin			
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	N
1	0	3	0.660057	0.118032	3	1.050204	0.089883	3	1.138781	0.137285	3

Table S. 22Data table for Figure 4-8A

	Control			1 mM SB			5 mM SE	3		TSA			
	Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	
8 h	1	0	3	0.72662	0.10130	3	0.80361	0.16853	3				
16 h	1	0	3	0.55585	0.13482	3	0.58905	0.16519	3	0.60962	0.0870	3	

	Control			1 mM SE	3		5 mM SE	3		TSA			
	Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	
8 h	1	0	3	0.86875	0.15438	3	1.31019	0.60413	3				
16 h	1	0	3	0.76542	0.04133	3	0.67166	0.12489	3	0.58327	0.18924	3	

Table S. 23Data table for Figure 4-8B

Table S. 24Data table for Figure 4-8C

Untreated		Untreated				
Mean	SEM	N	Mean	SEM	N	
1	0.076046	6	0.693246	0.11573	6	

Table S. 25Data table for Figure 4	1-8D
------------------------------------	------

Untreated			Sodium Butyrate					
Mean	SEM	N	Mean	SEM	N			
1	0.096883	6	0.976737	0.267836	5			

Table S. 26	Data table for Figure 4	1-9A
-------------	-------------------------	------

Control			1 mM SB		5 mM SB				
Mean	SEM	N	Mean	SEM	N	Mean	SEM	Ν	
1	0	5	0.727582	0.115316	3	0.422157	0.056063	5	
DMSO		-	Atorvastatir	1					
Mean	SEM	N	Mean	SEM	N				
0.999054	0.086362	4	0.558304	0.058791	5				

Table S. 27Data table for Figure 4-9B

Control			Sodium	Butyrate	DMSO		Atorvastatin				
Mean	SE	N	Mean	SEM	N	Mean	SEM	Ν	Mean	SEM	N
	М										
1	0	3	0.3184	0.0485	3	1.0844	0.0736	2	1.5145	0.185	3
			03	96		09	97		69	13	

Control			1 mM SB			5 mM SB				
Mean	SEM	N	Mean SEM N			Mean	SEM	N		
1	0	3	1.949224	0.303266	3	1.767324	0.144834	3		
DMSO	-		Atorvastatir	1			-			
Mean	SEM	N	Mean	SEM	N					
0.903413	0.107551	3	0.469336	0.071727	3					

Table S. 28Data table for Figure 4-9C

Table S. 29Data table for Figure 4-9D

Control			1 mM SB			5 mM SB					
Mean	SEM	N	Mean SEM N		Mean	SEM	N				
1	0	5	0.94971	0.007517	3	0.767369	0.059091	5			
DMSO			Atorvastatir	1			-				
Mean	SEM	N	Mean	SEM	N						
0.86024	0.10773	5	1.065901	0.096204	5						

Table S. 30Data table for Figure 4-9E

Control			Sodium Bu	utyrate	DMSO			Atorvastatin			
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	N
1	0	3	0.795773	0.07447	3	1.037112	0.01734	2	0.824361	0.07919	3

Table S. 31Data table for Figure 4-9F

Control			Sodium Bu	utyrate	DMSO			Atorvastatin			
Mean	SEM	N	Mean	Mean SEM N		Mean SEM			Mean	SEM	N
1	0	3	1.654262	0.25269	3	1.115571	0.15223	3	1.038203	0.23389	3

	Control			1 mM SB		5 mM SB			TSA			
	Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	N
8 h	1	0	3	0.616607	0.07379	3	0.523082	0.10350	3			
16 h	1	0	3	0.504396	0.09811	3	0.296376	0.06969	3	0.39140	0.1374	3

ABCG1			CAV1			SREBF1			MVK			
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	
668.3042	171.426	3	163.329	43.3000	3	15.39177	2.3664	3	3.07924	1.58021	3	
SCARB1		-	APOB		APOA1	-	HMGCR	HMGCR				
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	
2.575575	0.09559	4	1.997781	0.88145	3	1.984166	0.14157	3	1.56235	0.5659	3	
LDLR			NR1H3		SREBF2		ABCA1					
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	
1.217397	0.38314	4	1.194207	0.10248	3	0.887693	0.09144	4	0.19922	0.0303	4	
CYP7A1					-							
Mean	SEM	N										
0.051334	0.01367	3										

Table S. 33Data table for Figure 4-11A

Table S. 34Data table for Figure 4-11B

HMGCR			MVK			APOB			LDLR			
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	
2.163681	0.54228	3	1.637362	0.18192	3	1.456659	0.7207	3	1.340921	0.43170	3	
APOA1			SCARB1		SREBF2			CAV1				
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	Ν	
1.197124	0.14569	3	1.031221	0.24284	3	0.912963	0.1326	3	0.882503	0.17505	3	
ABCA1		•	CYP7A1			ABCG1			SREBF1			
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	
0.839699	0.13737	3	0.819803	0.02644	3	0.594524	0.2546	3	0.537201	0.04215	3	

Table S. 35Data table for Figure 4-11C

Abca1			Abcg1			Srebf1		Ldlr			
Mean	SEM	Ν	Mean	SEM	N	Mean	SEM	N	Mean	SEM	N
7.0290	1.0536	3	2.3227	0.1839	3	1.9528	0.3383	3	0.2267	0.0524	3

Ldlr			Srebf1			Abcg1			Abca1		
Mean	SEM	N	Mean	Mean SEM N		Mean	Mean SEM N			SEM	N
2.7587	0.5323	3	1.0508	0.3389	3	0.27059	0.0562	3	0.0952	0.0193	3

Table S. 36Data table for Figure 4-11D

Table S. 37Data table for Figure 4-12A

Control			1 mM SB			5 mM SB			
Mean	SEM	Ν	Mean	SEM	Ν	Mean	SEM	Ν	
1	0	3	0.714237	0.249601	3	0.947624	0.056863	3	
DMSO			Atorv						
Mean	SEM	Ν	Mean	SEM	Ν				
1.018796	0.063817	3	1.121307	0.097899	3				

Table S. 38Data table for Figure 4-12B

Control			1 mM SB			5 mM SB		
Mean	SEM	N	Mean	SEM	N	Mean	SEM	Ν
1	0	3	0.7846	0.015367	3	1.228377	0.338997	3
DMSO			Atorvastatir	1	-			
Mean	SEM	N	Mean	SEM	N			
0.987154	0.120969	3	0.662959	0.238234	3			

Table S. 39Data table for Figure 5-1A

	Chronic		Stimulated	Stimulated					
	Mean	SEM	N	Mean	SEM	N			
Control	218.3954	21.82505	9	1013.544	112.7708	9			
1 mM SB	170.0649	22.69239	9	1288.458	126.3635	9			
5 mM SB	94.55493	14.41834	9	390.4675	43.45239	9			

	Chronic		Stimulated	Stimulated				
	Mean	SEM	N	Mean	SEM	N		
Control	206.262	26.21946	9	1013.544	112.7708	9		
TSA	88.61675	21.22867	9	177.4669	39.6835	9		

Table S. 40Data table for Figure 5-1B

Table S. 41Data table for Figure 5-1C

Control			1 mM SB			5 mM SB			TSA			
Mean	SEM	N Mean SEM N			Mean SEM N			Mean	SEM	Ν		
637.3141	92.3387	9	461.7518	83.4407	9	334.4585	52.4280	9	186.8659	14.4519	9	

Table S. 42Data table for Figure 5-1D

Control			1 mM SB			5 mM SB	5		TSA			
Mean	SEM	N	Mean	Mean SEM N			Mean SEM N			Mean SEM N		
3.716685	0.70238	9	6.850159	0.97249	9	2.56137	0.36144	9	1.745774	0.30346	9	

Table S. 43	Data table for Figure 5-	2A
-------------	--------------------------	----

	Basal		Stimulated			
	Mean	SEM	N	Mean	SEM	N
Control	4.902342	0.332749	9	20.27087	2.255416	9
5 mM SB	6.150682	0.504484	9	26.36796	3.230384	9

Table S. 44	Data table for Figure	5-2B
-------------	------------------------------	------

Control		5 mM SB					
Mean	SEM	N	Mean	SEM	N		
20.27087	2.255416	9	22.7241	2.049333	9		

pIRS1	Control			Sodium	butyrate		DMSO			Atorvas	tatin	
	Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	N
Unstimulated	1	0	4	1.3729	0.2634	4	0.8253	0.1640	4	1.1697	0.2739	4
Insulin	1.7829	0.5930	4	1.7669	0.6572	4	1.4467	0.1098	4	2.6732	0.9892	4
stimulated												
pIRB	Control			Sodium	butyrate		DMSO			Atorvast	tatin	
Unstimulated	1	0	3	1.0251	0.1993	3	1.1349	0.1649	3	0.8930	0.3080	3
Insulin	1.2153	0.3081	3	0.8496	0.2245	3	1.3579	0.4763	3	1.7021	0.4609	3
stimulated												
рАКТ	Control			Sodium	butyrate		DMSO			Atorvast	tatin	-
Unstimulated	1	0	3	0.6457	0.0774	3	1.0303	0.1289	3	0.9559	0.0539	3
Insulin	1.6149	0.4924	3	0.7272	0.1061	3	1.1741	0.1147	3	0.9469	0.2327	3
stimulated												

Table S. 45Data table for Figure 5-3

Table S. 46Data table for Figure 4-4A

Control	Control					5 mM SB		
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N
1	0	3	1.553009	0.260041	3	2.224221	0.346369	3
DMSO			Atorvastatir	1				
Mean	SEM	N	Mean	SEM	N			
1.087269	0.144291	3	1.068166	0.074624	3			

Table S. 47Data table for Figure 4-4B

Control			1 mM SB			5 mM SB			
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	
1	0	3	1.44517	0.270056	3	1.320628	0.176685	3	
DMSO			Atorvastatir	1					
Mean	SEM	N	Mean	SEM	N				
1.167706	0.308929	3	1.053607	0.190113	3				

Control	Control				5 mM SB			
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N
1	0	3	0.576051	0.158942	3	0.749055	0.222725	3
DMSO		-	Atorvastatir	1	-			
Mean	SEM	N	Mean	SEM	N			
0.704479	0.300398	3	0.760322	0.249118	3			

Table S. 48Data table for Figure 4-4C

Table S. 49Data table for Figure 4-4D

Control			5 mM SB			DMSO			Atorvastatin		
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	Ν
1	0	3	0.91124	0.110143	3	1.088658	0.022506	2	1.26797	0.141355	3

Table S. 50Data table for Figure 4-5A

Control	Control					5 mM SB		
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N
1	0	3	1.680099	0.17753	3	3.001881	1.068331	3
DMSO			Atorvastatir	1				
Mean	SEM	N	Mean	SEM	N			
1.324455	0.306324	3	1.652686	0.178387	3			

Table S. 51Data table for Figure 4-5B

Control	Control					5 mM SB	5 mM SB			
Mean	SEM	Ν	Mean	SEM	N	Mean	SEM	N		
1	0	3	0.760989	0.126105	3	0.811917	0.128182	3		
DMSO			Atorvastatir	1						
Mean	SEM	N	Mean	SEM	N					
1.262924	0.21487	3	1.054274	0.212314	3					

	Control			1 mM SB		5 mM SB			TSA			
	Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	Ν
8 h	1	0	3	1.804797	0.72871	3	2.198968	0.76037	3			
16 h	1	0	3	3.100068	1.59447	3	4.746619	3.02747	3	6.464769	4.6509	3

Table S. 52Data table for Figure 4-6

Table S. 53Data table for Figure 5-7A

ABCG1			HK1			CAV1			SREBF1		
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	Ν
668.3042	171.426	3	222.415	93.9468	4	163.329	43.3000	3	15.39177	2.3664	3
GLUT2			CPT1A		•	MVK			SCARB1		
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	Ν
4.603193	0.53638	3	3.410493	0.14747	3	3.079249	1.58021	3	2.575575	0.0955	4
APOB			APOA1		-	PPARG			HMGCR		
Mean	SEM	N									
1.997781	0.88145	3	1.984166	0.14157	3	1.685258	0.17573	3	1.562352	0.5659	3
LDLR			NR1H3		•	FOX01			SREBF2		
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	Ν
1.217397	0.38314	4	1.194207	0.10248	3	1.011749	0.05941	2	0.887693	0.09144	4
PEPCK			CEBPA		•	INSR			HNF4A		
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	Ν
0.69026	0.08299	3	0.669441	0.06776	3	0.522486	0.08699	3	0.47728	0.25262	3
GK			ABCA1		-	CYP7A1					-
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	1		
0.376628	0.07832	3	0.199221	0.03030	4	0.051334	0.01367	3			

Hnf4a			Txnip			Abca1	Abca1			
Mean	SEM	N	Mean	SEM	N	Mean	SEM	Ν		
34.41935	14.4156	3	9.493874	3.038801	3	7.029065	1.053698	3		
Hk1	-	-	Abcg1		-	Srebf1		-		
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N		
2.861145	0.466113	3	2.322731	0.183997	3	1.952818	0.338361	3		
Pdx1	-		Ins		-	Ldlr				
Mean	SEM	N	Mean	SEM	N	Mean	SEM	Ν		
1.560767	0.355013	3	1.515524	0.211544	3	0.226792	0.052465	3		

Table S. 54Data table for Figure 5-7B

Table S. 55Data table for Figure 5-7C

PPARG			HMGCR			MVK			APOB		
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	Ν
7.115879	0.95227	3	2.163681	0.54228	3	1.637362	0.18192	3	1.456659	0.7207	3
LDLR			CPT1A			APOA1			PEPCK		-
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	Ν
1.340921	0.43170	3	1.213748	0.03436	3	1.197124	0.14569	3	1.19407	0.24131	3
HNF4A			GK			NR1H3			CEBPA		
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	Ν
1.218385	0.54250	3	0.982249	0.03704	3	0.970094	0.00620	2	1.043452	0.09026	3
SCARB1			FOYO1			INCD			SDEBEJ		
SCARDI			голог			IINSK			SKEDI ⁻ 2		
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	N
Mean 1.031221	SEM 0.24284	N 3	Mean 1.028486	SEM 0.04348	N 2	Mean 0.927445	SEM 0.36718	N 3	Mean 0.912963	SEM 0.13260	N 3
Mean 1.031221 CAV1	SEM 0.24284	N 3	Mean 1.028486 ABCA1	SEM 0.04348	N 2	Mean 0.927445 GLUT2	SEM 0.36718	N 3	Mean 0.912963 CYP7A1	SEM 0.13260	N 3
Mean 1.031221 CAV1 Mean	SEM 0.24284 SEM	N 3 N	Mean 1.028486 ABCA1 Mean	SEM 0.04348 SEM	N 2 N	Mean 0.927445 GLUT2 Mean	SEM 0.36718 SEM	N 3 N	Mean 0.912963 CYP7A1 Mean	SEM 0.13260 SEM	N 3 N
Mean 1.031221 CAV1 Mean 0.882503	SEM 0.24284 SEM 0.17505	N 3 N 3	Mean 1.028486 ABCA1 Mean 0.839699	SEM 0.04348 SEM 0.13737	N 2 N 3	Mean 0.927445 GLUT2 Mean 0.834456	SEM 0.36718 SEM 0.13281	N 3 N 3	Mean 0.912963 CYP7A1 Mean 0.819803	SEM 0.13260 SEM 0.02644	N 3 N 3
Mean 1.031221 CAV1 Mean 0.882503 ABCG1	SEM 0.24284 SEM 0.17505	N 3 N 3	Mean 1.028486 ABCA1 Mean 0.839699 SREBF1	SEM 0.04348 SEM 0.13737	N 2 N 3	Mean 0.927445 GLUT2 Mean 0.834456 HK1	SEM 0.36718 SEM 0.13281	N 3 N 3	Mean 0.912963 CYP7A1 Mean 0.819803	SEM 0.13260 SEM 0.02644	N 3 N 3
Mean 1.031221 CAV1 Mean 0.882503 ABCG1 Mean	SEM 0.24284 SEM 0.17505 SEM	N 3 N 3	Mean 1.028486 ABCA1 Mean 0.839699 SREBF1 Mean	SEM 0.04348 SEM 0.13737 SEM	N 2 N 3	Mean 0.927445 GLUT2 Mean 0.834456 HK1 Mean	SEM 0.36718 SEM 0.13281 SEM	N 3 N 3	Mean 0.912963 CYP7A1 Mean 0.819803	SEM 0.13260 SEM 0.02644	N 3 N 3

Ldlr			Hk1			Srebf1	Srebf1			
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N		
2.758786	0.532392	3	1.160061	0.224709	3	1.050852	0.33897	3		
Pdx1			Ins		•	Txnip				
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N		
0.962424	0.059743	3	0.870342	0.100955	3	0.872894	0.211935	3		
Hnf4a	-	_	Abcg1		-	Abca1				
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N		
0.565406	0.086168	3	0.270599	0.056288	3	0.095209	0.01934	3		

Table S. 56Data table for Figure 5-7D

Table S. 57Data table for Figure 6-1A

	Sodium Butyrate						
Concentration	Mean	SEM	N				
0	100	0.961637	3				
0.1	78.72977	16.19318	3				
0.5	42.61836	4.544317	3				
1	32.8029	1.065968	3				
5	25.24123	4.056318	3				

Table S. 58Data table for Figure 5-1B

Control			DMSO		Atorvastat	in	Simvastatin				
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	Ν
1	0.03977	6	1.104573	0.05643	5	1.220747	0.08595	6	1.198252	0.05247	6
Rosuvasta	atin		Pravastatir	1		Sodium Butyrate			Trichostatin A		
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	Ν
1.08882	0.06367	6	1.295094	0.14316	6	0.327732	0.04628	6	-0.01126	0.02055	6

	Sodium Buty	Sodium Butyrate						
Concentration	Mean	SEM	N					
0	100	0	3					
0.05	67.82084	9.012196	3					
0.1	46.24424	10.80396	3					
0.5	22.72317	4.996715	3					
1	16.14158	3.794745	3					
5	9.260247	3.908096	3					

Table S. 59Data table for Figure 5-1C

Table S. 60Data table for Figure 5-1D

Control			Atorvastatin			Pravastatin			TSA		
Mean	SEM	N	Mean	SEM	N	Mean SEM N		N	Mean	SEM	Ν
1	0.012439	6	0.985926	0.024931	6	1.018173	0.041494	6	0.036571	0.022419	5

Table S. 61Data table for Figure 5-2

	Atorvastatin	Atorvastatin						
Concentration	Mean	SEM	N					
0	1	0	3					
5	1.143464	0.11431	5					
10	1.116016	0.053551	6					
50	1.050396	0.101692	6					

Table S. 62Data table for Figure 6-3A

	Sodium Buty	Sodium Butyrate						
Concentration	Mean	SEM	N					
0	100	0	3					
0.1	69.72656	2.732904	3					
0.25	61.80136	7.56832	2					
0.5	50.63009	6.328042	3					
1	30.52603	5.587201	3					
5	12.62711	1.864643	3					

Control			Atorvastatir	1	Rosuvastatin			
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N
1	0.052643	8	0.905545	0.053925	6	0.972426	0.073523	5
Pravastatin			Simvastatin		Sodium Butyrate			
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N
0.971606	0.069404	5	0.928791	0.08354	4	0.125745	0.018713	3

Table S. 63Data table for Figure 6-3B

Table S. 64Data table for Figure 6-3C

	Atorvastatin		
Concentration	Mean	SEM	N
1	0.854076	0.037438	5
10	0.905545	0.053925	6
50	0.875636	0.072929	5

Table S. 65Data table for Figure 6-3D

Control			Atorvastati	in	Rosuvastat	in	Pravastatin	1			
Mean	SEM	Ν	Mean	SEM	N	Mean SEM N		N	Mean	SEM	Ν
1	0.03296	10	0.997812	0.03693	6	1.098003	0.03306	6	1.055965	0.03041	6
Atorvastatin 50uM			Sodium Butyrate			Trichostati	n A				
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N			
0.918106	0.03744	6	0.272226	0.04164	5	-0.0002 0.00290 5					

	HepG2			HeLa			
	Mean	SEM	Ν	Mean	SEM	Ν	
Control	1	0	3	1	0	3	
DMSO	0.894935	0.053136	3	1.054556	0.018282	3	
Atorvastatin	0.950889	0.024353	3	1.05843	0.022441	3	
Pravastatin	1.000319	0.005347	3	1.120982	0.010564	3	
Sodium	0.452774	0.039868	3	0.502634	0.037546	3	
butyrate							
Trichostatin A	0.044317	0.040273	3	0.051373	0.059027	3	

Table S. 66Data table for Figure 6-4

Table S. 67Data table for Figure 6-5

V-ND			V-HFD			A-ND			A-HFD		
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	N
1	0	3	0.955623	0.325894	3	0.99469	0.046384	3	0.974464	0.225863	3

Table S. 68Data table for Figure 6-6

DMSO		Curcumin			Atorvastatin			
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N
1	0.182752	5	0.285028	0.056433	5	0.938461	0.14097	5

Table S. 69Data table for Figure 6-7A

Control			1 mM SB		5 mM SB			
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N
1	0	5	2.000413	0.462792	3	8.863314	2.468012	5
DMSO			Atorvastatin			Curcumin		
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N
0.771394	0.149974	5	1.249411	0.658782	5	0.914562	0.11411	3

Control			1 mM SB			5 mM SB		
Mean	SEM	N	Mean	SEM	N	Mean	SEM	Ν
1	0	3	18.14247	12.12224	3	115.8061	84.04357	3
DMSO			Atorvastatir	1		Curcumin		
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N
0.734829	0.300808	3	1.056317	0.510828	3	1.66294	0.809463	3

Table S. 70Data table for Figure 6-7B

Table S. 71Data table for Figure 6-7C

Control			1 mM SB			5 mM SB		
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N
1	0	3	0.947585	0.288335	3	0.733673	0.182408	3
DMSO			Atorvastatir	1		Curcumin		-
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N
1.064864	0.35665	3	2.044735	1.09324	3	1.10677	0.160821	3

Table S. 72Data table for Figure 6-7D

Control			1 mM SB			5 mM SB		
Mean	SEM	N	Mean	SEM	N	Mean	SEM	Ν
1	0	3	0.877485	0.059914	3	0.710059	0.071135	3
DMSO			Atorvastatir	1		Curcumin		
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N

Table S. 73	Data table for Figure 6-8A
-------------	----------------------------

Control Sodiu			Sodium bu	ityrate		DMSO			Atorvastatin		
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	Ν
1	0	3	6.295907	1.172002	3	1.489678	0.582385	3	1.369342	0.471931	3

Control			Sodium butyrate			DMSO			Atorvastatin		
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	N
1	0	3	2.254542	0.051935	3	0.777882	0.168929	3	0.674292	0.157879	3

Table S. 74Data table for Figure 6-8B

Table S. 75Data table for Figure 6-9A

Control			Atorvastat	in		RG108			Curcumin		
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	Ν
1	0	4	1.243168	0.142191	4	0.74815	0.085685	4	0.35475	0.077943	4

Table S. 76Data table for Figure 6-9B

Contro	Control DMSO				Atorvastatin			Curcumin			
Mean	SEM	N	Mean	SEM	N	Mean	SEM	N	Mean	SEM	Ν
1	0.022094	4	0.778467	0.120491	3	0.926802	0.242326	4	0.632198	0.270369	4

Table S. 77Data table for Figure 6-10A

	Control			Atorvastatin		
	Mean	SEM	N	Mean	SEM	N
H3K9ac	17.42667	0.646022	3	21.72333	2.807266	3
H4K8ac	4.17	0.912798	3	8.853333	0.834293	3
Beads only	0.196667	0.098206	3	0.463333	0.13776	3

Table S. 78Data table for Figure 6-10B

	Control			Atorvastatin		
	Mean	SEM	N	Mean	SEM	N
H3K9ac	23.41	1.718206	3	25.1	6.351795	3
H4K8ac	4.886667	1.309317	3	8.516667	1.190803	3
Beads only	0.17	0.045092	3	0.26	0.09609	3

	Control			Atorvastatin		
	Mean	SEM	N	Mean	SEM	Ν
H3K9ac	18.11667	2.727088	3	24.62333	7.75242	3
H4K8ac	4.87	0.670149	3	9.566667	3.110757	3
Beads only	0.183333	0.061734	3	0.523333	0.078811	3

Table S. 79Data table for Figure 6-10C

Table S. 80Data table for Figure 6-10D

	Control			Atorvastatin		
	Mean	SEM	N	Mean	SEM	N
Н3К9ас	27.82333	10.42558	3	26.24333	12.07619	3
H4K8ac	3.863333	0.758339	3	5.896667	2.012008	3
Beads only	0.216667	0.056667	3	0.25	0.087369	3

Table S. 81	Data	table	for	Figure	6-10E
-------------	------	-------	-----	--------	-------

	Control			Atorvastatin		
	Mean	SEM	N	Mean	SEM	N
H3K9ac	15.91333	6.25713	3	12.38333	3.029331	3
H4K8ac	3.953333	2.242456	3	3.98	0.661085	3
Beads only	0.203333	0.088757	3	0.25	0.091652	3

Table S. 82Data table for Figure 6-10F

	Control			Atorvastatin		
	Mean	SEM	N	Mean	SEM	N
H3K9ac	19.65111	1.339233	9	23.81556	3.05049	9
H4K8ac	4.642222	0.513479	9	8.978889	1.003258	9
Beads only	0.183333	0.036132	9	0.415556	0.066752	9

	Control			Butyrate		
	Mean	SEM	N	Mean	SEM	N
GAPDH	22.9225	4.669926	4	62.2825	14.5707	4
HK1	26.31	5.510706	4	86.7425	12.08141	4
ABCG1	3.8075	1.476947	4	10.5625	2.040631	4
ABCA1	19.1475	5.476867	4	59.2325	9.625128	4
GLUT2	3.085	0.898429	4	7.7625	0.779694	4
HMGCR	12.18	2.290986	4	27.285	1.916148	4
LDLR	17.3575	6.124379	4	54.2725	9.440125	4
PPARG	11.4675	2.137886	4	49.0375	10.1554	4

Table S. 83Data table for Figure 6-11A

Table S. 84Data table for Figure 6-11B

	Control			Butyrate		
	Mean	SEM	Ν	Mean	SEM	Ν
Н3	14.53469	1.901658	32	44.64719	5.38298	32
H3K9ac	23.52063	2.133185	32	65.59625	7.366463	32
H4K8ac	5.385938	0.565206	32	31.17688	3.124793	32
H3K9me3	14.75094	1.425563	32	34.64969	2.75697	32
Beads	0.299375	0.09317	32	0.287813	0.046265	32
only						

Table S. 85Data table for Figure 6-12

	Control			Butyrate		
	Mean	SEM	Ν	Mean	SEM	Ν
H3K9ac	2.170474	0.191635	32	2.062987	0.242709	32
H4K8ac	0.500303	0.048381	32	1.033261	0.120515	32
H3K9me3	1.425377	0.184921	32	1.135229	0.136204	32

	Control			Butyrate		
	Mean	SEM	N	Mean	SEM	N
H3K9ac	3.082598	0.585862	4	3.783811	0.562843	4
H4K8ac	0.533706	0.134205	4	2.072116	0.236257	4
H3K9me3	2.472461	0.704868	4	2.308554	0.417101	4

Table S. 86Data table for Figure 6-13A

Table S. 87Data table for Figure 6-13B

	Control			Butyrate		
	Mean	SEM	N	Mean	SEM	N
H3K9ac	2.187561	0.294915	4	1.937312	0.420533	4
H4K8ac	0.537339	0.109476	4	1.056104	0.1439	4
H3K9me3	1.478608	0.383547	4	1.297084	0.237913	4

Table S. 88Data table for Figure 6-13C

	Control			Butyrate		
	Mean	SEM	N	Mean	SEM	N
H3K9ac	2.560523	0.470258	4	4.078429	1.241096	4
H4K8ac	1.634905	0.341974	4	5.414223	1.846206	4
H3K9me3	3.031282	0.870822	4	4.625379	1.244075	4

Table S. 89Data table for Figure 6-13D

	Control			Butyrate		
	Mean	SEM	N	Mean	SEM	N
H3K9ac	1.940387	0.386981	4	1.79002	0.304871	4
H4K8ac	1.823381	0.463461	4	2.53088	0.734037	4
H3K9me3	1.708368	0.353384	4	2.504702	0.672459	4

Appendix D: Copyright Permissions

PERMISSION TO USE COPYRIGHT MATERIAL AS SPECIFIED BELOW:

Bridgeman, S., W. Northrop, G. Ellison, T. Sabapathy, P. E. Melton, P. Newsholme and C. D. S. Mamotte (2019). "Statins Do Not Directly Inhibit the Activity of Major Epigenetic Modifying Enzymes." Cancers 11(4): 516.

Bridgeman, S. C., G. C. Ellison, P. E. Melton, P. Newsholme and C. D. S. Mamotte (2018). "Epigenetic effects of metformin: From molecular mechanisms to clinical implications." Diabetes Obes Metab 20(7): 1553-1562.

Bridgeman, S. C., W. Northrop, P. E. Melton, G. C. Ellison, P. Newsholme and C. D. S. Mamotte (2020). "Butyrate, generated by gut microbiota, and its therapeutic role in metabolic syndrome." Pharmacological Research: 105174.

I hereby give permission for **Stephanie Bridgeman** to include the abovementioned material(s) in his/her higher degree thesis for Curtin University, and to communicate this material via the espace institutional repository. This permission is granted on a non-exclusive basis and for an indefinite period.

I confirm that I am the copyright owner of the specified material.

Signed:

Name: Phillip E. Melton

Position: Senior Research Fellow

Date: 15/09/2020

PERMISSION TO USE COPYRIGHT MATERIAL AS SPECIFIED BELOW:

Allen SC, Mamotte CDS. Pleiotropic and adverse effects of statins: do epigenetics play a role? J Pharmacol Exp Ther 2017; 362(2): 319-326. doi: 10.1124/jpet.117.242081. (Review)

Bridgeman SC, Ellison GC, Melton PE, Newsholme P, Mamotte CDS. Epigenetic effects of metformin: From molecular mechanisms to clinical implications. *Diabetes Obes Metab* 2018; 20: 1553-1562. doi: 10.1111/dom.13262. (Review)

Bridgeman S, Northrop W, Ellison G, Sabapathy T, Melton PE, Newsholme P, Mamotte CDS. Statins do not directly inhibit the activity of major epigenetic modifying enzymes. *Cancers* 2019; 11(4): 516. doi: 10.3390/cancers11040516.

Bridgeman SC, Northrop W, Melton PE, Ellison GC, Newsholme P, Mamotte CDS. Butyrate, generated by gut microbiota, and its therapeutic role in metabolic syndrome. *Pharmacol Res* 2020; 160: 105174. doi: 10.1016/j.phrs.2020.105174. (Review)

Bridgeman S, Ellison G, Newsholme P, Mamotte, C. The HDAC inhibitor butyrate impairs β cell function and activates the disallowed gene hexokinase I. *Int J Mol Sci* 2021;22(24):13330. doi: 10.3390/ijms222413330.

I hereby give permission for **Stephanie Bridgeman** to include the abovementioned material(s) in his/her higher degree thesis for Curtin University, and to communicate this material via the espace institutional repository. This permission is granted on a non-exclusive basis and for an indefinite period.

I confirm that I am the copyright owner of the specified material.

Signed:	
Name:	Gaewyn Ellison
Position:	Postdoctoral researcher
Date:	13/5/22
Allen SC, Mamotte CDS. Pleiotropic and adverse effects of statins: do epigenetics play a role? J Pharmacol Exp Ther 2017; 362(2): 319-326. doi: 10.1124/jpet.117.242081. (Review)

Bridgeman SC, Ellison GC, Melton PE, Newsholme P, Mamotte CDS. Epigenetic effects of metformin: From molecular mechanisms to clinical implications. *Diabetes Obes Metab* 2018; 20: 1553-1562. doi: 10.1111/dom.13262. (Review)

Bridgeman S, Northrop W, Ellison G, Sabapathy T, Melton PE, Newsholme P, Mamotte CDS. Statins do not directly inhibit the activity of major epigenetic modifying enzymes. *Cancers* 2019; 11(4): 516. doi: 10.3390/cancers11040516.

Bridgeman SC, Northrop W, Melton PE, Ellison GC, Newsholme P, Mamotte CDS. Butyrate, generated by gut microbiota, and its therapeutic role in metabolic syndrome. *Pharmacol Res* 2020; 160: 105174. doi: 10.1016/j.phrs.2020.105174. (Review)

Bridgeman S, Ellison G, Newsholme P, Mamotte, C. The HDAC inhibitor butyrate impairs β cell function and activates the disallowed gene hexokinase I. *Int J Mol Sci* 2021;22(24):13330. doi: 10.3390/ijms222413330.

I hereby give permission for **Stephanie Bridgeman** to include the abovementioned material(s) in his/her higher degree thesis for Curtin University, and to communicate this material via the espace institutional repository. This permission is granted on a non-exclusive basis and for an indefinite period.

I confirm that I am the copyright owner of the specified material.

Signed:

Name: Philip Newsholme

Position Professor, Curtin Medical School

Date: 17th May 2022

Allen, S. C. and C. D. S. Mamotte (2017). "Pleiotropic and Adverse Effects of Statins—Do Epigenetics Play a Role?" Journal of Pharmacology and Experimental Therapeutics 362(2): 319-326.

Bridgeman, S., W. Northrop, G. Ellison, T. Sabapathy, P. E. Melton, P. Newsholme and C. D. S. Mamotte (2019). "Statins Do Not Directly Inhibit the Activity of Major Epigenetic Modifying Enzymes." Cancers 11(4): 516.

Bridgeman, S. C., G. C. Ellison, P. E. Melton, P. Newsholme and C. D. S. Mamotte (2018). "Epigenetic effects of metformin: From molecular mechanisms to clinical implications." Diabetes Obes Metab 20(7): 1553-1562.

Bridgeman, S. C., W. Northrop, P. E. Melton, G. C. Ellison, P. Newsholme and C. D. S. Mamotte (2020). "Butyrate, generated by gut microbiota, and its therapeutic role in metabolic syndrome." Pharmacological Research: 105174.

I hereby give permission for Stephanie Bridgeman to include the abovementioned material(s) in his/her higher degree thesis for Curtin University, and to communicate this material via the espace institutional repository. This permission is granted on a non-exclusive basis and for an indefinite period.

I confirm that I am the copyright owner of the specified material.

Signed:

Name: Thiruvarutchelvan Sabapathy

Position: Research Assistant

Date: 23.09.2020

Bridgeman, S., W. Northrop, G. Ellison, T. Sabapathy, P. E. Melton, P. Newsholme and C. D. S. Mamotte (2019). "Statins Do Not Directly Inhibit the Activity of Major Epigenetic Modifying Enzymes." Cancers 11(4): 516.

Bridgeman, S. C., W. Northrop, P. E. Melton, G. C. Ellison, P. Newsholme and C. D. S. Mamotte (2020). "Butyrate, generated by gut microbiota, and its therapeutic role in metabolic syndrome." Pharmacological Research: 105174.

I hereby give permission for **Stephanie Bridgeman** to include the abovementioned material(s) in his/her higher degree thesis for Curtin University, and to communicate this material via the espace institutional repository. This permission is granted on a non-exclusive basis and for an indefinite period.

I confirm that I am the copyright owner of the specified material.

Signed:

Name: Wendy Northrop

Position: Scientific Officer, Curtin University

Date: 15/9/2020

Allen SC, Mamotte CDS. Pleiotropic and adverse effects of statins: do epigenetics play a role? J Pharmacol Exp Ther 2017; 362(2): 319-326. doi: 10.1124/jpet.117.242081. (Review)

Bridgeman SC, Ellison GC, Melton PE, Newsholme P, Mamotte CDS. Epigenetic effects of metformin: From molecular mechanisms to clinical implications. *Diabetes Obes Metab* 2018; 20: 1553-1562. doi: 10.1111/dom.13262. (Review)

Bridgeman S, Northrop W, Ellison G, Sabapathy T, Melton PE, Newsholme P, Mamotte CDS. Statins do not directly inhibit the activity of major epigenetic modifying enzymes. *Cancers* 2019; 11(4): 516. doi: 10.3390/cancers11040516.

Bridgeman SC, Northrop W, Melton PE, Ellison GC, Newsholme P, Mamotte CDS. Butyrate, generated by gut microbiota, and its therapeutic role in metabolic syndrome. *Pharmacol Res* 2020; 160: 105174. doi: 10.1016/j.phrs.2020.105174. (Review)

Bridgeman S, Ellison G, Newsholme P, Mamotte, C. The HDAC inhibitor butyrate impairs β cell function and activates the disallowed gene hexokinase I. *Int J Mol Sci* 2021;22(24):13330. doi: 10.3390/ijms222413330.

I hereby give permission for **Stephanie Bridgeman** to include the abovementioned material(s) in his/her higher degree thesis for Curtin University, and to communicate this material via the espace institutional repository. This permission is granted on a non-exclusive basis and for an indefinite period.

I confirm that I am the copyright owner of the specified material.

Signed:

Name: Cyril Mamotte

Position: Associate Professor

Date: 17/5/2022